

Copyright Undertaking

This thesis is protected by copyright, with all rights reserved.

By reading and using the thesis, the reader understands and agrees to the following terms:

- 1. The reader will abide by the rules and legal ordinances governing copyright regarding the use of the thesis.
- 2. The reader will use the thesis for the purpose of research or private study only and not for distribution or further reproduction or any other purpose.
- 3. The reader agrees to indemnify and hold the University harmless from and against any loss, damage, cost, liability or expenses arising from copyright infringement or unauthorized usage.

If you have reasons to believe that any materials in this thesis are deemed not suitable to be distributed in this form, or a copyright owner having difficulty with the material being included in our database, please contact lbsys@polyu.edu.hk providing details. The Library will look into your claim and consider taking remedial action upon receipt of the written requests.

Pao Yue-kong Library, The Hong Kong Polytechnic University, Hung Hom, Kowloon, Hong Kong

http://www.lib.polyu.edu.hk

The Hong Kong Polytechnic University Department of Civil and Structural Engineering

ATMOSPHERIC TRANSPORT AND DEPOSITION OF TRACE METALS IN REMOTE RESERVOIRS OF THE SOUTH CHINA COASTAL REGION

By

JIANG Xunyi

A thesis submitted in partial fulfillment of the requirements for the

Degree of Master of Philosophy

July 2009

I

CERTIFICATE OF ORIGINALITY

I hereby declare that this thesis is my own work and that, to the best of my knowledge and belief, it reproduces no material previously published or written nor material which has been accepted for the award of any other degree or diploma, except where due acknowledgement has been made in the text.

JIANG Xunyi

Abstract of thesis entitled 'Atmospheric Transport and Deposition of Trace Metals in Remote Reservoirs of the South China Coastal Region' submitted by JIANG Xunyi for the Degree of Master of Philosophy at the Hong Kong Polytechnic University in July 2009.

ABSTRACT

In the past several decades, enormous economic development has taken place in the South China coastal region with intensive industrialization and urbanization. The increasing emissions of air pollutants from various anthropogenic sources have consequently resulted in substantial enrichment of heavy metals in the atmosphere and have deteriorated the air quality. The outflows of airborne heavy metals might have significant environmental implications for adjacent regions or countries. The present study aims to investigate the atmospheric transport and deposition of some major trace metals (Cu, Ni, Pb and Zn) in some remote areas of the South China coastal region, and to further elucidate the significance of potential outflows of air pollutants to the major receptors at the Western Pacific Rim.

Sediment cores were collected at four remote reservoirs (*i.e.* Tingxi Reservoir, Xinfengjiang Reservoir, Luokeng Reservoir and Songtao Reservoir) along the South China coastline to examine the historical inputs of heavy metals into the aqueous environment. Surface soil samples were collected in the reservoir catchment area to study the elemental concentrations of baseline soils in the drainage basin, and to provide background information of regional geochemical conditions. Moss samples were also collected simultaneously to assess the significance of current atmospheric deposition of trace metals in the study area.

The analytical results showed that the concentrations of heavy metals (e.g. Cu, Pb, and Zn) were elevated significantly in upper layers of sediments at Xinfengijang Reservoir, reflecting the increasing inputs of regional air pollutants in recent years. The lead isotopic compositions of mosses imply that the atmosphere over the study area has been strongly influenced by the regional anthropogenic sources, which might contribute substantial amounts of anthropogenic heavy metals in the surface environment by atmospheric deposition. The intensive industrial activities processing lead-containing ores and coal combustion in the PRD region are probably the major contributing sources of anthropogenic lead at the reservoir. Moreover, the long persistence of past-released Pb, possibly from vehicle exhausts of leaded gasoline, may also contribute to the anthropogenic lead in the atmosphere. The increment of Pb concentrations in top sediments at Xinfengjiang Reservoir might thus be attributed to the increasing atmospheric deposition of anthropogenic pollutants released and transported from the PRD region in the past several years.

IV

However, there is no obvious contamination sign of sediments at Tingxi Reservoir, Luokeng Reservoir and Songtao Reservoir, as the concentrations of trace elements were closely associated with the lithogenic major elements Al and Fe. The variations of trace metal concentrations along the sediment cores at these three reservoirs might be attributed to changes of the natural environment. When analyzing the moss samples, the concentrations of heavy metals accumulated in moss samples were found to be rather low at Songtao Reservoir, probably suggesting that the atmospheric deposition fluxes of heavy metals at this reservoir are less significant. In addition, the lead isotopic compositions of mosses at this reservoir were similar to the values of natural backgrounds, possibly reflecting that the air quality is well preserved in this area. The good air quality might be attributed to its specific topographic characteristics of the reservoir, which may prevent the intrusion of polluted air masses coming from the neighboring developing countries (e.g. Pakistan, Thailand, Vietnam, Indonesia) in Southeast Asia.

Although the sediments are not likely to be in contaminated level at Tingxi Reservoir and Luokeng Reservoir, the low ²⁰⁶Pb/²⁰⁷Pb ratios of mosses at these two reservoirs probably indicate that there are some anthropogenic inputs of Pb by atmospheric deposition from regional emission sources. The inappropriateness of using the reservoir sediments as geochemical archives of the environment changes might be attributed to the low transmission efficiency

through the reservoir catchments due to their small catchment areas, which may subsequently affect the deposition and accumulation of airborne heavy metals in the sediments of the two reservoirs. Further studies need to be conducted to provide additional evidence.

ACKNOWLEDGEMENT

I would like to express my sincere gratitude to my supervisor, Professor Xiangdong Li, for his advice and support during the whole period of my research study. I also want to devote my thanks to Mr. W.S. Lam and Mr. C.W. Lau for their assistance in purchasing chemicals and tutoring in the usage of the analytical equipment. I am also very grateful to both the current and former members of the PolyU Environmental Geochemistry Research Group, namely Dr. Lu Zhao, Ms. Carman C.M. Ip, Dr. Celine S.L. Lee, Dr. Lucy C.L. Luo, Ms. Grace P.Y. Sin, Mr. Clement O.K. Pang and Ms. Chloe W.Y. Tang for their encouragement and help in the past two years. Many thanks also go to Dr. G. Zhang and Mr. T. Lin from the State Key Laboratory of Organic Geochemistry, Guangzhou Institute of Geochemistry, for their cooperation and coordination in preparing the research proposal and conducting the field sampling work. Last but not the least, I also want to devote my special thanks to my parents and all my friends in PolyU for their continuous encouragement and moral support.

TABLE OF CONTENT

CERTIFICATE OF ORIGINALITY	II
ABSTRACT	III
ACKNOWLEDGEMENT	VII
TABLE OF CONTENT	. VIII
LIST OF TABLES	XI
LIST OF FIGURES	. XIII
LIST OF FIGURES	. XIII
CHAPTER 1 – INTRODUCTION	1
1.1 Background	1
1.2 Objectives	5
1.3 Scopes	6
1.4 Organization	7
CHAPTER 2 – LITERATURE REVIEW	9
2 1 Metals & Environmental Geochemistry	ر 9
2.1 1 Metals & adverse impacts of heavy metals	ر 9
2.1.1 Neural biogeochemical cycling of heavy metals	
2.1.2 Anthropogenic sources of atmospheric heavy metals and the	
potential influences.	15
2.1.4 Pathways of accumulation and deposition of heavy metals in	
freshwater ecosystems and influencing factors	19
2.1.5 Long-range transport & monsoon activities	40
2.2 Monitoring Atmospheric Heavy Metals	45
2.2.1 Aerosols	46
2.2.2 Mosses	48
2.2.3 Long-term historical record	50
2.3 Approaches for Pollution Source Identification	59
2.3.1 Lead isotopic analysis	59
2.3.2 Statistical analyses	61
2.3.3 Air back-trajectory analysis	65
	_

CHAPTER 3 – STUDY AREA & METHODOLOGY	74
3.1 Study Area & Geochemical Characteristics	74
3.1.1 Fujian Province	75
3.1.2 Guangdong Province	79
3.1.3 Hainan Province	
3.2 Sampling Locations & Sample Collection	
3.2.1 Descriptions of reservoirs	
3.2.2 Sample collection	
3.3 Sample Preparation and Analyses	
3.3.1 Sample preparation	
3.3.2 Determination of total metal concentrations	
3.3.3 Lead-210 dating of sediment cores	
3.3.4 Lead isotopic composition analysis	
3.3.5 Air back-trajectory analysis	
3.3.6 Statistical analyses	
CHAPTER 4 – ATMOSPHERIC DEPOSITION OF ANTH	ROPOGENIC
HEAVY METALS AT XINFENGJIANG RESERVOIR - IMPL	ICATION OF
REGIONAL AIR POLLUTION	
4.1 Metal concentrations in sediment profiles	
4.2 Enrichment of heavy metals	
4.3 Contributions of potential sources to the enrichment	
4.4 Identification of regional sources	

CHAPTER 5 – THE SUITABILITY OF USING RESERVOIR SEDIMENTS AS GEOCHEMICAL ARCHIVES OF ATMOSPHERIC HEAVY METAL
INPUTS – FACTORS INFLUENCING TRANSPORT AND DEPOSITION OF
AIRBOREN HEAVY METALS
5.1 Concentration Profiles of Metals in Sediment Cores at Different
Reservoirs in the Remote Area124
5.1.1 Tingxi Reservoir124
5.1.2 Luokeng Reservoir129
5.1.3 Songtao Reservoir
5.1.4 Xinfengjiang Reservoir
5.2 Biomonitoring of the Air Quality by Mosses
5.2.1 Metal concentrations in mosses
5.2.2 Lead isotopic compositions144
5.3 Classification of Influencing Air Masses151
5.4 Other Possible Influencing Factors
CHAPTER 6 – CONCLUSIONS AND RECOMMENDATIONS
6.1 Conclusions
6.2 Recommendations
6.3 Future research

REFERENCES
APPENDIX A – ANALYTICAL RESULTS OF SEDIMENT CORES192
APPENDIX B – ANALYTICAL RESULTS OF SURFACE SOILS & MOSSES
APPENDIX C – ANALYTICAL RESULTS OF ²¹⁰ Pb DATING ANALYSIS 220

LIST OF TABLES

Table	2.1 Elemental Concentrations in the Earth's Crust and in Abundant Rock Species (unit: ppm) (Wedepohl <i>et al.</i> , 1991)10
Table	2.2 Comparison of Annual Atmospheric Emissions of Metal Elements from Natural and Anthropogenic Sources (unit: 10 ³ ton) (Compiled from (a) Lantzy and Mackenzie, 1979; (b) Pacyna, 1986; (c) Salomons, 1986, reviewed by Morgan and Stumm, 1991)
Table	2.3 The Annual Global Emissions of Trace Metals to the Atmosphere from Natural Sources (unit: 10^{6} kg) (Hans, 1991)22
Table	2.4 The Annual Global Emissions of Trace Metals to the Atmosphere from Anthropogenic Sources (unit: 10^6 kg) (Hans, 1991)23
Table	2.5 Microtopographic characteristics of some typical natural surfaces affecting dry deposition efficient of air particles (Livett, 1988)31
Table	2.6 Bulk atmospheric deposition fluxes (mg/m ² /yr) for heavy metals obtained from various studies worldwide
Table	3.1 Background elemental concentrations in Fujian aqueous sediments compared to national background levels (unit: $\mu g/g$, otherwise indicated) (Fujian Bureau of Geology, 2008)
Table	3.2 Details of sediment cores collected in the four reservoirs
Table	3.3 Number of moss and surface soil samples collected at the four remote reservoirs
Table	3.4 The general detection limits of ICP-AES for different metal elements (unit: ppb)
Table	3.5 Setup details of air back-trajectories calculated for the four reservoirs
Table	4.1 Pearson correlation coefficients between elemental concentrations in sediment cores at Xinfengjiang Reservoir $(n = 114)^a$ 110
Table	4.2 Lead isotopic compositions of potential natural and anthropogenic sources

Table	5.1 in sed	Pearson correlation coefficients between elemental concentrations iments of Tingxi Reservoir $(n = 30)^a$
Table	5.2 in sed	Pearson correlation coefficients between elemental concentrations iment cores at Luokeng Reservoir $(n = 105)^a$
Table	5.3 in sed	Pearson correlation coefficients between elemental concentrations iment cores at Songtao Reservoir $(n = 107)^a$
Table	5.4 sedim anthro	Lead isotopic compositions of the moss, catchment soil and selected ent samples at the four reservoirs, compared to potential natural and opogenic sources
Table	5.5	Comparison of key characteristics of the four remote reservoirs .159

LIST OF FIGURES

Figure a	2.1 quatic	Pathways of accumulation and deposition of heavy metals in ecosystems and possible influencing factors
Figure	2.2	Particle size distributions of anthropogenic aerosols (Livett, 1988)
Figure	2.3 partic	Size distribution of major and trace metals in atmospheric ulates in Central England (Allen <i>et al.</i> , 2001)26
Figure	3.1	Map of the study area (colored in green)75
Figure	3.2 (Fujia	Gross output values of agriculture and industry from 1952 to 2006 n Bureau of Statistics, 2008)78
Figure	3.3 (Gaze repres (Natio	Air quality in several major cities of Fujian Province in 2006 tte 2006 of Fujian Environment Protection Bureau, 2008). Grade I sents the best air quality, while Grade III represents the worst onal Ambient Air Quality Standard)
Figure	3.4 1978	Gross output values of Guangdong industrial productions since (Guangdong Bureau of Statistics, 2008)
Figure	3.5 atmos 2008)	Increasing amount of total industrial exhaust emission into the phere during 2000-2006 in Guangdong Province (Guangdong EPD,
Figure	3.6 Janua Spatia	Statistics of Regional Air Quality Indexes recorded during ry – June, 2007. (a) Percentages of each air quality grade; (b) Il distribution of mean air quality indexes. (Guangdong EPD, 2008)
Figure	3.7	Map of sampling sites along the South China coastline
Figure	3.8	Sampling locations of sediment cores at Tingxi Reservoir
Figure	3.9	Sampling locations of sediment cores at Xinfengjiang Reservoir.90
Figure	3.10	Sampling locations of sediment cores at Luokeng Reservoir91
Figure	3.11	Sampling locations of sediment cores at Songtao Reservoir92

Figure 3.12 Pictures of a) portable Kajak gravity corer; b) one collected sediment core to be sliced into sub-samples in-situ immediately
Figure 4.1 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Cu (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Xinfengjiang Reservoir
 Figure 4.2 Variations of heavy metal concentrations at different depths of the sediment cores at Xinfengjiang Reservoir: Cu (a), Ni (b), Pb (c), and Zn (d). The annotation 'a' indicates a significant difference of metal concentrations with the above sediment layer at p < 0.05 level, while 'b' implies no statistically significant difference between the two adjacent layers
Figure 4.3 Scatter plots of Cu-Fe (a), Ni-Fe (b), Pb-Fe (c) and Zn-Fe (d) relationships in sediments of Xinfengjiang Reservoir. Linear regression baselines (obtained based on the metal concentrations in bottom sediments (<i>i.e.</i> 15-38 cm) of Core WLH-D) with 95% confidence intervals are superimposed on the data for comparison
Figure 4.4 Comparisons of Pb isotopic compositions between sediments, mosses, and surface soils at Xinfengjiang Reservoir, with reference to available data of backgrounds and potential anthropogenic sources in the PRD region
Figure 5.1 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Co (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Tingxi Reservoir
Figure 5.2 Plot of the unsupported ²¹⁰ Pb radioactivity against the depth of the sediment core LK-B from Luokeng Reservoir
Figure 5.3 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Cu (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Luokeng Reservoir
Figure 5.4 Plot of the unsupported ²¹⁰ Pb radioactivity against the depth of the sediment core LK-B from Luokeng Reservoir
Figure 5.5 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Cu (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Songtao Reservoir
Figure 5.6 Plot of the unsupported ²¹⁰ Pb radioactivity against the depth of the

	sediment core ST-A from Songtao Reservoir
Figure	5.7 Comparison of concentrations of trace elements Cu, Ni, Pb and Zn in moss samples collected at different reservoirs
Figure	5.8 Comparisons of Pb isotopic compositions between sediments, mosses, and surface soils at Tingxi Reservoir
Figure	 5.9 Cluster means of back trajectories ending at Tingxi Reservoir during different time periods: (a) Jan. 2006 – Mar. 2006; (b) Apr. 2006 – May. 2006; (c) Jun. 2006 – Jul. 2006; (d) Aug. 2006 – Oct. 2006; (e) Nov. 2006 – Feb. 2007
Figure	 5.10 Cluster means of back trajectories ending at Xinfengjiang Reservoir during different time periods: (a) Jan. 2006 – Mar. 2006; (b) Apr. 2006 – May. 2006; (c) Jun. 2006 – Jul. 2006; (d) Aug. 2006 – Oct. 2006; (e) Nov. 2006 – Feb. 2007
Figure	5.11 Cluster means of back trajectories ending at Luokeng Reservoir during different time periods: (a) Feb. 2006 – May. 2006; (b) Jun. 2006 – Aug. 2006; (c) Sep. 2006 – Oct. 2006; (d) Nov. 2006 – Jan. 2007155

Figure 5.12 Cluster means of back trajectories ending at Songtao Reservoir during different time periods: (a) Feb. 2006 – May. 2006; (b) Jun. 2006 – Aug. 2006; (c) Sep. 2006 – Nov. 2006; (d) Dec. 2006 – Jan. 2007 156

CHAPTER 1 – INTRODUCTION

For many years, heavy metals have been among the most commonly investigated environmental pollutants, because they are harmful to the ecosystems due to their high toxicity and long persistence in the environment. It has been widely recognized that the ever-increasing emissions of airborne heavy metals from various anthropogenic sources have been endangering the air quality in both the urban and remote areas since the onset of industrial periods worldwide. The surface environment might be subsequently impaired due to atmospheric deposition of the air pollutants. As the metal associated air particles may be subject to regional or long-range atmospheric transport under certain circumstances, the atmospheric heavy metal pollution has inevitably become a global environmental issue.

1.1 Background

Nowadays, the Asian Continent has been regarded as one of the major contributing sources of anthropogenic air pollutants, as enormous economic development has taken place without strict enforcement of proper pollution prevention or control policies. The outflow of air toxics from the Asian Continent might exert adverse impacts on the environment of some major potential receptors, such as Japan, Korea, and southeast coastline of China. Recently, successive evidence has been obtained on the long-range transport of air pollutants from the continental Asia (In and Park, 2003; Mori *et al.*, 2003; Bellis *et al.*, 2005; Jaffe *et al.*, 2005; Lee *et al.*, 2007).

The South China coastal region is one of the most rapidly developing and densely populated areas in China. Since the economic reform in the late 1970s, enormous economic development has taken place with intensive industrialization and urbanization, particularly in the Pearl River Delta (PRD) region. The potential environmental burdens imposed by the drastic development in the past three decades are causing increasing concerns. Recently, several investigations have been conducted to monitor the air pollution problem in South China (Wong et al., 2003; Wang et al., 2005; Hagler et al., 2006; Choi et al., 2008; Fang et al., 2008). The findings suggested that not only the atmosphere in this region had been seriously deteriorated, but also the surface environment (e.g. urban soils, surface waters) which had been inevitably impaired as a consequence of atmospheric deposition. As the South China coastal region is situated in a typical subtropical monsoon area, it is expected that continuous fluxes of air pollutants might be delivered towards the downwind countries or regions by prevailing winds during monsoon seasons (Louie et al., 2005). The air quality condition in this region might thus have significant implications for the environment of adjacent areas or countries, or locations even further away.

A variety of environmental analytical tools have been developed to investigate and monitor the atmospheric pollution of heavy metals. The most common examples include direct measurements of aerosols (Wong et al., 2003; Hsu et al., 2005; Witt et al., 2006), indirect assessment of natural surface soils (Steinnes et al., 1989; Monastra et al., 2004), aquatic sediment cores (Blais and Kalff, 1993; Flower et al., 1997; Yang et al., 2002; Han et al., 2007), and biological samples such as mosses and lichens (Reimann et al., 1997; Szczepaniak and Biziuk, 2003; Harmens et al., 2004; Ruhling and Tyler, 2004). Most of the previous investigations merely focused on a single environmental medium. However, certain limitations have been found in using each type of environment samples as an approach to monitor the air pollution, such as possible post-depositional disturbance in sediment cores (Weiss et al., 1999; Jackson et al., 2004), vulnerability to meteorological changes in ambient environment for passive air samples (Klanova et al., 2008), and inadequacy in comparison between different species for mosses (Lee et al., 2005). A multi-media approach by examining several different environmental components is thus considered as a more effective way to assess the pathway of trace metals in the environment, and to identify signatures of contamination from distinct sources and differentiate the relative contributions of these sources (Duzgoren-Aydin, 2007).

Although the issues concerning airborne heavy metal pollution in South China have been investigated in many studies recently (Fang *et al.*, 1999; Fang *et al.*,

2000; Zhu *et al.*, 2001; Lee *et al.*, 2007; Zhang *et al.*, 2007), most of them are confined to urban areas and they mainly focused on assessing the concentrations of airborne heavy metals, characterizing seasonal variation patterns, evaluating metal associated particle size distributions, or identifying their potential contributing sources. Limited studies have been conducted to assess regional or long-range atmospheric transport and deposition fluxes of heavy metals. The impacts of regional air pollution on ecosystems in rural or remote areas along the South China coastline also remain poorly understood. Moreover, the existing investigations generally assessed the current or recent situations of atmospheric heavy metal deposition fluxes to the surface environment is still scarce.

In order to have a composite and comprehensive view of the situation in the South China coastal region as a whole, there is an emerging need to conduct a more systematic and intensive investigation on the atmospheric heavy metal pollution in a regional scale. Both the long-term variations and current significance of regional atmospheric deposition of trace metals will be assessed. Consequently, the elevated concentrations of airborne heavy metals, increasing regional deposition fluxes, as well as potential outflows of air toxics by long-range transport can be well monitored and controlled. All the findings will then be used for implementing proper mitigation measures and reasonable city planning to avoid or alleviate potential adverse impacts of hazardous trace metals in the coastal ecosystems.

1.2 Objectives

The aim of the study is to conduct a systematic and comprehensive investigation on the significance of regional atmospheric pollution of some major trace metals (Cu, Ni, Pb and Zn) in the South China coastal region, and further elucidate the potential continental outflow of the air toxics from the South China Coastline to the Western Pacific Rim Region.

The major objectives of this study are as follows:

- To assess the historical records and variation trends of heavy metal inputs in well-dated sediment cores from remote reservoirs along the South China coastline.
- To examine the significance of atmospheric deposition of trace metals in the remote area of South China by moss sample analysis.
- To distinguish the input sources of trace metals between natural and anthropogenic sources by lead isotopic composition analysis and elemental association analysis.
- To identify the potential contributing sources of anthropogenic air pollutants by conducting the air back trajectory analysis.
- 5) To establish a database to serve as the solid foundation for future regional

planning and pollution control programs.

1.3 Scopes

The thesis is mainly concerned with the atmospheric deposition of anthropogenic trace metals in the remote area of South China coastal region. The sampling locations were selected at four typical remote reservoirs along the South China coastline, including Tingxi Reservoir in Fujian Province, Xinfengjiang Reservoir in and Luokeng Reservoir in Guangdong Province, and Songtao Reservoir in Hainan Province. The reservoir sediment cores, and surface soil samples and moss samples from the reservoir catchment area were collected at each reservoir for subsequent laboratory analyses. All the samples will be digested by a strong acid mixture of HNO_3 (69%) and $HClO_4$ (70%), and the metal concentrations will be determined by the ICP-AES. Dating of undisturbed sediment cores will be conducted by using the ²¹⁰Pb dating technology. The long-term variations of metal concentrations along the sediment cores will then be examined to evaluate the enrichment of trace metals over time. Metal associations will be investigated by determining the Pearson correlation coefficients and plotting the relationships between the major and trace elements. Lead isotopic compositions of selected sediment samples, catchment soils and mosses will be analyzed by the ICP-MS. In addition, three-day air back-trajectory analysis will be performed by using the HYSPLIT4 model to identify the dominant air masses influencing the air quality

in the study area and the potential anthropogenic sources of heavy metals.

1.4 Organization

The body of the thesis consists of six chapters:

- Chapter 1 gives a general description of project background information and defines the major research objectives.
- 2) Chapter 2 presents a comprehensive literature review about the characteristics of trace metals and their adverse impacts, the metal biogeochemical cycling, the transport pathways of airborne trace metals in the atmosphere and the aqueous environment, the most commonly used technologies for monitoring of atmospheric deposition of heavy metals, as well as the existing investigations on atmospheric heavy metal pollution in the South China coastal region and other countries.
- 3) Chapter 3 illustrates the detailed information of the study area and describes the methodologies adopted for the present research study, including field sampling, sample preparations in laboratory, chemical analysis, air back-trajectory analysis, and statistical analysis.
- Chapter 4 focuses on the results and discussion on the atmospheric deposition of anthropogenic heavy metals at Xinfengjiang Reservoir for implication of potential regional air pollution.
- 5) Chapter 5 compares the results from Tingxi Reservoir, Luokeng Reservoir,

and Songtao Reservoir with those at Xinfengjiang Reservoir to assess the air quality at these three reservoirs and discuss in detail the suitability of using the reservoir sediments as environmental archives for atmospheric deposition of anthropogenic heavy metals.

- 6) Chapter 6 provides a brief summary of the major findings from this research study and also proposes suggestions for improvements and further studies in future.
- All the references supporting the evidence provided in the thesis are listed at the back in the section of References.
- 8) The raw data from which the results were derived are presented in Appendices A to C.

CHAPTER 2 – LITERATURE REVIEW

2.1 Metals & Environmental Geochemistry

Metals and metalloid elements are ubiquitous on the earth. They can be introduced into the environment from both natural and anthropogenic sources in a variety of forms. Environmental geochemistry is the discipline concerning physicochemical parameters and interaction processes that control the distribution of chemical elements over the earth's surface among four reservoirs, namely the lithosphere (the solid earth), atmosphere (the gaseous envelope), hydrosphere (oceans, rivers, streams, lakes, reservoirs, and ground waters), and biosphere (living organisms) (Eby, 2004). Understanding the transport and fate of the elements among various environmental medium can provide a basis for proper management and development of resources, and effective alleviation of the pollution problem.

2.1.1 Metals & adverse impacts of heavy metals

Nearly 99% of the earth's total crustal mass is made of eight elements, including O, Si, Al, Fe, Ca, Na, K, and Mg (Benjamin and Honeyman, 2000). Al, Fe, Ca, Na, K and Mg are therefore considered as major metals due to their large abundance in the crust. Trace elements are normally referred to as the metal elements which are residual and present in trace concentrations (*i.e.* several parts

per million or even less) in the natural environment. Table 2.1 lists the typical concentrations of some major and trace metals in the continental and oceanic earth's crust and in abundant rock types.

Table 2.1Elemental Concentrations in the Earth's Crust and in AbundantRock Species (unit: ppm) (Wedepohl *et al.*, 1991)

	Mg	Fe	Cd	Co	Cu	Cr	Mn	Ni	Pb	Zn
Continental	16000	42000	0.10	19	35	88	800	45	15	69
crust										
Oceanic	50000	70000	0.13	45	81	317	12000	144	0.89	78
crust										
Shales	16000	48000	0.13	19	45	90	850	68	22	95
Greywackes	13000	38000	0.09	20	45	50	750	40	14	105
Limestones	26000	15000	0.16	2	4	11	700	15	5	23
Granites	6000	20000	0.09	4	13	12	325	7	32	50
Gneisses	13000	33000	0.10	13	23	76	600	26	16	65
Basalts	37000	86000	0.10	48	90	168	1390	134	3.5	100
Granulites	14000	38000	0.10	15	27	88	895	33	9.8	65

Generally, metals with atomic number 20 or greater are defined as heavy metals, the pollution of which has gained great environmental concern in recent years (Eby, 2004). Most of these elements belong to trace elements with the exception of Fe (Plant and Raiswell, 1983). Heavy metals can mainly be distinguished into two different groups: essential and non-essential elements. Essential heavy metals are regarded as micro-nutrients that are needed in micro-quantities (*i.e.* a few mg or µg per day) by critical human organs or other organisms to maintain good living or growing status, such as Co, Cu, Mn, Zn (Siegel, 2002). A deficiency of certain micronutrient may result in abnormal conditions or diseases in organisms. However, they could be toxic in case of excessive intake. Inhalation of fumes of volatilized metallic copper can result in nausea, gastric pain, or diarrhea (Scheinberg, 1991). High zinc intake may affect cholesterol metabolism in human beings (Bradl, 2005).

Non-essential heavy metals, such as Be, Cd and Pb, are not required by living organisms. Ingestion of these metals, even in trace levels, could be potentially toxic and harmful. Heavy metals are also well known for their long persistence in the environment. They can retain in the soft tissues of living organisms and thus bio-accumulate to certain levels that stunt plant growth and interfere with animal life (Schwedt, 2001). For instance, excessive Cd can affect kidneys and lungs, leading to protein-urea and emphysema. Long-term exposure to Pb could have poisoning effects on the hemoglobin synthesis, central nervous system and reproduction function (Bradl, 2005). Therefore, it is important to monitor heavy metal enrichment in the environment, and to control and alleviate the pollution problem.

The present study mainly focuses on four trace metals, including Cu, Ni, Pb and Zn. Cu, Ni and Zn are essential to plants, and/or to animals and human beings, but excesses can produce toxicity. Pb is potentially toxic and can exert adverse physiological impacts on living organisms even at trace concentrations. The major elements Al and Fe are also considered as they can play an important role in controlling the primary distribution and secondary dispersion of trace elements (Plant and Raiswell, 1983), and can also be used to assess the extent of heavy metal contamination.

2.1.2 Natural biogeochemical cycling of heavy metals

Unlike many of the synthetic organic compounds, metals are part of natural biogeochemical cycles (Benjamin and Honeyman, 2000). The metal cycling is initiated with the formation of plutonic rocks and ore minerals (Schwedt, 2001). Metal elements are incorporated into orderly internal crystalline structures of the igneous rock minerals upon cooling of magma, which is transported to the earth's surface by intensive geological processes (*e.g.* volcanism or plate tectonics). A wide range of minerals and chemical compositions can be found in the rocks forming at different cooling stages, and certain potentially toxic metals can form their own minerals or comprise large percent of a rock-forming mineral, such as Pb in galena (PbS) and Cr in chromite (FeCr₂O₄) (Siegel, 2002).

The metal elements can then be mobilized or redistributed via physical, chemical and biological weathering and other geochemical processes, and enter into other reservoirs (*i.e.* atmosphere, hydrosphere and biosphere) afterwards. With long-term weathering, rocks are disintegrated or decomposed into loose particles (sediments), which are then accumulated in sedimentary depositories (*e.g.* lakes) or consolidated into sedimentary rocks in geological time scales. Sedimentary rocks are porous and permeable, thus they can be invaded by ore-metal (*e.g.* Pb and Zn) bearing hydrothermal fluid and form ore deposits. The parent rocks could be transformed into metamorphic rocks under increased temperature and pressure conditions. With complicated physical and other geochemical processes, rocks can be further disintegrated into soils (Bradl, 2005). Soils originated from different types of rocks might have markedly varying chemical compositions, resulting in notable differences in natural background concentrations of metals between different regions. For example, soils formed from basalts are potentially enriched in Cr, Ni and Co (Siegel, 2002).

Metals are introduced into the aquatic system mainly in the following ways: 1) surface waters flow through soils and rocks and undergo complicated chemical processes; 2) ground water moves across disintegrated rock layers and receives leaching from overlying soils; 3) the water body receives metal inputs from atmospheric deposition (Bradl, 2005). The chemical compositions of rock-forming minerals or rocks which the water flows on or through can greatly influence its chemical properties. The subsequent fate of the metals in the water body is dependent on many different factors such as temperature, pH, redox potential, adsorption and desorption by suspended particles or bottom sediments, and metabolic processes by microorganisms (Schwedt, 2001).

Naturally, heavy metals in the atmosphere are mostly originated from mineral dusts (from ore deposits or soils), volcano eruptions, sea salt particles, thermal springs, and extraterrestrial matters (Bradl, 2005). They are mainly carried by particulates, aerosols, or gases in some cases (e.g Hg). After emission from the sources, metals may be dispersed and transported by prevailing air mass movement. During the transport process, they may be removed from the atmosphere from time to time by wet or dry deposition onto soils, waters and plants. Atmospheric inputs of metals to the surface environment are geographically different due to varying meteorological factors and distinctive residence time of a particular metal element in the atmosphere (Siegel, 2002).

The biosphere also plays an important role in metal cycling. Metals from waters and soils are capable of entering the food chain via microorganisms, plants and animals, and accumulating to as high concentrations as to be toxic to human beings. Generally, the lithosphere (*i.e.* parent rocks and metallic minerals) is the dominant source of all metals and also serves as the sink for these metals. The hydrosphere, atmosphere and biosphere act as temporary while important reaction reservoirs for the metal elements (Eby, 2004). It is therefore of great importance to study the interactions among the inter-related reservoirs (*i.e.* lithosphere, hydrosphere, atmosphere, and biosphere) in order to assess the transport and fate of heavy metals in the surface environment.

2.1.3 Anthropogenic sources of atmospheric heavy metals and the potential influences

In the previous section (Section 2.1.2), natural sources and biogeochemical cycling of heavy metals are introduced. The metal cycling was maintained basically in a steady state before the technological times. However, large amount of heavy metals, exceeding naturally occurring concentrations, have been released into the environment from many anthropogenic sources (human activities) since the onset of industrialization worldwide (Nyangababo et al., 2005). Anthropogenic contributions to annual heavy metal emissions and environmental burdens were found to have significantly dominated over the natural emissions in many cases (Nriagu, 1989). In Table 2.2, the annual atmospheric emissions of heavy metals from natural and anthropogenic sources are listed and compared. Except Mn and Co, the anthropogenic inputs for the most important trace metals (e.g. As, Cd, Cu, Cr, Pb and Zn) exceeded the contributions from natural sources on the global scale, while natural emission rates were much higher for major elements. The natural biogeochemical cycling of heavy metals in many ecosystems has inevitably been overwhelmed, leading to enormous enrichment in soils (Romic et al., 2003; Loska et al., 2004; Krishna and Govil, 2005; Yang et al., 2007), plants (Bech et al., 1997; Rosen et al., 2001; Al-Homaidan, 2007), aquatic sediments (Sanchez et al., 1998; Huh and Chen, 1999; Li et al., 2000; Feng et al., 2004; Backstrom et al., 2006), air (Deboudt et al., 1999; Petaloti et al., 2006; Tasdemir et al., 2006), and resulting in subsequent accumulation of toxic metals in food chain and living organisms (Outridege *et al.*, 1997; Ip *et al.*, 2005). Besides altering the transferring rates of metals among different reservoirs, the interruption by human activities is also present in altering the form of metals from the originally deposited one, causing changes in metal speciation and potential metal remobilization (Benjamin and Honeyman, 2000).

Table 2.2 Comparison of Annual Atmospheric Emissions of Metal Elements from Natural and Anthropogenic Sources (unit: 10^3 ton) (Compiled from (a) Lantzy and Mackenzie, 1979; (b) Pacyna, 1986; (c) Salomons, 1986, reviewed by Morgan and Stumm, 1991)

Sources		Al	Fe	As	Cd	Co	Cu	Cr	Mn	Ni	Pb	Zn
(a)	Natural	48900	27800	2.8	0.3	7	19	58	605	28	5.9	36
	Anthropogenic	7200	10700	78	5.5	4.4	263	94	316	98	2030	840
	Anthropogenic	0.15	0.38	27.9	18.3	0.63	13.8	1.62	0.52	3.50	344	23.3
	/ Natural											
				_						_		
(b)	Natural	-	-	8	1	5	19	9	520	26	19	45
	Anthropogenic	-	-	24	7	-	56	-	-	47	450	310
	Anthropogenic	-	-	3.00	7.00	-	2.95	-	-	1.81	23.7	6.89
	/ Natural											
(c)	Natural	-	-	21	0.3	7	19	58	610	28	4	36
	Anthropogenic	_	_	78	6	5	260	94	320	98	400	840
	Anthropogenie			2 71	20.0	0.71	12.7	1.60	0.52	2 50	100	22.2
	Anthropogenic	-	-	3.71	20.0	0.71	15.7	1.02	0.52	3.50	100	23.3
	/ Natural											

Historically, anthropogenic inputs of heavy metals can be dated back as early as ancient metal mining and smelting industry during the progression of civilization. Metallic ores were heated, pulverized, and dissolved, and toxic metals became environmentally labile and entered into the natural cycling process until their final deposition in sediments, soils, or biomass (Plant and Raiswell, 1983). In recent time, the major anthropogenic sources of heavy metals include agricultural activities, industrial activities, transportation and solid waste disposal. With development in agricultural productions, fertilizers and pesticides are widely applied to provide additional nutrients to crops and protect them from insects. Mineral (*e.g.* phosphate fertilizers) or micronutrient fertilizers may contain various amounts of potentially harmful heavy metals such as As, Cd, Cu, Pb and Zn (Saltali *et al.*, 2005; Nziguheba and Smolders, 2008). Zarcinas *et al.* (2003) found that the concentrations of As, Cd and Cu in soil samples collected in Peninsular Malaysia were closely correlated to the soluble soil phosphorus, suggesting the contributions of agricultural fertilizers to the enrichment of heavy metals in the soils.

Regarding industrial activities, metallurgical activities including mining, smelting, metal finishing, and others, coal and petroleum combustion for energy production and transportation are considered as the major contributors to the release of heavy metals into the environment, especially to the atmosphere (Bradl, 2005). Waste rocks from mine tailings normally contain high concentrations of As, Cd, Cu, Ni, Pb and Zn (Siegel, 2002). An investigation on heavy metal pollution near a copper smelting plant northeast Turkey showed significant elevation of Cu, Pb and Zn contents in soils, plants and waters nearby (Cubukcu and Tuysuz, 2007). Notable amount of toxic metals, such as Co, Cd, Ni, Pb and Zn, can be found in fly ash from coal combustion, particularly the burning of low quality lignite (Mehra *et al.*, 1998; Baba and Kaya, 2004). In India, the disposal

of coal fly ash has been a major environmental concern due to a yearly production of over 100 million tonnes of ash (Praharaj *et al.*, 2002).

The road traffic is considered as another important contributor of heavy metal emissions to the air. The vehicle based sources, such as deicing salts, combustion exhaust, galvanized parts and railings, fuel and oil combustion, brake linings, and tire-wearing particles, are regarded to have released many anthropogenic heavy metals into the atmosphere. Lead (Pb) was once widely used as a gasoline additive (antiknock agent), and large amount of anthropogenic Pb in the atmosphere was attributed to the emission from vehicle exhausts (Monna et al., 1999). Leaded gasoline has been gradually phased out in developed countries (e.g. western Europe and North America) since the late 1970s, and a declined trend of Pb contents in the aerosols was then observed, from 600 μ g/g in 1979 to 200 μ g/g in 1990 (Wang *et al.*, 2006). The phase-out of leaded gasoline was executed all over China in 2000. However, high Pb concentrations were still detected in the atmosphere over urban areas, possibly due to the long persistence of past-released Pb in different environmental medium or the existence of other potential Pb sources (Chen et al., 2005). The metals of concern emitted from brake linings usually include Cd, Cr, Cu, Ni, Pb, Sb and Zn (Hjortenkrans et al., 2006). Wear from vehicle tires is also a potential source for many toxic trace metals, especially for Zn (Councell et al., 2004; Lim et al., 2006). It was estimated that the traffic sector contributed about 90% of Cd, 40% of Cu, over 99% of both Cr and Ni, 85% of Pb, and 80% of Zn, out of the total diffuse metal emissions in Sweden (Hjortenkrans *et al.*, 2006).

Considerable amount of heavy metals can also stem from sewage effluents and solid wastes generated from both agricultural and industrial processes. Liu *et al.* (2005) reported that long-term irrigation with sewage effluent in Beijing had increased the amount and availability of heavy metals in soils. In southeastern Nigeria, which has had a 40-year disposal history of sewage wastes, heavy metal concentrations in soils were enriched high enough to exert health and phytotoxic risks, particularly for Cd, Cu, Pb and Zn (Udom *et al.*, 2004).

2.1.4 Pathways of accumulation and deposition of heavy metals in freshwater ecosystems and influencing factors

The present study is mainly concerned with issues of regional or long-range atmospheric heavy metal pollution. As mentioned in the introduction, anthropogenic inputs of heavy metals from various emission sources in continental Asia have been endangering the air quality in urban areas. The potential adverse impacts of heavy metal pollution in remote areas have also drawn increasing concerns. Aquatic sediments serve as important ultimate long-term conservative depository for heavy metals and might provide information on the historical trend and extent of atmospheric inputs of heavy metals in certain circumstances (*e.g.* remote lakes without point sources). In order to better interpret the sediment profiles, pathways of heavy metal accumulation and deposition in aquatic ecosystems need to be fully assessed, and possible factors influencing the transport and fate of the metal elements should also be examined. In the following part, the pathways of heavy metals will be described and illustrated through four sections, including emission by potential anthropogenic sources, transportation by prevailing air mass over vast distance, transfer mechanisms in a catchment area, and final deposition in lacustrine sediments. A comprehensive summary of heavy metal transferring processes in various stages is presented and illustrated in Fig. 2.1.


Figure 2.1 Pathways of accumulation and deposition of heavy metals in aquatic ecosystems and possible influencing factors.

2.1.4.1 Emission sources

Nature of emissions

The atmospheric occurrence of trace metals is firstly determined by the nature of emission sources. As mentioned in Section 2.1.3, there is a clear dominance of anthropogenic emissions for most trace elements on the global scale. Different types of natural and anthropogenic sources are also dominated by various types of trace metals. The annual global emissions of trace elements from some major natural sources and anthropogenic sources are presented in Table 2.3 and Table 2.4 respectively.

Table 2.3 The Annual Global Emissions of Trace Metals to the Atmosphere from Natural Sources (unit: 10^6 kg) (Hans, 1991)

Sources	As	Cd	Cu	Cr	Mn	Ni	Pb	Zn
Atmospheric	0.24	0.25	12	50	425	20	10	25
Dust								
Volcanoes	7	0.5	4	3.9	82	3.8	6.4	10
Forest Fires	0.16	0.01	0.3	-	-	0.6	0.5	0.5
Vegetation	0.26	0.2	2.5	-	5	1.6	1.6	10
Sea Spray	0.14	0.002	0.1	-	4	0.1	0.1	0.02

Sources	As	Cd	Cu	Ni	Pb	Zn
Mining, nonferrous	0.013	0.002	0.8	-	8.2	1.6
metals						
Primary nonferrous	15.2	4.71	20.8	9.4	76.5	107
metal production						
Secondary	-	0.60	0.33	0.2	0.8	9.5
nonferrous metal						
production						
Iron and steel	4.2	0.07	5.9	1.2	50	35
production						
Industrial	0.02	0.05	4.9	1.9	7.4	26
applications						
Coal combustion	0.55	0.06	4.7	0.7	14	15
Oil and gasoline	0.004	0.003	0.74	27	273	0.1
combustion						
Wood combustion	0.60	0.20	12	3.0	4.5	75
Waste incineration	0.43	1.40	5.3	3.4	8.9	37
Phosphate fertilizer	2.66	0.21	0.6	0.6	0.05	1.8
manufacture						
Miscellaneous	-	-	-	-	5.9	6.7
Total	23.6	7.3	56	47	449	314

Table 2.4 The Annual Global Emissions of Trace Metals to the Atmosphere from Anthropogenic Sources (unit: 10^6 kg) (Hans, 1991)

As the data show, atmospheric dusts are the dominant source of natural mineral particles containing Mn, Cu, Cr, Ni, Pb and Zn. Volcanoes are regarded as the major emission sources of As, Cd, Pb and Zn. Exudations from plants also contribute markedly to the natural fluxes of As, Cd and Zn compounds. Considering anthropogenic sources, Ni is found to be released mostly from oil combustion. Automobile exhaust emitted by fuel combustion is the predominant source for anthropogenic Pb in the atmosphere. As, Cd, Cu and Zn are mainly derived from non-ferrous smelting industries and secondary metal manufacturing plants, which are also important sources for Ni and Pb.

Size and compositions of source particles

Apart from Hg, which is mainly found in gaseous phase in the atmosphere, trace metals are usually associated with atmospheric particulates (aerosols) in the atmosphere (Witt, 2006). Aerosols can be derived from natural sources such as mineral dust, sea salt and volcanic activities, or are originated from a variety of anthropogenic processes (*e.g.* combustion, industrial activities and automobile) (Motelay-Massei *et al.*, 2005; Tasdemir *et al.*, 2006). It is also considered that secondary particulate matters, in addition to primary pollutants directly released from sources, could be produced during emission, transport and deposition processes such as homogeneous and/or heterogeneous nucleation (Querol *et al.*, 1996). Upon the formation of secondary aerosols, they may exert potentially adverse impacts on the environment, depending on their particle size and the associated acidity (Querol *et al.*, 1999).

The particle size of aerosols from different emission sources can vary significantly. Previous studies revealed that aerosols in coarser size fraction (aerodynamic diameter (*i.e.* AD) > 2.5 μ m) are normally associated with natural sources or produced by mechanical processes, such as erosion or abrasion and during combustion of ash containing fuels (*e.g.* mineral dusts) (Hans, 1991). Although anthropogenic aerosols shows a broad size-distribution from 0.001 μ m

 $50 \mu m$ AD (see Fig. 2.2), most anthropogenic activities involve to high-temperature processes, thus tending to generate much finer particles (i.e. with AD < 2 μ m as the modal class) (Eleftheriadis and Colbeck, 2001; Espinosa et al., 2001). Lam et al. (1999) observed PM_{2.5} constituted up to 70% of the total PM_{10} when investigating the suspended particulate matters in Hong Kong. In such conditions, heavy metals are dominantly released to the atmosphere in their gaseous form and preferentially attached onto finer particles with largest surface areas. However, different heavy metals also show various affinities to different size fractions. It was observed that volatile trace elements such as As, Cd, Se and Pb are more likely to be enriched in fine particles (i.e. accumulation mode 0.3-0.8 µm AD), while oxides of Al, Si, Ca, Fe, Na, Mg and K exist predominantly in coarse particles. Cr, Cu, Mn, Ni, V and Zn exhibit an intermediate behavior and distribute in both fine and coarse particles, with slight enrichment in the finer particle fraction (1-5 µm AD) (Hans, 1991; Sweet et al., 1998; Allen et al., 2001). Fig. 2.3 shows the mass distribution of some typical major and trace metals according to size fractions. As can be noticed from the graph, PM_{1.0} accounted for over 70% and PM_{2.5} comprised over 90% of the total mass for Cd, Pb and Se. As for Zn, Ni, Co, Cu and Mn, the percent out of total mass for $PM_{1.0}$ was 43-62% and the PM_{10} fraction still exceeded 90%. The $PM_{1.0}$ fractions for Ba, Fe and Sr were all less than 20%.



Figure 2.2 Particle size distributions of anthropogenic aerosols (Livett, 1988)



Figure 2.3 Size distribution of major and trace metals in atmospheric particulates in Central England (Allen *et al.*, 2001)

Many researchers suggested that particles less than 10 μ m in diameter (*i.e.* respirable suspended particulates), especially those less than 2.5 μ m, have high capability of respiratory penetration and retention, and are readily taken into human lungs, causing tissue damage and in turn increasing morbidity and mortality (Chan and Kwok, 2000; Cabada *et al.*, 2004; Fang *et al.*, 2005). The particle size distribution of heavy metal associated air particulates is thus

considered of great importance in influencing the potential toxicity of the metal elements.

2.1.4.2 Atmospheric transport and deposition

Transport in the air

1) Particle size distribution

It has been realized that it is not merely the concentrations of total suspended air particulates and metals, but the particle size distribution and associated metal contents, can significantly influence the transport and fate of heavy metals, as well as the potential adverse impacts on human beings (Salma et al., 2005; Srivastava and Jain, 2007). After emission into the atmosphere, aerosols can be carried and dispersed by atmospheric motions over a wide range of distances, and gradually removed from the atmosphere by either dry deposition (e.g. impaction or sedimentation) or wet deposition (e.g. rainout or washout) (Eby, 2004). The particle size is one of the most important controlling factors influencing the extent of metal dispersion via atmospheric transport (i.e. residence time in the atmosphere) and deposition rate of metals to the surface environment (Samara and Voutsa, 2005). Coarse particles are more readily removed from the atmosphere locally, while fine particles might be subject to atmospheric transport over vast distances and finally deposit at regions far away from the emission sources (Allen et al., 2001). The size distribution away from sources is gradually shifted towards smaller particles (Mason et al., 2002). It was

reported that particles larger than 5 μ m were generally absent in rural regions and heavy metals were most frequently associated with small aggregated particles of 0.1-1.0 μ m in polar regions (Livett, 1988).

2) Meteorological parameters

The fate of atmospheric particles and behavior of heavy metals in the atmosphere are also greatly influenced by some meteorological parameters such as the temperature, humidity, rainfalls and winds (Ragosta et al., 2002). These factors probably account for seasonal variations of concentrations of atmospheric suspended particles and associated heavy metals as investigated by many researchers (Querol et al., 1998; Yatin et al., 2000; Fang et al., 2006; Shah et al., 2008). Temperature might affect the levels of total suspended particles in the atmosphere. It was suggested that a higher temperature could favor resuspension of surface soils or dusts (Harrison et al., 1997; Colombo et al., 1999). Humidity and amount of rainfalls are also considered as important factors influencing levels of air particles. Harrison et al. (1997) observed that the concentrations of coarse particles were strongly correlated with wind-derived resuspension processes of dusts during drier seasons of the year and contributed substantially to the elevated levels of total suspended particulates. Higher precipitations were believed to cause more intense scavenging (wash-out) process of aerosols, resulting in lower levels of air particulates (Colombo et al., 1999; Shaheen et al., 2005; Shah et al., 2008).

Another important parameter is the wind effect, which plays a significant role in controlling the transport and distribution patterns of atmospheric particulates, and the behavior of heavy metals in fine and coarse particle fractions. The wind effect mainly refers to the influences of wind speed, wind direction and the mixing height. The wind speed determines the frequency of surface winds from different source sectors. It is therefore closely correlated with the hourly concentrations of metal concentrations in the atmosphere (Ragosta *et al.*, 2002). A lower wind speed normally results in homogeneous distribution of metal concentrations in all sectors (Yatin et al., 2000). Harrison et al. (1997) observed a negative correlation between the wind speed and the levels of fine particles all the year round. However, coarse particles from resuspension exhibited a positive dependence on the wind speed. As for wind direction, the episodic variations can switch on or off the contributions from a specific emission source, leading to seasonal, monthly or even weekly variations of concentrations of atmospheric particulates and associated metals (Ragosta et al., 2002). This parameter is especially important for regions vulnerable to monsoon climate, which could cause significant changes of weather conditions (e.g. prevailing wind direction) and subsequently influence the concentrations of air pollutants (Cheng and Lam, 1998). The anemometric data are usually input in models for predicting the concentration maxima or estimating directions with the worst pollution conditions. In addition, change in the mixing height can cause common

variations in all other meteorological parameters such as temperature, pressure and wind speed. The dependence of elemental concentrations on the mixing height is thus expected to occur. It was revealed that concentrations of metal elements were readily to decrease with increasing mixing height (Yatin *et al.*, 2000).

Moreover, the amount of air particles and associated heavy metals can also be subject to the concurrent influences of other meteorological factors, as well as the seasonal variations of emission source nature and strength (Yatin *et al.*, 2000). Querol *et al.* (1998) noticed that higher oxidation induced by higher insolation and oxidant levels in summer could increase the levels of secondary total suspended particles. They also suggested that higher convective circulation might increase levels of soil-related particles and primary anthropogenic particles.

Atmospheric deposition

There are four distinct processes of dry deposition based on the particle sizes of air particulates, namely gravitational settling, impaction, turbulent transfer, and transfer by Brownian motion (Livett, 1988). Gravitational settling is the major means of deposition for coarse particles, while the principal mechanisms for deposition of fine particles are impaction and turbulent transfer. The deposition efficiency of fine particles in remote areas is thus greatly dependent on the characteristics of the recipient surface. Table 2.5 shows the properties of some natural surfaces and their aerodynamic resistance to deposition. An area with a higher aerodynamic resistance factor is expected to have lower deposition efficiency.

Table 2.5	Microtopographic	characteristics	of some	typical	natural	surfaces
affecting dry	deposition efficien	t of air particles	s (Livett,	1988)		

Surface	Geochemical monitoring equivalent	Canopy height (m)	Roughness length (cm)	Aerodynamic resistance (s cm ⁻¹)	
Smooth ice	Ice deposits	-	0.003	0.46	
Ocean	Aquatic sediments	-	0.005	0.44	
Tilled soil	Bare peat / rough snow	-	0.1	0.32	
Thick grass	Vegetated peat (e.g.	0.1	2.3	0.19	
	Eriophorum vaginatum)				
Dwarf shrubs	Vegetated peat (e.g.	< 1.0	-	-	
	Ericaceae)				
Shrubs	-	1.5	20.0	0.10	

As the data indicate, the dry deposition of air particles might be most effective for densely vegetated surfaces while least effective for snow and ice. The aerodynamic resistance of aquatic sediments is also found to be comparatively higher, approximately the value of ice deposits.

Wet atmospheric deposition (*e.g.* rainfalls and snows) mainly involves two processes, including nucleation (*i.e.* condensation of water vapor droplets), and within- or below-could scavenging of wetted aerosol particles by falling raindrops. Rainwater is regarded as the most effective scavenging agent for removing particulate or dissolved gaseous pollutants from the atmosphere (Al-Khashman, 2005). The scavenging of atmospheric particulates has a great influence on the chemical compositions and acidity (*i.e.* pH) of the rainwater via an interplay of the acidic and alkaline components. The acidic compounds generally come from gases given out by soil microbial activity, terrestrial and marine biological activity, and industrial and urban emissions, whilst alkaline components are mainly coarse aerosols (e.g. dusts) derived from soil resuspension (Datar *et al.*, 1996). The efficiency of wet deposition, described by washout factor (i.e. concentrations of an element in the precipitation vs. concentrations of an element in the air), depends on the solubility of the metals and their compounds, as well as the absolute concentration of impurities in the aerosol (Livett, 1988). Desboeufs et al. (2000) noticed that metals contained in carbonaceous aerosols were easier to be dissolved than those in aluminosilicated particles. As the metals associated with carbonaceous particles are very soluble, the dissolution is not likely to be dependent on the pH. However, metals associated with aluminosilicated particles are less soluble, especially for those constitutive of the matrix network (e.g. Fe, Mn), and have a dissolution highly influenced by the acidity. Normally, the highest concentrations of elements can be observed at the beginning of the rainfall season when large amounts of dust accumulated in the atmosphere are scavenged by rain (Al-Khashman, 2005). In case of clean air condition, the low concentrations of air impurities normally result in poor efficiency of atmospheric cleansing, leading to a close similarity of chemical compositions between the precipitation and air.

It was suggested that the atmospheric dry fluxes are more significant for crustal elements (*e.g.* Al, Fe and Mn), while trace elements (*e.g.* Cd, Cr, Cu, Pb and Zn) are predominant in wet fluxes (Church *et al.*, 2002). The percentage of heavy metal soluble fraction in dry deposition is generally lower than in wet deposition. Considering different trace metals, Cd, V, Cu and Zn exhibit a higher average solubility than Cr, Ni and Pb in both dry and wet deposition (Morselli *et al.*, 2003).

2.1.4.3 Deposition and transmission from the catchment

Atmospheric heavy metals can enter the freshwater system (*e.g.* lakes or reservoirs) either via direct fluxes to the surface water by dry or wet deposition, or by indirect transmission of atmospheric inputs through the watershed (Church *et al.*, 2002). The metals in the terrestrial drainage basin may be carried into the freshwater system by headwater stream discharges (both dissolved and particulate fluxes), or they may be leached through permeable soil layers (leaching fluxes) or dissolved in the groundwater (Schwedt, 2001). The contributions of atmospheric heavy metal inputs from the catchment might be appreciable, or even dominant over direct atmospheric deposition in case of high drainage ratio (catchment area vs. open surface water area) of the lake or reservoir (Shotbolt *et al.*, 2006). Ukonmaanaho *et al.* (2001) found that two thirds of the inputs of Cd, Ni, Pb and Zn to the receiving lake were from the terrestrial component of the catchment, with the remaining being directly from

atmospheric deposition. Compared to natural lakes, water-supply reservoirs are more susceptible to the strong influences of metals exported from the catchment with relatively higher sedimentation rates, as they are generally located at regions where large catchment area can be utilized and preferably abundant precipitations are available (Shotbolt et al., 2005). The significance of indirect loading atmospheric heavy metals of is closely related to the transmission/retention capacity of the catchment, which are mainly influenced by soil characteristics (e.g. pH, the abundance of Fe/Mn oxides and organic matters, clay content, surface area, and cation exchange capacity), chemical properties of the metal (e.g metal affinity to particles), geomorphology, land use, amount of rainfalls, and the basic hydrological characteristics (Schmitt and Sticher, 1991; Blais and Kalff, 1993; Church et al., 2002). Organic-rich soils, such as forest soils, might be more effective for metal retention (Ukonmaanaho et al., 2001). The flux of organic matter to the receiving water, on the other hand, significantly governs metal export from catchment soils (Dillon et al., 1988). Metals can be present in the soil solution in the form of metal ions or of inorganic or organic complexes, and can also be adsorbed to the surface of clay minerals or clay-humus complexes. The bond type of heavy metals, and the redox condition and pH of soils can greatly affect the solubility and mobility of the metal elements (Schwedt, 2001). Metals in easily exchangeable forms might be more susceptible to pH changes so that accelerated deposition of strong acid or acid precursors (SO_2) is expected to cause increased mobility of the metals (Dillon *et*

al., 1988). Different metal elements exhibited distinctive behaviors in catchment soils due to different physico-chemical properties. Cr has a high mobility at pH > 7, while the solubility of Cd, Pb and Zn increases in more acidic conditions (Schwedt, 2001). Cr normally forms oxide anions in soils, which tend to be repelled by clays and Fe oxyhydroxides thus enhancing export fluxes from soils. Cu, Cr and Pb are found to be strongly correlated with dissolved organic carbon while Cd, Ni and Zn have a lower affinity for organic matters (Blais and Kalff, 1993; Ukonmaanaho et al., 2001). Moreover, the catchment metal export behaviors are not consistent, possibly ascribed to particular geological conditions in different regions. Some researchers noticed that anthropogenic Cu, Pb and Zn could be effectively retained by catchments while Cr and Ni were readily exported (Dillon et al., 1988; Blais and Kalff, 1993; Ukonmaanaho et al., 2001). However, Church et al. (2002) observed that the atmospheric Pb inputs to the catchment were almost fully delivered to the receiving water, while Cr, Cu and Zn were retained by the terrestrial basin to certain extent and tended to be periodically remobilized by acidic precipitation events. In addition to the indirect fluxes of atmospheric heavy metals, metals exported from the terrestrial basin into the freshwater system can also be derived from erosion of natural catchment soils. The amount and particle size of the soils transported into the water body may vary temporally and spatially. This portion of inputs is considered of importance for particle-reactive elements, such as Fe (Church et al., 2002). The fluxes of eroded materials from catchments with serious erosion could be high

enough to the point of obscuring the evidence of atmospheric inputs, thus leading to unsuitability of the aquatic sediments as geochemical archives for historical aerial deposition of heavy metals (Yeloff *et al.*, 2005). The possible usage of solid control system can be used to reduce sedimentation rate, but might also alter the particle size distribution, possibly leading to shifting to finer-grained sediment components.

2.1.4.4 Transfer mechanisms in the aquatic system

Water column

The behavior of metals in the aquatic ecosystem is closely related to a number of distinct physical and chemical interactions between different components within both particulate and aqueous phases. Considerable amount of atmospheric heavy metals entering the aquatic environment are associated with fine particles (< 1 μ m) or in soluble forms (Lion *et al.*, 1982; Livett, 1988). The interface between the water and the atmosphere (the uppermost thin water film with thickness of 50-100 μ m) known as microlayer is regarded as the first site where complex physicochemical interactions could occur. Heavy metals concentrated in the surface microlayer might be derived from atmospheric deposition, or brought by bubble flotation from the water column (Cuong *et al.*, 2008). They are subject to scavenging by surface-active molecules in surface nanolayer (0.1-1.0 μ m) or particulate matters and microorganisms enriched in surface microlayer (<10 μ m) (Wurl and Obbard, 2004). The organic acids, proteinaceous materials, and other

surface-active organic substances and particles might provide complexing sites for many heavy metals and thus be responsible for the transport and enrichment of these metals at the water surface (Duce *et al.*, 1972). Many researchers have reported the high enrichment of heavy metals in the microlayers relative to the bulk water column (Duce *et al.*, 1972; Wurl and Obbard, 2004; Cuong *et al.*, 2008).

In bulk water column, scavenging of heavy metals by particulate matters is the most important process controlling the accumulation and distribution of metals (Gonzalez-Davila, 1995). Heavy metals can be transferred from aqueous phases particulate phases by several reactions, including precipitation and to co-precipitation with hydroxides, sulphides carbonates; cation or exchange/sorption on clay minerals, iron/manganese oxides and sulphides, carbonates, phosphates and organic matters; and complexation with organic matters (Livett, 1988). These reaction processes are highly dependent on the binding characteristics of the particulate matters, such as the grain size. Marked grain size partitioning of heavy metal concentrations in the sediments has been widely recognized. Fine particles are believed to bind with metal ions more strongly than coarse particles as they provide much larger surface-active coating areas containing organic matters and Fe/Mn oxide (Yang et al., 2002; Ip et al., 2007). The efficiency of a distinct reaction also depends notably on the affinity of metal elements for a particular ligand and their respective concentrations, as well as the redox potential and pH. Organic matters, either natural dissolved or in particulate form, are regarded as critical scavengers of heavy metals in natural waters (Gonzalez-Davila, 1995). Serious impairment of the aquatic ecosystem might be caused by heavy metal enrichment during phytoplankton bloom periods (Admiraal *et al.*, 1995; Garcia-Hernandez *et al.*, 2005). Metals can be directly accumulated by the algae, or complexed by their extracellular metabolites, and significant amount of heavy metals may be subsequently incorporated into the sediments below (Garban *et al.*, 1999). The pH is another major factor influencing the behavior of heavy metals in the water column. A lower pH might increase the concentrations of soluble heavy metals, leading to decreased efficiency in scavenging mechanisms.

<u>Sediments</u>

Particles associated heavy metals will be gradually removed from the water column, and sink and accumulate in the underlying sediments. The newly deposited particulate matters form a layer over the primary sediment layer and may particularly susceptible to physical and chemical remobilization. The sediment redistribution (*i.e.* sediment focusing) triggered by thermal cycles or seasonal turbulence may consequently result in a proportionately greater loading of heavy metals in deeper layers of sediments (Yang *et al.*, 2002). The impacts of drawdown on sedimentation and sediment distribution are considered minor in deepwater sediments (Shotbolt *et al.*, 2005).

Heavy metal associations are subject to diagenetic modification when entering the primary sediment layer. Degradation of organic matters first occurs, which is especially faster in aerobic conditions. The released heavy metals usually accumulate in the top 1-3 cm of sediments. Fate of these metal elements greatly depends on the redox potential of the sediments and the overlying water column. For example, Cd, Cu, Ni and Pb are likely to be fixed in reducing conditions, while Fe and Mn are normally fixed in oxidizing conditions (Livett, 1988). The formation of sulphide in seasonal anoxic and eutrophic sediments from degradation of organic materials can accumulate large amount of heavy metals, which might be released to the water column in case of periodic oxidizing conditions (Canavan et al., 2007). Coprecipitation with Fe/Mn oxides is another important mechanism influencing the accumulation, speciation and vertical distribution of heavy metals in sediments (Douglas and Adeney, 2000). Fe²⁺ and Mn^{2+} tend to diffuse upwards to the redox boundary and be oxidized to Fe³⁺ and Mn⁴⁺, which are efficient scavengers of released heavy metals and can regulate the flux of metals into the overlying water column. The enrichment of Fe/Mn oxides can usually be observed at or near the sediment-water interface, and is mostly associated with peaks of heavy metal concentrations.

The acidity of bulk water and pore waters can also affect the fate of heavy metals in the sediments. In case the water body is suffering serious acidification, the lowered pH may not only reduce the efficiency of heavy metal scavenging by particulates, but also potentially remobilize metals from the surface sediments to the overlying water column, such as Cd, Ni and Zn (Fjeld *et al.*, 1994; Kjoller *et al.*, 2004). The distribution of heavy metals may also be significantly influenced by bioturbation. It may lead to mechanical mixing of surface sediments, or facilitate chemical remobilization of heavy metals by sediment aeration (Livett, 1988).

2.1.5 Long-range transport & monsoon activities

In the previous section (Section 2.1.4), the pathways of heavy metals from the emission sources to final deposition in sediments are described. It can be noticed that atmospheric transport of heavy metals by air mass movements (*i.e.* the wind effect) plays an important role in controlling the distribution and fate of anthropogenic air pollutants. In this section, two major issues considering the atmospheric transport of heavy metals are presented.

Recently, the term "long-range transport" (LRT) has gained great attention. It is widely recognized that LRT has promoted the globalization of environmental pollution. The cross-boundary transport of air pollutants was historically studied in European countries to assess the significance of acid deposition in the 1970s (Mukai *et al.*, 1994). The LRT was also recognized when investigating the Kosa phenomenon in spring, when yellow sands were blown from Asian Continent to Japan. It was then widely investigated by many researchers in North America, revealing the possible transport of airborne particulate matters from other regions (Kritz et al., 1988; Parrish et al., 1992; Jacob et al., 1999; Jaffe et al., 2003). They suggested that high levels of air toxics observed in North America might be attributed to two major influences: 1) 'events', whereby air masses containing relatively high concentrations of pollutants from Asia are rapidly transported across the Pacific; and 2) background and local contributions. Successive evidence has been released that trans-Pacific pollutant transport or inter-continental global transit could be expected to occur under some circumstances. However, most of the air toxics examined focused on dusts (Fang et al., 1999; McKendry et al., 2001; In et al., 2003), non-seasalt sulfate (Berresheim, 1993), combined nitrogen (Nakamura et al., 2005), CO (Jaffe et al., 1997; Chan et al., 2002; Goldstein et al., 2004), mercury (Jaffe et al., 2005), radon (Kritz et al., 1988), ozone (Jacob et al., 1999; Chan et al., 2002), non-methane volatile organic compounds (Guo et al., 2006), and peroxyacetyl nitrate (Jaffe et al., 1999).

Considering trace metals, they are highly persistent in the environment and are not biodegradable. Since heavy metals of anthropogenic origins are normally associated with fine air particles or in gaseous form (*e.g.* Hg), they might be subject to possible LRT by prevailing air mass movements, and settled by dry or

wet deposition in area far away from the source areas. Recently, more evidence on the possibility of long-range transport of heavy metals has been found (Reimann et al., 1997; Steinnes et al., 1997; Bellis et al., 2005; Hsu et al., 2005; Lee et al., 2007). The outflow of heavy metals from continental Asia has been reported by many researchers. Bellis et al. (2005) observed that lead isotopic ratios of aerosols in Japan during winter were indicative of the contributions by certain lead ores and coals in continental Asia, reflecting the long-range transport of Pb from continental Asian to Japan. Lee et al. (2007) revealed that the enrichment of heavy metals in South China coastal region (e.g. Hong Kong and Guangzhou) was closely associated with the air masses from the northern inland areas of China by conducting the air back-trajectory analysis. Han et al. (2004) also attributed the increased atmospheric dry deposition of heavy metals in Korea mainly to the long-range transport of Al and Pb from major industrial areas and deserts of North China during yellow-sand events in spring. In North America, DesJardins et al. (2004) reported that the lead isotopic compositions in Kejimkujik National Park, Nova Scotia were influenced by the binary mixing of anthropogenic pollutants transported from both southeastern Canada and northeastern US. Moreover, it was revealed there was a strong outflow of Hg transported across the Pacific to North America (Jaffe et al., 2005). Therefore, it is inevitable that the air quality in rural or remote areas might also be influenced greatly by the long-range transport of heavy metals from industrial and urbanized areas (Flower et al., 1997; Fitzgerald et al., 1998; Rose and Rippey, 2002). In

certain circumstances, the aerosols from Norilsk and Kola Peninsula were even detected in the Arctic atmosphere (Shevchenko *et al.*, 2003).

It has been widely recognized that the long-range transport of heavy metals is dependent on seasonal variations of prevailing winds, caused by strong monsoon activities (Mukai *et al.*, 1994; Siefert *et al.*, 1999; Matsuki *et al.*, 2003; Lee *et al.*, 2007). Monsoons are normally described as the seasonal reversal of wind direction, caused by temperature differences between the land mass and ocean area. They play an important role in the management of global climates. Monsoons can affect the way that the land, sea and atmosphere regimes interact with each other and influence weather through the exchange of moisture and energy. Therefore, monsoons are considered to have a great impact on the atmospheric circulation and the geographic environment, thus affecting the source strength and pathways of air pollutants.

The Eastern China and South China coastal region are typical monsoon areas, which are sensitive to monsoon fluctuations and subject to significant seasonal climate changes (Ding, 1994). It was recently reported by Han *et al.* (2007) that the East Asian monsoon is a significant factor that affects atmospheric heavy metal deposition, as indicated by strong associations between the variations of heavy metal concentrations and climate changes in accordance with the East Asian monsoons.

In general, the summer monsoon corresponds to warm and wet climate, while the winter monsoon corresponds to dry and cold climate in China (Han et al., 2007). During May to August, the summer monsoon prevails in China. It originates from three airflows: 1) the Indian summer monsoonal airflow; 2) the cross-equatorial airflow from Australian region flowing over Southeast Asia and the South China Sea; 3) the southeast monsoon coming from the southern flank of the subtropical high over the western Pacific (Ding, 1994). The summer monsoon advances northward in stepwise way with simultaneous northward movement of precipitation belts. When it finally comes to mid October, the summer monsoon completely withdraws from the Mainland and the winter monsoon prevails instead. The characteristic weather of winter monsoon is marked by the invasion of extremely dry and cold air and strong winds over Mongolia, northern China, and Korea. The Siberian high which intrudes into China has two tracks, the major one of which extends southeastward from the source region to the west of Lake Baikal, passing the middle Yellow River Valley, northern China, and the lower Yangtze River Valley, and eventually extends toward the sea region to the south of Japan or down to southern China (Ding, 1994).

For a long time, many researchers have been engaged in numerous studies to assess the impacts of monsoons on the weather and climate changes (An *et al.*,

2000). However, the studies of monsoons on air pollutants dispersion and transport, as well as atmospheric deposition are still scarce. As the study area of the present investigation are located in regions vulnerable to monsoons activities, it is necessary to take this factor into consideration when analyzing seasonal variations of long-range transport and atmospheric deposition of heavy metals. The results might help to effectively monitor the air quality and control regional pollutions.

2.2 Monitoring Atmospheric Heavy Metals

Nowadays, a variety of analytical tools (*i.e.* environmental materials) have been developed to investigate and monitor the atmospheric pollution of heavy metals. Examples include direct measurements of aerosols (Wong *et al.*, 2003; Hsu *et al.*, 2005; Witt *et al.*, 2006; Lee *et al.*, 2007), indirect assessment of natural surface soils (Steinnes *et al.*, 1989; Zhu *et al.*, 2001), rain and snow samples (Takeda *et al.*, 2000), ice cores, peat bogs, sediment cores (Blais and Kalff, 1993; Flower *et al.*, 1997; Yang *et al.*, 2002; Han *et al.*, 2007), and biological samples such as mosses and lichens (Reimann *et al.*, 1997; Szczepaniak and Biziuk, 2003; Harmens *et al.*, 2004; Ruhling and Tyler, 2004), tree trunk bark pockets (Satake *et al.*, 1996; Bellis *et al.*, 2005).

Monitoring of air pollution by a relatively systematic and quantitative analytical

approach was pioneered by A. C. Smith in the mid-nineteenth century (Livett, 1988). It is not until the middle of twenties century that more attention began to focus on heavy metals, especially Pb, and conventional monitoring methods were established. Alternative monitoring means by biomonitors, such as lichens and mosses, were developed around the 1970s (Ruhling and Tyler, 1968, 1969)

In the following four sections, the most widely used approaches, namely aerosols, sediment cores, peat bogs, and mosses, are illustrated respectively.

2.2.1 Aerosols

Collection of aerosol samples by different sampling devices is the most direct approach to analyze the atmospheric pollution of heavy metals. The most commonly used sampling devices include low-volume air samplers (Mukai *et al.*, 2001), high-volume devices (Lee *et al.*, 2007), and passive bulk deposition samplers (Wong *et al.*, 2003).

With regard to low-volume and high-volume air sampling, air is actively drawn through filters, on which the air particles are retained and collected. This type of measurements could effectively provide information on the chemistry of aerosols suspended in the atmosphere. For years, active air sampling has been widely applied in combination with air back trajectory analysis and source attribution analysis to provide useful information on the long-range transport of trace metals (Mori *et al.*, 2003; Witt *et al.*, 2006; Lee *et al.*, 2007). However, it was reported (Guo *et al.*, 2003) that high-volume sampling on fiber filters is prone to sampling losses due to 'blow-off' effect on volatile species and filter-catalyzed oxidation. Therefore, the particulate phase of the target compounds is normally collected using a fiber or membrane filter (*e.g.* glass/quartz micro-fiber filters (GFF/QFF)), while the portion associated with gas phase is sampled using an adsorbent trap (*e.g.* polyurethane foam (PUF)) (Hart *et al.*, 1992). Moreover, active air sampling requires pumps, sampling heads, and electricity or other power sources to draw air samples actively, thus it is not applicable in remote sites.

The obtained data from active air sampling do not reflect the chemistry of aerosols removed from the atmosphere by dry or wet deposition. Therefore, passive air sampling technology by alternative samplers has been introduced. The collection of atmospheric depositions is merely dependent on the ambient environment and allows chemical accumulation via natural diffusion, which gives more reliable information as to the influences of atmosphere inputs of heavy metals on the surface environment (Wong *et al.*, 2003). Nevertheless, passive air sampling process normally takes much longer time and might also be sensitive to sudden changes of meteorological conditions such as wind speed and temperatures.

Considering the merits as well as drawbacks of both sampling technologies, passive air sampling is suggested to be a method complementary to the collection of high-volume active samples for monitoring long-term trends of heavy metal variations, which can enable the cost-effective collection of monthly or seasonally integrated air samples deployed throughout a large geographic region (Gouin *et al.*, 2005).

2.2.2 Mosses

If aerosols are considered to be a direct approach to analyze heavy metals, then analyzing organisms as bio-accumulators could be regarded as an indirect monitoring method. Biomonitoring of heavy metals in the atmosphere by mosses is a well-developed technique since the 1960s (Ruhling *et al.*, 2004). It has been widely applied in many parts of the world for monitoring airborne heavy metal deposition, especially in countries all over the Europe (Reimann *et al.*, 1997; Harmens *et al*, 2004) and North America (Schilling & Lehman, 2002).

The increasing use of mosses as a biomonitor is based on the fact that mosses have no root system and can receive water and nutrients merely from precipitation or dry deposition; thus the uptake of mineral substrates from soils is quite insignificant (Ruhling *et al.*, 2004). In addition, they do not have epidermis and cuticle, so metal ions can easily penetrate the cell wall and be strongly concentrated in mosses (Szczepaniak and Biziuk, 2003). Many moss species are geographically widespread and can grow in different environmental conditions. As this technique is easy to handle (fewer analytical problems than lichens and tree barks) and less costly, it has been widely adopted for surveys of atmospheric heavy metal deposition in large areas or even nations (Harmens *et al.*, 2004; Ruhling *et al.*, 2004; Lee *et al.*, 2005).

The most preferred species of mosses for the use in Europe are the carpet-forming mosses *Pleurozium schreberi* and *Hylocomium splendens* due to their great sensitivity to heavy metal pollutions (Szczepaniak and Biziuk, 2003). Other most frequently sampled species include *Hypnum cupressiforme* and *Scleropodium purum*, which are more widespread in south areas of Europe (Lee *et al.*, 2005). However, the studies on atmospheric deposition of heavy metals conducted based on mosses are still quite scarce in Asian countries. Recently, Lee *et al.* (2005) attempted to use the moss *Hypnum plumaeforme*, which is mostly commonly found in dense mats in many Asian countries (*e.g.* China, Japan, Korea, Nepal and Philippines), to analyze heavy metal deposition. The results turned out to be quite promising, for this type of moss is able to capture heavy metals from the atmosphere efficiently and presents high bioaccumulation rates.

However, when a moss survey is conducted over vast areas, it is inevitable that a

wide range of moss species are involved. Therefore, data obtained from one certain specie of moss should be only comparable to similar species belonging to the same class. On the other hand, natural variability in ambient climate conditions, such as acidity, temperature, humidity, light and altitude, or ambient elemental occurrences, may affect the bioaccumulation behavior of mosses. Furthermore, the proliferation of indigenous moss is normally not encouraged in urbanized or contaminated areas (Wong *et al.*, 2003). The major disadvantage of the passive biomonitoring method is the difficulty in source identification. Since all processes and sources occur simultaneously, there is no possibility to separate them and look for a particular one (Szczepaniak and Biziuk, 2003).

2.2.3 Long-term historical record

The naturally occurring deposits, such as lake sediments, glacial ice, snow, and ombrotrophic peat bogs, have been most extensively used as historical archives to monitor long-term atmospheric deposition of heavy metals. However, each of them has certain drawbacks. For example, peat bogs and sediment cores might be subject to remobilization of heavy metals due to physical and chemical changes. All of them are thus required to meet with certain requisite criteria in order to serve as potential geochemical archives. They should receive only atmospheric inputs of anthropogenic heavy metals and be subject to minimal post-depositional migration of deposited substances. In addition, they must be able to accumulate sufficiently measurable concentrations of elements of interest (Benoit *et al.*, 1998).

2.2.3.1 Peat bogs

Recently, there is an increasing interest in the use of ombrotrophic peat bogs as geochemical archive. Peat consists of organic deposited matter produced by vegetation. In ombrotrophic peat bogs, the only source of water and nutrients is from atmosphere by wet and dry deposition (Weiss *et al.*, 1999), because the surface layers of ombrotrophic peat bogs are isolated from the underlying mineral soil or local ground and surface waters (Ukonmaanaho *et al.*, 2004).

Although mosses are more widely used as bioindicators of atmospheric pollutions, they only indicate current situations. In contrast, the vertical distribution of elements in peat bogs could provide information on recent deposition of trace elements and preserve historical records for temporal trends of atmospheric fluxes. They can be precisely dated using ¹⁴C (bomb-pulse and traditional) or ²¹⁰Pb dating technologies (Le Roux *et al.*, 2005). Therefore, ombrotrophic peat bogs, together with glacial ice cores, are considered as the only archives recording exclusively atmospheric deposition of heavy metals in the last few hundred years. Nevertheless, measurement of heavy metals in ice cores has been found to be difficult due to the low concentrations (Benoit *et al.*, 1998; Weiss *et al.*, 1999). In addition, peatlands are widely distributed all over

the world, thus the approach using peat bogs is applicable at many conditions for their high availability (Le Roux *et al.*, 2005). For another reason, peat formation in the northern hemisphere began after the retreat of glacial ice, which makes it possible to record the entire Holocene (Weiss *et al.*, 1999). Therefore, this method has been proved to be an effective tool to reconstruct the historical records of atmospheric deposition of heavy metals.

However, the possible importance of post-depositional migration in peat bogs still remains uncertain for most of the metals of interest (Weiss et al., 1999; Shotyk et al., 2000). Peat bog waters are naturally acidic and organic-rich, and may vary between oxic and anoxic, depending on the depth of the water table which fluctuates seasonally (Shotyk et al., 2000). Hence, metal elements may be transformed chemically and liberated from peat bogs. There are even arguments about the accuracy of this approach when comparisons were made between atmospheric deposition rates and accumulation rates by peat bogs, and discrepancies were found (Le Roux et al., 2005). Therefore, it is necessary to understand peat diagenetic processes in order to verify that the investigated elements are permanently immobilized in the peat and do not migrate after decomposition (Benoit et al., 1998; Weiss et al., 1999; Ukonmaanaho et al., 2004). Meanwhile, validated sample preparation methods and certified standards reference material are still required to ensure high analytical accuracy and good data quality.

2.2.3.2 Sediment cores

In the absence of historic deposition measurements, undisturbed dated sediment cores may also act as useful geochemical archives and provide information on the atmospheric loading of heavy metals through time. Sediments are regarded as one of the major endpoints or sinks of many anthropogenic trace elements (Blais and Kalff, 1993; Yang *et al.*, 2002; Ip *et al.*, 2004). However, deep sea sediments are not normally used to assess short-term variations of atmospheric deposition due to low accumulation rates. Coastal or estuary sediments accumulate elements much faster than deep sea sediments, thus they are widely sampled to establish regional historical trends. Nevertheless, the possible biological or geological perturbations may have significant influences on metal distributions (Weiss *et al.*, 1999).

Comparatively, sediments in the freshwater ecosystems (*e.g.* lake sediments) are more widely and commonly used to analyze long-term accumulation history of pollutants including heavy metals. This technology has been well developed in Europe (Brannvall *et al.*, 1999; Renberg *et al.*, 2002; Yang and Rose, 2005; Shotbolt *et al.*, 2006) and North America (Blais and Kalff, 1993; Jackson *et al.*, 2004) to investigate atmospheric inputs of heavy metals into lake ecosystems.

Lake sediments can accumulate heavy metals by diverse pathways, of both

natural and anthropogenic origins, such as natural weathering of bedrocks, catchment inwash, atmospheric deposition (Renberg et al., 2002). Thus, concentration profiles (vertical distribution) of heavy metals in sediment cores not only reflect atmospheric inputs of the pollutants, but also demonstrate historical environmental changes (e.g. past drainage basin or land-use histories). Metals of different origins are not readily separable in the sediments. Among various types of anthropogenic sources, atmospheric deposition is considered as the major contributor for most of the trace elements of interest in case of a remote lake without direct point sources (Jackson et al., 2004; Han et al., 2007). In order to assess the significance of atmospheric deposition, it is necessary to distinguish non-geogenic components from geogenic trace element background. A comparison between measured concentrations and baseline values could result in clear pictures of historical changes of atmospheric inputs. Alternatively, this can also be achieved by using elemental normalizer to evaluate the pollution magnitude. Normally, Ti, Zr, loss on ignition (LOI), Si, Al and Fe are selected as the normalizers. It is also a general practice to identify different sources of air pollutants. Pb in terms of specific isotopic fingerprints can be used as a possible pollution tracer element to assess origins of heavy metals (Yang and Rose, 2005).

In most cases, the studied freshwater is located in remote areas, which is only for recreation use and not exposed industrial point sources of pollutants. It is thus assumed to receive anthropogenic pollutants mainly from atmospheric dry or wet deposition (Yang *et al.*, 2002). The obtained sediment records thus provide information on the significance of long-range atmospheric transport of anthropogenic heavy metals to the aqueous ecosystem (Fitzgerald *et al.*, 1998).

Dating of sediment cores by radiometric techniques can be performed by measuring independent chronological markers, among which ²¹⁰Pb, ¹³⁷Cs and ¹⁴C are most frequently used (Eby, 2004). Hence, historical trends of atmospheric deposition can be clearly demonstrated with the combination of accurate chronology along the sediment core and concentration profiles of anthropogenic heavy metals. ¹⁴C has a long half-life of 5.7×10^3 years, thus not accurate for recent deposits. ²²⁸Th has very short half-life (1.91 years), which makes it adequate enough for dating sediments accumulated over the last decade or so. Dating by ²¹⁰Pb is most widely used for recent deposits of age around 100 years with the half-life of 22.3 years (Livett, 1988). However, one major disadvantage of this technique is that ²¹⁰Pb may be subject to remobilization, possibly leading to misestimate of ombrogenic ²¹⁰Pb.

Since sediments are also the potential future pollutant sources to water body and aquatic biota, it is important to investigate geochemical phases of trace elements in sediments and evaluate their bioavailabilities. The analysis is vital to the fate of heavy metals in the aquatic ecosystem and the subsequent impacts on the environment. Tessier *et al.* (1979) suggested a standard approach 'sequential

extraction analysis' to determine five different geochemical phases of heavy metals in the sediments, namely 1) exchangeable; 2) bound to carbonates or specially adsorbed; 3) bound to iron and manganese oxides; 4) bound to organic matter or sulphide; 5) residual.

However, sediment records are sometimes complicated and confounded by additional influencing factors. With much contaminated catchment soil inwash, sediment pollution level might be enhanced. On the other hand, the pollution signal may be diluted when less or no contaminated soils are released into the lake (Yang and Rose, 2005). Moreover, possible postdepositional remobilization processes (*e.g.* resuspension and bioturbation) might occur as a result of significant changes in pH or other geochemical conditions (Weiss *et al.*, 1999; Jackson *et al.*, 2004). Therefore, great attention should be paid during the whole process, from the selection of sampling locations, the sampling of sediment cores, sample preparations, to the final analyzing processes.

In case suitable lakes are not available around the area in question, sediments in reservoirs could be a possible alternative. Similar to the lake sediments, reservoir sediments are also important sinks for both catchment and atmospheric derived pollutants and might as well act as environmental archives for temporal changes of these inputs. However, while the use of lake sediments for assessing historical variations has been well established, reservoirs are less commonly used and even
considered unsuitable due to the lack of understanding of the sedimentation processes involved (Shotbolt *et al.*, 2005).

Since reservoir sediments have the potential to provide additional information to reconstruct historical atmospheric deposition records of pollutants, many investigations have been conducted to examine the practicability of reservoir sediments (Park and Curtis, 1997; Van Metre and Callender, 1997; Shotbolt *et al.*, 2005; Yeloff *et al.*, 2005; Shotbolt *et al.*, 2006). It was reported that the major problem associated with reservoir sedimentary records is the uncertainty of post-depositional sediment redistribution. In case of water-level drawdown, particularly coinciding with a storm event, significant sediment redistribution might be caused, and the area of undisturbed sediments within the reservoir might also be reduced. Although it is believed that deep-water sediments are less vulnerable to water-level fluctuations, the existence of undisturbed and continuously deposited sediment cores is also considered critical to reconstruct past pollution trends. In addition, sediment distribution patterns are not readily known with rather complex bathymetry (Shotbolt *et al.*, 2006).

Nevertheless, reservoirs have much higher sedimentation rates due to their riverine hydrodynamic nature and large catchment/reservoir area ratio. Therefore, they have the advantage to give high-resolution temporal records of historical pollution inputs and minimize possible diagenesis process within sediment cores (Yeloff *et al.*, 2005). The reservoir sediments also have the potential to provide historical records of atmospheric inputs and are regarded as valuable sources for further exploitations. It is generally suggested that assessment of sediment distribution should be conducted prior to site selection and sediment core collection, so that sediment cores of minimal stratigraphic disturbance could be sampled.

In summary, each approach mentioned above has its merits and disadvantages. Aerosols and mosses are most commonly used to provide current information on atmospheric heavy metal pollution. The active air sampling is the most effective and direct approach to monitor the airborne heavy metals except its infeasibility in remote sites without pumps and power sources. Passive air sampling, as a complementary method to the collection of active samples, allows chemical deposition and accumulation merely via natural diffusion. However, it is time-consuming and sensitive to changes of meteorological conditions. As an effective biomonitor of airborne heavy metals, mosses have been widely used as many moss species are geographically widespread. This technique is also easy to handle and costs less, and is especially effective for surveys over large areas or even nations. The major disadvantages of this approach are inadequacy in comparison between different species and vulnerability to natural variability in ambient climate conditions. On the other hand, the naturally occurring peat bogs and aqueous sediments can be used as historical archives to monitor long-term

atmospheric deposition of heavy metals. However, they can only be applied in case the anthropogenic heavy metals are dominantly derived from atmospheric deposition. Moreover, they must be subject to minimal post-depositional substances to preserve the record. Therefore, a multi-media approach, including sampling of both mosses and sediment cores, was adopted in the present study to monitor both the current situation and long-term variations of atmospheric heavy metal deposition in remote areas of the South China coastal region. Sediment cores were collected in remote reservoirs without direct point sources of anthropogenic pollutants and were regarded adequate to provide historical records of atmospheric heavy metal inputs. As for monitoring the current situation, sampling of mosses is considered more appropriate than aerosol samples in remote areas.

2.3 Approaches for Pollution Source Identification

2.3.1 Lead isotopic analysis

Lead isotopes have been widely used to identify sources of atmospheric pollution and proved to be particularly useful in a variety of different environmental samples, including sediments (Farmer *et al.*, 1996; Kober *et al.*, 1999; Renberg *et al.*, 2002; Ip *et al.*, 2004; Jackson *et al.*, 2004), aerosols (Mukai *et al.*, 1993; Zhu *et al.*, 2001; Wong *et al.*, 2003; Bellis *et al.*, 2005; Witt *et al.*, 2006; Lee *et al.*, 2007), mosses (Farmer *et al.*, 2002; Lee *et al.*, 2005). Lead has four naturally occurring long-lived isotopes, namely ²⁰⁴Pb, ²⁰⁶Pb, ²⁰⁷Pb, and ²⁰⁸Pb. However, only ²⁰⁴Pb is the stable isotope of lead, while ²⁰⁶Pb, ²⁰⁷Pb, and ²⁰⁸Pb are radiogenic progenies of ²³⁸U, ²³⁵U, and ²³²Th respectively. Therefore, the abundance of the last three isotopes can slowly change over long period and ratios among the four isotopes are also varying gradually. Each lead containing ore has specific lead isotopic ratios, which are dependent on the age and initial geological compositions of its parent rock (e.g. amount of co-existing U and Th). Normally, older lead ores experience less radioactive decay before separation from the parent rocks, thus they tend to have low ²⁰⁶Pb/²⁰⁷Pb ratio; vise versa (Witt et al., 2006). Likewise, other natural materials, such as coal, also have unique values of lead isotopic ratios. These isotopic ratios are not subject to changes afterwards in the environment or during industrial productions, as no more fractionation takes place by chemical or biological processes. Lead in the atmosphere can be derived from many different origins, such as dust from natural ores, lead additives in gasoline, coal combustion, as well as various types of industrial emissions (Siegel, 2002; Bradl, 2005). Hence, lead isotopic ratios of different origins can be combined in various ways to result in other lead isotopic ratios. With comparisons between measured ratios of samples and known ratios of possible surrounding sources, origins of the pollutants could be identified. Therefore, lead is regarded as a useful tracer of environmental pollution since its isotopic compositions are specific to different emission sources.

2.3.2 Statistical analyses

Metals can occur naturally in silt and clay bearing minerals of terrestrial and marine geological deposits (Schiff and Weisberg, 1999). The natural occurrence of metals in the environment can complicate the evaluation of heavy metal enrichments contributed by anthropogenic sources, as the background metal concentrations might vary widely due to different parent materials and deposit properties (e.g. particle size) (Schropp et al., 1990; Liu et al., 2003). Therefore, elevated concentrations of heavy metals in aquatic sediments may not necessarily reflect anthropogenic contaminations. In order to distinguish the natural components, geochemical normalization of trace metal concentrations to a reference element has been commonly adopted to account for the natural mineralogical and granulometric variations and assess the presence and magnitude of anthropogenic enrichments of heavy metals (Weisberg et al., 2000). The reference element (normalizer) is normally a fundamental or conservative tracer of the natural metal-bound phases, which is expected to have a uniform flux from natural sources and not sensitive to inputs from anthropogenic inputs (Liu *et al.*, 2003). They should not be vulnerable to physicochemical processes, such as reduction/oxidation, adsorption/desorption, or other diagenetic processes. The most widely used normalizers mainly include Al, Fe, Li, Sc, Cs, rare earth elements, radioisotope tracers, total organic carbon and the grain size (Van Metre

& Callender, 1997; Schiff and Weisberg, 1999).

Al has high natural concentration and minimal anthropogenic contamination. It is a structural element of clays. As a central component of clay (fine particles), Al can serve as a measure of clay content of sediments, with which bulk of trace elements are associated. Since trace metals tent to be absorbed on the surfaces of particles, their concentrations naturally increase as particle sizes decrease. However, Al has a much lower recovery rates for this study due to the adopted digestion method.

Fe is a clay mineral indicator element as an alternative to Al. It is not a matrix element like Al, but is like trace elements in being associated with surfaces. Fe geochemistry is similar to that of many trace metals both in oxic and anoixic environments. The direct association with trace elements and the fact that its natural concentrations in sediments are more uniform than Al, make it a promising normalizer. However, Fe has a great reduction potential in anaerobic sediments and may lead to an underestimation of the magnitude in anthropogenic enrichment of trace metals. Coprecipitation with Fe/Mn oxides is an important mechanism influencing the accumulation, speciation and vertical distribution of heavy metals in sediments. Fe²⁺ and Mn²⁺ tend to diffuse upwards to the redox boundary and be oxidized to Fe³⁺ and Mn⁴⁺, which are efficient scavengers of released heavy metals. The reduction/oxidation process effectively concentrates

iron in surface sediments at or near the sediment-water interface, and is mostly associated with peaks of heavy metal concentrations.

TOC serves as a matrix on particle surfaces for complex formation. TOC itself, however, is subject to considerable augmentation by human activity and is often a contamination. It is usually difficult to assess the background concentration.

Li is a constituent of fine-grained material of sediments, such as primary micas and ferromagnesium minerals and secondary clay minerals, and Li concentrations closely associated with clay contents. Like Al, Li has a low recovery rate in the present study.

It is assumed that the concentrations of natural occurring trace metals generally vary in proportion to that of the normalizer. The relationships between trace metals and the normalizers thus can be plotted to identify the presence of possible contaminations. A possible best normalizer for a particular metal element is usually the highest geochemically sensible tracer, referring the strongest correlator (Yang and Rose, 2005). The Pearson correlation coefficients between the metals and potential reference elements are usually calculated to identify the proper normalizers. Baseline relationships between the metal of interest and its most appropriate normalizer can be derived by establishing a simple linear regression model based on several selected uncontaminated samples in the study area. A close correlation between the major elements and trace metals is normally expected for uncontaminated sediments. The 95% prediction intervals are then derived from the linear regression models for the purpose of comparing individual samples to the baseline relationships. The linear regression lines and 95% prediction limits are then superimposed on scatter plots of heavy metals and the respective normalizers. It is suggested that any samples above the upper 95% prediction limit are probably contaminated by heavy metals from anthropogenic sources, while data points within the prediction limits are originated from natural sources. Moreover, the correlation coefficients between trace elements may provide additional information on source identification. Two closely correlated elements are supposed to have the same origin.

The Principal Component Analysis (PCA) together with the Pearson correlations is another useful multivariate analytical tool that is frequently performed to identify the anthropogenic sources of enriched heavy metals (Singh *et al.*, 2005; Wu *et al.*, 2007). It evaluates the inter-correlations between the variables in the original data set and enables a data reduction into a few independent underlying factors that account for the majority of total variance and allow easier interpretation (DelValls *et al.*, 1998). The extracted principal components (PC) are those whose eigenvalue is higher than 1 (Loska and Wiechula, 2003). As heavy metal concentrations normally varies by different orders of magnitude, PCA can be applied to the correlation matrix and normalize each variable to unit variance (Han *et al.*, 2006). Varimax normalized rotation can be further conducted to maximize the variances of factor loadings and eliminate invalid components, which makes the results more easily interpretable (Loska and Wiechula, 2003). Each PC provides information on loadings of the metal elements (variables) indicating the relative contribution of each element. The dominance of metal elements in each component can be identified and origins of heavy metals can thus be implicated.

2.3.3 Air back-trajectory analysis

In order to identify the origins of specific pollutants, it is necessary to examine the exact paths of air mass transport. The air mass back trajectories have thus been frequently used to investigate the potential sources of contaminants (Siefert *et al.*, 1999; Bellis *et al.*, 2005; Witt *et al.*, 2006; Lee *et al.*, 2007). Since air mass trajectory models may include many possible source areas, it is normally used in combination with other source apportioning methods, such as isotopic signatures. The analysis is generally conducted with the aid of computer models. It was found that the type and concentration of air-borne pollutants at certain sampling locations vary in accordance with wind directions. In addition, air mass trajectory could change as a function of time, even the general wind pattern is known (Eby, 2004). Thus, many important components, such as meteorological data, should be considered and included in the computer model to give accurate calculations. After obtaining plots of air mass back trajectories, it is possible to identify places that air mass has traveled across and thus locate possible sources of the air pollutants. However, the time range for back trajectory calculation should be carefully considered in order to get reliable results and solid interpretations.

2.4 Atmospheric Heavy Metal Pollution in South China & Worldwide

As mentioned in Section 2.1.3, the ever intensive human activities have brought abundant potentially hazardous heavy metals into the environment since the onset of industrialization worldwide, which might pose adverse impacts on the ecosystems. In the past several decades, numerous studies have been conducted to assess atmospheric heavy metal contamination and evaluate the significance of atmospheric depositions of heavy metals, particularly in European countries (Guieu et al., 1997; Injuk et al., 1998; Motelay-Massei et al., 2005; Rossini et al., 2005; Lavilla et al., 2006) and North America (Golomb et al., 1997; Sweet et al., 1998; Riget et al., 2000; Schilling and Lehman, 2002). Increasing evidence has revealed that the air quality in not only urban areas (Wong et al. 2003; Tasdemir et al., 2006) but also remote areas (Lee et al., 2005; Yang and Rose, 2005) has been deteriorated seriously. Atmospheric emissions and depositions of anthropogenic heavy metals have been regarded as significant contributors to the accumulation and enrichment of toxic trace metals in the surface environment, such as lake sediments (Dauvalter et al., 1994; Kober et al., 1999) and surface soils (Steinnes et al., 1993; Lee et al., 2006).

Many attempts have been made to determine the annual atmospheric fluxes by assessing passive bulk deposition samples. In Table 2.6, the significance of atmospheric heavy metal inputs in several study areas are listed and compared. As the data indicate, the atmospheric deposition fluxes of heavy metals in urban or industrial areas are generally greater than the amounts in rural or relatively remote areas. By comparison among different regions, it can also be noticed that the atmospheric deposition fluxes of anthropogenic heavy metal in the South China coastal region have been significantly enhanced, which are comparable or even stronger than some industrial areas in Europe and North America, especially for Cr, Ni and Pb. The South China coastal region, particularly the Pearl River Delta region, is one of the most rapidly economically developing and densely populated areas in China. The economic reform since the late 1970s has brought about enormous economic bloom with intensive industrialization and urbanization. The potential environmental burdens imposed by the drastic development in the past three decades are causing increasing concerns. Recent studies regarding the air pollution in South China revealed that not only the atmosphere in this region has been seriously deteriorated, but also the surface environment (e.g. urban soils, surface waters) has been inevitably impaired as a consequence of atmospheric deposition (Wong et al, 2003; Wang et al., 2005; Hagler et al., 2006; Choi et al., 2008; Fang et al., 2008). The contamination by anthropogenic Cu, Pb and Zn in the PRD region needs particular attention (Fang et al., 1999; Fang et al., 2000; Zhu et al., 2001; Lee et al., 2007; Zhang et al., 2007).

The atmospheric heavy metal pollution has also been widely investigated by

analysis of terrestrial mosses to assess the current atmospheric fluxes of heavy metals. The most commonly used moss species include Hylocomium splendens and Pleurozium schreberi in European countries (Zechmeister, 1995; Berg and Steinnes, 1997; Harmens et al., 2004). Thuidium delicatulum in southeastern US (Schilling and Lehman, 2002), and Hypnum plumaeform in Asian countries (Lee et al., 2005). This technology has been well developed especially in western countries, where moss surveys over several decades were conducted to assess the temporal deposition trends and to identify the potential sources (Kubin and Lippo, 1996; Rosman et al., 1998; Harmens et al, 2004; Ruhling and Tyler, 2004; Shotbolt et al., 2007). The findings revealed that the highest concentrations of heavy metals in mosses by atmospheric depositions occurred in the 19th century (Shotbolt et al., 2007) and there was a continuous decrease of metal concentrations since the end of 1970s due to the reduction of industrial emissions (Ruhling and Tyler, 2004). The decline of atmospheric lead levels was particularly evident since the ban of leaded petrol in the 1960s (Berg and Steinnes, 1997; Farmer et al., 2002). The long-range trans-boundary transport of air pollutants from industrial sites might account for the elevated airborne heavy metal concentrations in remote areas. The studies conducted in China are quite limited. Past investigations were mostly confined to using the moss technology to monitor the dry deposition fluxes of heavy metals near some emission sources (Xiao et al., 1998; Tan et al., 2000; Bi et al., 2006) or in relatively remote areas (Lee et al., 2005), and to identify potential sources. Limited studies focus on

examining the temporal variations of airborne heavy metal concentrations. One recent study by Cao *et al.* (2008) reported that the levels of heavy metals in the atmosphere in Shanghai were distinctly elevated from 1965 to 2005, especially after the 1980s.

It has been widely recognized that aquatic sediment cores collected from lakes or reservoirs could preserve a historical record of heavy metal concentrations with accurate chronological dating (Renberg et al., 2002; Yang and Rose, 2005; Shotbolt et al., 2006). In case of a freshwater system without direct point sources, the obtained sediment records are considered suitable to provide information on the atmospheric transport and deposition of heavy metals to the aqueous ecosystem at regional scales (Fitzgerald et al., 1998). Regarding this, sediment profiles of heavy metal concentrations have been widely investigated to monitor the long-term atmospheric inputs of heavy metals in both urban and remote areas in Europe (Foster et al., 1991; Fitzgerald et al., 1998; Brannvall et al., 1999; Renberg et al., 2001; Yang et al., 2002; Shotbolt et al., 2006) and North America (Chillrud et al., 1999; Belzile et al., 2004; Jackson et al., 2004). The sediment records clearly showed that the atmospheric emissions and deposition fluxes of anthropogenic heavy metals have been markedly increased since the onset of the industrialization period. However, some studies revealed that the Pb concentrations appeared to have a decreasing trend in recent sediments due to reduced Pb emissions from vehicle exhausts, though they were still much higher

than the background values (Renberg et al., 2002). Nevertheless, the recovery signal of anthropogenic Pb concentrations in sediments was not recorded at some other sampling sites (Kober et al., 1999; Shotbolt et al., 2006; Yang et al., 2007). This might be attributed to the fact that anthropogenically derived Pb initially deposited onto drainage basin soils keeps entering the freshwater system via erosion and leaching processes. The catchment soils began to act as a contributing source of heavy metals instead of a major depository sink. While the temporal trends of atmospheric heavy metal depositions have been well established in western countries, the investigations on monitoring of atmospheric heavy metal pollution by aquatic sediments in China are still quite scarce (Han et al., 2007). The sediment profiles of metal concentrations have been mostly used to assess potential contamination of sediments with direct discharges of municipal and industrial wastewater (Huh and Chen, 1998; Liu et al., 2003; Feng et al., 2004; Ip et al., 2004). Several recent studies suggested that the sediment levels of heavy metal concentrations in coastal and estuarial sediments are comparable or even higher than the values reported in urban areas of the western countries. Although the enrichment of heavy metals is dominantly ascribed to wastes discharged from point sources, the increasing atmospheric input of anthropogenic pollutants is also considered as a potential source. Regarding the ever-increasing emissions of air pollutants in China, there is an urgent need to quantify the deposition fluxes of airborne heavy metals and establish long-term records of heavy metal concentrations in aquatic sediments from lakes or

reservoirs without direct point sources.

	Location	Period	Cd	Cr	Cu	Ni	Pb	Zn	Remarks
Europe	Le Havre, France ^a	2001 - 2002	0.36		9.4	4.1	18.4	1172	Industrialized area
	NDG, France ^a	2001	0.23		5.2	2.0	4.8	59	Urban area
	Rouen, France ^a	2001 - 2002	0.19		2.8	0.44	3.0	27	Urban city center
	Evreux, France ^a	2001 - 2002	0.20		7.5	0.34	1.7	38	Semi-rural area
	Paris, France ^a	2001 - 2002	0.24		6.0	0.62	4.2	30	Very urbanized and industrialized area
	Ligurian Sea ^b	1997 – 1998	0.06	0.62	1.28	1.10	1.20	41.2	Coastal area
	North Sea ^c	1993 – 1994		2.43	1.30	1.21	3.69	6.50	Coastal area
	Irish Sea ^d	1993 – 1994			2.60		1.62		Coastal area
	Mediterranean Coast ^e	1988 – 1993	0.31 - 1.3	2.5	2.7	0.9	3.2	2.0	Coastal area
	Lagoon of Venice ^f	1998 - 1999	0.15		4.3	1.9	3.6	29	Coastal area
North America	Lake Superior ^g	1993 – 1994	0.46	0.22	3.1		1.5	8.8	Away from sources
	Lake Michigan ^g	1993 – 1994	0.45	0.20	1.9		1.6	6.0	Away from sources
	Lake Erie ^g	1993 – 1994	0.49	1.10	4.2		1.8	17.0	Near industrial and urban emissions
	Massachusetts Bay h	1992 - 1993	0.45	2.7	3.5	7.2	2.7	7.8	No point sources
Asia	Pearl River Delta ⁱ	2001 - 2002	0.07	6.43	18.6	8.35	12.7	104	Urban/suburban/rural without point sources
	Xinghua Bay ^j	2004 - 2005	0.08	11.63	2.72	7.81	3.83	14.39	Coastal area
	Quanzhou Bay ^k	2004 - 2005	0.08	5.62	2.79	6.28	3.5	11.47	Coastal area

Table 2.6 Bulk atmospheric deposition fluxes $(mg/m^2/yr)$ for heavy metals obtained from various studies worldwide

a. Motelay-Massei *et al.*, 2005; b. Sandroni and Migon (2002); c. Injuk *et al.*, 1998; d. Williams *et al.*, 1998; e. Guieu *et al.*, 1997; f. Rossini *et al.*, 2005; g. Sweet *et al.*, 1998; h. Golomb *et al.*, 1997; i. Wong *et al.*, 2003; j. Wu *et al.*, 2006(a); k. Wu *et al.*, 2006(b).

CHAPTER 3 – STUDY AREA & METHODOLOGY

In this chapter, detailed descriptions of the project study area are presented, including geographical locations, geochemical characteristics, and historical economic and industrial developments. In addition, the sampling procedures and analytical methods adopted during the study are also illustrated.

3.1 Study Area & Geochemical Characteristics

The present study was conducted in three major southeastern provinces along the South China coastline, including Fujian, Guangdong, and Hainan (Fig. 3.1). The South China coastal region is one of the most rapidly developing and densely populated areas in East Asia. Enormous economic development has taken place since the economic reform in the late 1970s. However, most of the economic achievements have been made at the cost of environmental degradation in the past several decades.



Figure 3.1 Map of the study area (colored in green)

3.1.1 Fujian Province

Fujian Province is located on the southeast coast of China $(23^{\circ}31' \sim 28^{\circ}18'N;$ 115°50'~120°43'E), lying along the East China Sea to the east, South China Sea to the south, and Taiwan Strait to the southeast. It has a total land area of 121,400 km² and supports a population of 35.81 millions (Fujian Gov., 2008). Over 80% of Fujian's land area is mountainous with two major mountain belts striding over the province longitudinally in the west and the center area. Plain areas are generally located at coastal regions, mainly Fuzhou Plain, Putian Plain, Quanzhou Plain, and Zhangzhou Plain, which are the most economically developed and urbanized areas in Fujian Province. Fujian has the most complicated and the second longest coastline in China, facilitating the formation of many deep bays as natural harbors. Although it is situated at a typical sub-tropic zone and is sensitive to the impact of East Asia monsoon activities, Fujian has a relatively mild and humid climate with abundant rainfalls. The Mountain Wuyi lying across the northwest part of Fujian prevents the intrusion of dry and cold winter monsoons from North China, while sufficient moisture is brought inland from the ocean in summer. According to Fujian Statistical Yearbook 2008, the annual average temperatures and precipitations are 15.8-22.6 °C and 932.7-2265.2 mm respectively (Fujian Gov., 2008).

Situated at the jointing boundary of the Eurasia Plate and Pacific Ocean Plate, Fujian is characterized by complicated geological formations and dominated by igneous rocks. Granite is the most widely spread rock type in the province, accounting for nearly 1/3 of the total land area, particularly at coastal hills and mesas. The coastal estuary plains and harbor areas are characterized by sedimentary rocks, while the northwest region is mainly composed of metamorphic rocks and igneous rocks. Due to historical intrusions of abundant granites, elements such as U, Th, La, Y, Zr, W, Sn, Mo, Ag, Pb and Zn are greatly enriched in the lithosphere while Fe is relatively lacked (Fujian Bureau of Geology, 2008). As Fujian has plentiful rainfalls with a humid sub-tropic climate, some elements like Sr and Cu are drained away significantly. Table 3.1 summarizes the background elemental concentrations in aqueous sediments in Fujian compared to the national average background levels.

Table 3.1 Background elemental concentrations in Fujian aqueous sediments compared to national background levels (unit: $\mu g/g$, otherwise indicated) (Fujian Bureau of Geology, 2008)

Elements	As	Cd	Со	Cr	Cu	Mn	Ni	
Fujian	3.47	0.15	7.63	22.44	9.78	602	8.48	
Mean								
National	9.08	0.15	12.32	58.45	21.56	678	24.41	
Background								
Flomonts	Pb	Zn	Al_2O_3	Fe ₂ O ₃	CaO	MgO		
Liements			(%)	(%)	(%)	(%)		
Fujian	42.65	78.8	14.42	3.48	0.25	0.45		
Mean								
National	25.96	68.47	12.97	4.44	1.76	1.31		
Background								

Fujian's economy was mainly dependent on farming, forestry, animal husbandry and fishery until the adoption of the economic reform in the 1970s. As Fig. 3.2 indicates, industries have been developing enormously since the mid 1970s, and gradually replaced the dominance of agriculture. According to Fujian Statistical Yearbook 2008, the percentage of industries in the entire economic output is increasing continuously every year and reaches 89.5% in 2007. Nowadays, electronic industry, petrochemical industry and mechanical industry have become the three leading industries in Fujian. With large-scale industrial productions, however, excessive amount of pollutants have been released into the atmosphere. It was reported (Fujian Environment Protection Bereau, 2008) that the annual discharges of soots, and industrial powders/dusts in 2006 amounted to 129,500 tons and 173,000 tons respectively, which exerted potential adverse impacts on the ecosystems. Fig. 3.3 presents the air quality conditions in some major cities of Fujian in 2006 reported by the Fujian Environment Protection Bereau. As the figure shows, the air pollution problem is particularly serious in Sanming and Longyan where metallurgical and mining industrial activities are intensive. Since air pollutants might travel over vast areas and finally deposit far away from the emission sources, it is thus important to investigate the potential atmospheric outflows and inputs of anthropogenic pollutants to the surrounding areas.



Figure 3.2 Gross output values of agriculture and industry from 1952 to 2006 (Fujian Bureau of Statistics, 2008)



Figure 3.3 Air quality in several major cities of Fujian Province in 2006 (Gazette 2006 of Fujian Environment Protection Bureau, 2008). Grade I represents the best air quality, while Grade III represents the worst (National Ambient Air Quality Standard).

3.1.2 Guangdong Province

Bordering Fujian on the northeast, Guangdong Province $(20^{\circ}13' \sim 25^{\circ}31'\text{N};$ $109^{\circ}39' \sim 117^{\circ}19'\text{E})$ is situated at low-lying areas along the South China coastline towards the South China Sea. It has a total land area of 179,800 km² and owns the longest coastline in China. The Tropic of Cancer runs through the province, which results in a subtropical climate in the north part while a tropic climate in the south part. Generally, it is warm and humid all year round with annual average temperatures and precipitations of 19-24 °C and 1300-2500 mm respectively (Guangdong Gov. 2008). Mountains in Guangdong are mostly located in the northwest region, most of which run in a northeast-to-southwest direction. Similar to Fujian, granite is the most common rock type in Guangdong as well, followed by sandstones and metamorphic rocks. Moreover, certain amount of limestone is scattered in the northwest area.

Guangdong is one of the most densely populated provinces in China and supports a population of almost 94.5 millions (Guangdong Gov., 2008), especially concentrated in the Pearl River Delta Region. In 2006, the town population in nine major cities (i.e. Guangzhou, Shenzhen, Zhuhai, Foshan, Jiangmen, Dongguan, Zhongshan, Huizhou, and Zhaoqing) in the PRD region amounted to 62.84% of the total town population in Guangdong Province. The Pearl River Delta is the second largest delta in China, where three inland rivers East River, Pearl River and West River converge and discharge into the South China Sea. Bordering on the east side of the Pearl River Estuary, Hong Kong is a well known international metropolis which is highly developed and urbanized. With a total land area of 1,104 km², Hong Kong supports a population of almost 6.92 millions (HKSAR Gov., 2008). In the past several decades, a large fraction of Hong Kong's heavy industries have been relocated to Guangdong Province, which has stimulated its economic bloom. Figure 3.4 depicts the industrial development in Guangdong since the economic reform in 1978 (Guangdong Bureau of Statistics, 2008). The gross output values of industries have been rising exponentially ever since. Nowadays, the PRD region has developed as one

of the leading economic centers in China, with massive manufacturing capacities. Electronic industry, mechanical industry, petrochemical industry, and automobile manufacturing industry are growing into the main heavy industry sectors in the delta region.



Figure 3.4 Gross output values of Guangdong industrial productions since 1978 (Guangdong Bureau of Statistics, 2008).

However, pollution problems become increasingly serious with extensive and intensive industrial activities, especially in recent years (see Fig. 3.5). The air quality, particularly over the PRD region, has been severely deteriorated. Traffic exhausts, intensive industrial activities and emissions from power plants have been regarded as the major anthropogenic sources of air pollutants in the PRD region (HK EPD, 2008). According to the record of Regional Air Quality Indexes (RAQI) released by Guangdong and HK Governments during Jan-Jun 2007, nearly 30% of the data exceeded Grade II of National Ambient Air Quality Standard (Fig. 3.6). In addition, Guangdong is located in the East Asian monsoon zone and is vulnerable to the impacts of monsoon activities, which might influence the transport of air pollutants and lead to variations in spatial and temporal distributions. Numerous evidences revealed that the adverse impacts of air pollution concern the entire PRD region rather than merely the local area. Moving and re-allocating heavy industries to remote areas thus can never be a long-term solution. Maximum concentrations of air pollutants usually occur in winter, while abundant precipitations brought from the ocean in summer normally result in much better air quality and the higher mixing layer height also facilitates the dispersion of air pollutants.



Figure 3.5 Increasing amount of total industrial exhaust emission into the atmosphere during 2000-2006 in Guangdong Province (Guangdong EPD, 2008)



Figure 3.6 Statistics of Regional Air Quality Indexes recorded during January – June, 2007. (a) Percentages of each air quality grade; (b) Spatial distribution of mean air quality indexes. (Guangdong EPD, 2008)

3.1.3 Hainan Province

Being at the southmost end of China, Hainan Province is located off the southern coastline of China, separated from the Leizhou Peninsula (Guangdong Province) by Qiongzhou Strait. The total land area of Hainan Province is $35,400 \text{ km}^2$, dominated by Hainan Island of $33,900 \text{ km}^2$. Hainan Island ($18^\circ 10' \sim 20^\circ 10' \text{N}$; $108^\circ 37' \sim 111^\circ 03' \text{E}$) is located at the north margin of the tropical zone, characterized by a moist monsoonal climate. The annual average temperature ranges from 22 °C to 26 °C. The island is usually in the path of typhoons, which bring about numerous precipitations from the South China Sea. The annual average precipitations are around 1111-2150 mm, particularly concentrated during May-October (Hainan Gov., 2008). Topographically, Hainan Island is elevated in the central region and gradually leveled off peripherally. Most of the rivers are originated from mountains in the central area and flow radially in

different directions. Igneous rocks are widely spread in Hainan Island due to intensive geotectonic and diastrophic activities, accounting for over 50% of the total land area. Ferrallitic soils or ferruginous soils are the dominant soil types on the island under the humid and hot environmental condition (Zhao *et al.*, 2005).

Compared to the previous two provinces, Hainan is much less densely populated. Till the end of 2004, the population in Hainan Province is about 8.18 millions (Hainan Gov., 2008). Historically, agriculture had played the major role in the economic development of the island in early years. It still kept the dominance (over 50%) in the whole economic gross output even in the early 1990s. Recently, enormous economic development, especially industries and tourism, has occurred resulting in great economic structure reform. According to the government yearbook (Hainan Gov., 2008), the ratios between three main productions were 36:26:38 in 2005. Petrochemical industry, automobile industry, medicines and drugs, as well as paper production industry have gradually become the pillar industries. For many years, Hainan Province is among the provinces with best air quality in China. The air quality in most cities of Hainan meets Grade I of National Air Quality Standard. As the government reported (Hainan Gov., 2008), SO₂, NO₂ and suspended particulates only accounted for 9%, 27%, 21% of the national average levels in 2002. However, there appears an increasing trend for the concentration of suspended particulates, especially in the western areas, such as Haikou, Sanya, Danzhou and Qionghai (Hainan Gov.,

2008). Moreover, the lead-free petroleum was still not widely introduced till the end of the twenties century. Considering the ever-increasing industrial productions, the air pollution problem cannot be neglected any longer, and the daily air quality should be monitored for effective environmental management.

All the three provinces are situated in typical monsoon zone and sensitive to monsoon fluctuations, which might facilitate the long-range transport of air pollutants from inland area of the continental Asia. Han *et al.* (2007) found a strong association between variations of heavy metal concentrations and climate changes in accordance with East Asia monsoons. Hsu *et al.* (2005) and Lee *et al.* (2007) recently provided evidence on possible long-range transport of air pollutants from northern China by monsoon activities. Therefore, the South China coastal region might be subject to air pollutants released from both local and regional sources via the long-range transport.

3.2 Sampling Locations & Sample Collection

The sampling work was conducted during September – November of 2006, including sampling of reservoir sediment cores, catchment surface soils, and mosses. The sampling locations were selected at four drinking water reservoirs in the remote areas of the South China coastal region, namely Xiamen Tingxi Reservoir in Fujian Province, Heyuan Xinfengjiang Reservoir and Maoming Luokeng Reservoir in Guangdong Province, and Danzhou Songtao Reservoir in

Hainan Province (see Fig. 3.7).



Figure 3.7 Map of sampling sites along the South China coastline

3.2.1 Descriptions of reservoirs

These selected reservoirs were all constructed between the 1950s – 1960s. Sediments are thus supposed to have a historical accumulation of around 50 years. All the sampling sites are far away from intensive industrial areas, avoiding direct inputs of anthropogenic pollutants. It is therefore reasonable to assume the major contribution of trace metal concentrations in the sediments is from the atmospheric deposition. In addition, the reservoirs are generally free from significant human activities, facilitating the formation of laminated sediments under natural conditions without much disturbance.

3.2.1.1 Tingxi Reservoir

Tingxi Reservoir (Fig. 3.8) is located in Tong'an of Xiamen, Fujian Province, about 7 km away from the urban area. It is an important source of fresh water and serves as one of the four major drinking-water supplies in Xiamen. Tingxi Reservoir was constructed in 1956 and expanded in 1966 to increase its capacity. At present, the total water storage volume amounts to 48.45 million m³ with effective reservoir volume of 37.63 million m³ (Yu et al., 2006). The catchment area is about 100.8 km². The main function of Tingxi Reservoir is to provide water for irrigation in the districts of Tong'an and Xiang'an. It also plays an essential role in floods prevention, drinking water supply, electricity generation, as well as aquatics breeding (Lv, 2007). Topographically, Tingxi Reservoir is elevated in the north region and gradually descended southwards. Most of the inhabitants living in the upper reach are involved in agricultural activities. Due to the increasing scale of agriculture productions and livestock raising activities, the water quality of the reservoir has been greatly deteriorated in recent years. In addition, soil erosion problem becomes increasing serious caused by excessive felling of trees in the upriver and decreasing forest cover in the surrounding area (Lv, 2007).



Figure 3.8 Sampling locations of sediment cores at Tingxi Reservoir

3.2.1.2 Xinfengjiang Reservoir

Xinfengjiang Reservoir (see Fig. 3.9) is the largest reservoir in South China with water area of 370 km² and storage volume of 13.9 billion m³ (Ye, 2000). It is located in a remote mountainous area of Guangdong Province, at a distance of around 150 km away from the capital city Guangzhou. The reservoir was constructed during 1958-1962 by damming up Xinfengjiang River downstream at Yaposhan Valley. The total catchment area amounts to 5,734 km² with the forest cover around 1,100 km² (Chen, 1999). The dam of the reservoir is 124 m in height and 440 m in length. In normal conditions, the characteristic water level (*i.e.* altitude) of the reservoir is around 116 m (Guangdong Bureau of Hydrology, 2008). As a multifunctional reservoir, Xinfengjiang Reservoir has played significant roles in electricity generation (annual yield up to 0.9 billion kw-h),

irrigation, shipping, flood prevention, and water supply. It is one of the major tributaries of East River, bearing a huge responsibility of providing high-quality water for people living in the PRD region.

The East River basin was originally dominated by agricultural production, and the natural environment and ecosystems were well conserved. During the late 1970s and mid 1980s, however, rapid industrialization and urbanization occurred firstly in the downstream PRD region and then gradually expanded to the whole East River region. The economy has boosted at the cost of ecological degradation and environmental pollution, potentially endangering the water and air quality at Xinfengjiang Reservoir (Chen, 1999; Ye, 2000; Che, 2004). In 1988, a large cement factory was established beside the reservoir, which produced numerous industrial dusts and powders and inevitably resulted in serious air and water pollutions. Being aware of the seriousness of ecocrisis, Guangdong government took immediate measures and shut down the cement factory in 1996. Nowadays, certain mining activities in the upriver region are still in operation (Ye, 2000). Around 232 ships or boats run across the reservoir every day for tourism or shipping activities. The discharges of untreated domestic sewages by around 40 thousand residents from the surrounding towns (e.g. Xichang, Huilong, Jiantou and Banjiang) are not properly managed. The intensively urbanized PRD region is located just within 200 km from the reservoir, and the air pollutants (e.g. heavy metals) released and transported from the PRD region might have adverse

impacts on the ecosystems at the reservoir.



Figure 3.9 Sampling locations of sediment cores at Xinfengjiang Reservoir

3.2.1.3 Luokeng Reservoir

Luokeng Reservoir, constructed in 1960, is one of the three major reservoirs in western Guangdong Province (see Fig. 3.10). It is originated from Ehuangzhang Mountain in the northeast part of Maoming City, and has a total reservoir storage capacity of 114.7 million m³ and catchment area of 77 km². The normal water level is about 114 m. Luokeng Reservoir is situated in the remote area, at a distance of around 60 km from the downtown region. As it is far away from the industrial zone and surrounded by large mountain ranges, a good air quality is likely to be preserved over the reservoir area.



Figure 3.10 Sampling locations of sediment cores at Luokeng Reservoir

3.2.1.4 Songtao Reservoir

As one of the ten largest reservoirs in China, Songtao Reservoir (see Fig. 3.11) is located in Danzhou, Hainan Province, about 20 km southeast of Danzhou urban area. Construction of the reservoir initiated in 1958 and lasted over 10 years till 1969, with a dam built at Qinzukou Valley, the upper reach of Nandujiang River (Zeng, 2000). The river flows northward into the reservoir, and then turns eastward (Deng *et al.*, 2007). Songtao Reservoir has a total reservoir area of 130.5 km², and water storage capacity of 3,345 million m³. The drainage area of the reservoir amounts to 1,496 km². As a multifunctional reservoir, it is dominantly used for irrigation and also functions for water supply, electricity generation, flood prevention, aquatic breeding and shipping (Yang *et al.*, 2007). It is responsible for irrigation of agricultural lands up to 136,700 hm², and water supply for over two millions inhabitants in Baisha, Danzhou, Lingao, Chengmai, Haikou (Yue *et al.*, 2005). The reservoir drainage region is relatively backward in economy and most of the residents are involved in agricultural activities. With rapid economic development and improvement of people's living conditions in recent years, however, the water and air quality at Songtao Reservoir are gradually endangered by the pollutions from industrial, domestic, agricultural sources, and shipping activities (Yue *et al.*, 2005). Although the overall water quality still reaches Grade II of National Surface Water Quality Standard, some water regions can only meet Grace IV or even Grade V. Most of the existing studies focused on merely the water quality of Songtao Reservoir, while the investigations on air quality over the reservoir area are rather scarce. The results from this present study can thus be used to examine the air quality at the reservoir, as well as the surrounding areas.



Figure 3.11 Sampling locations of sediment cores at Songtao Reservoir
3.2.2 Sample collection

Sediment cores were collected at locations with relatively deeper water depth at each reservoir, using a portable Kajak gravity corer with an automatic clutch and reverse catcher. In Fig. 3.8 - 3.11, the sampling locations of sediment cores are indicated for the four reservoirs respectively. The corer was first released and driven into the sediments by gravity. The sediments were then retained in a PVC tube of 70 mm in its inner diameter. After collection, the sediment cores, except the case of Xinfengjiang Reservoir, were immediately sliced into sub-samples at 1-cm intervals in situ. At Xinfengjiang Reservoir, the sediment cores were sliced into sub-samples at 0.5-cm intervals for the top 10 cm and at 1-cm intervals for the remaining parts. Table 3.2 summarizes the details for sediment cores at the four reservoirs, including locations, core lengths and sampling conditions.



Figure 3.12 Pictures of a) portable Kajak gravity corer; b) one collected sediment core to be sliced into sub-samples in-situ immediately

Name of	Time of	Sampling	Sediment	Latitude	Longitude	Length	Water
Reservoir	Foundation	Time	Cores			(cm)	Depth
							(m)
Tingxi	Jun., 1956	Sep. 18,	TX-A	24°48.48'N	118°08.53'E	32	25
Reservoir		2006	TX-B	24°48.20'N	118°08.36'E	46	30
			TX-C	24°47.85'N	118°08.23'E	30	27
Xinfengjiang	Oct., 1959	Sep. 8,	WLH-A	23°45.75'N	114°34.93'E	32	60
Reservoir		2006	WLH-B	23°45.75'N	114°34.93'E	26	60
			WLH-C	23°48.09'N	114°35.63'E	21	65
			WLH-D	23°47.51'N	114°34.10'E	37	80
Luokeng	1960	Sep. 14,	LK-A	21°49.08'N	111°16.68'E	37	30
Reservoir		2006	LK-B	21°49.66'N	111°17.28'E	40	25
			LK-C	21°49.66'N	111°17.28'E	50	25
Songtao	1969	Nov. 10,	ST-A	19°21.69'N	109°32.36'E	56	35
Reservoir		2006	ST-B	19°21.33'N	109°33.80'E	55	36
			ST-C	19°22.51'N	109°33.45'E	50	37
			ST-D	19°23.40'N	109°31.75'E	45	30

 Table 3.2
 Details of sediment cores collected in the four reservoirs

Surface soil samples were collected in the reservoir catchment area to study the elemental concentrations of baseline soils in the drainage basin, and to provide background information of regional geochemical conditions. Moss samples were also collected simultaneously to assess the significance of current atmospheric deposition of trace metals in the study area. The total number of mosses and surface soils collected are listed in Table 3.3. All the samples were stored in zipped polyethylene bags at 4 °C before laboratory analysis.

	Tingxi	Xinfengjiang	Luokeng	Songtao
	Reservoir	Reservoir	Reservoir	Reservoir
Moss	N = 5	N = 9	N = 3	N = 4
Surface Soil	N = 4	N = 5		N = 11

 Table 3.3
 Number of moss and surface soil samples collected at the four remote reservoirs

3.3 Sample Preparation and Analyses

3.3.1 Sample preparation

Prior to the chemical analysis, sediment samples were kept at around -26 °C to prepare for the subsequent freeze-drying process. The sediments were then freeze-dried at -45 °C for about three days. Freeze-drying was preferred to air drying in order to prevent baking of clay-rich samples at higher temperatures (Bradl, 2005). Weights before and after drying were determined for each sample to calculate dry densities. After drying, sediments were ground into fine particles in an agate grinder by a grinding roll, avoiding contamination caused by abrasion from the material of the grinding equipment. Soil samples were air-dried and grinded in a similar way. The impurities (*e.g.* stones, coarse materials, and other debris) in soil samples were screened out and removed before grinding. Mosses were pretreated by washing thoroughly with tap water first and then with distilled water ultrasonically for three times to remove soil particles in a modified method of Lee *et al.* (2005). The samples were subsequently air-dried at 60 °C and grinded until fine particles were obtained. The grinded samples were then stored in a desiccator, prepared for subsequent chemical analysis.

3.3.2 Determination of total metal concentrations

For analytical purpose, all the sample materials should be digested and transferred to a solution for final instrumental determination of elemental concentrations. Several concentrated acids and oxidizing reagents are suitable for acid digestion, such as hydrofluoric acid (HF), perchloric acid (HClO₄), nitric acid (HNO₃), sulfuric acid (H_2SO_4) and hydrogen peroxide (H_2O_2). The selection of acids depends on the constituents of the material for decomposition and the interference effect during the equipment analysis. To perform a complete decomposition of lattic constituents in a sample, HF should be used to break down the silicate structures. Other digestion methods without HF are considered as the pseudo-acid digestion. These methods can also provide consistent results and nearly total concentrations for most of the trace metals across the samples. The pseudo-acid digestion can give an assessment of the maximum potentially soluble or mobile contents of metals which are usually not bound in silicates. These metal contents most likely reflect the maximum potential hazard that could occur in the long term, or in extreme environmental regimes.

In the laboratory, a strong acid mixture of HNO_3 (69%) and $HClO_4$ (70%) was used for the sample digestion (Ip *et al.*, 2004). For sediments and soils, a

representative aliquot (about 0.25 g) of the sample was weighted and digested by 10 ml HNO₃ (69%) and 2.5 ml HClO₄ (70%) in a pre-cleaned Pyrex test tube. The digestion process was conducted in a specially designed exhaust by using an aluminum heating block, which heated from 50 °C up to 190 °C progressively for around 16 hours until complete dryness of the sediments. The residues were dissolved in 10 ml of 5% HNO₃ by leaching at 70 °C for 1 hour after cooling down. Digestion of moss samples was conducted in a similar way with 8 ml HNO₃ and 2 ml HClO₄. The solutions were then analyzed for the total concentrations of trace metals (*e.g.* Cd, Cu, Pb, Ni, Zn) and lithogenic major elements (*e.g.* Al, Fe) by Inductively Coupled Plasma – Atomic Emission Spectrometry (ICP-AES, Perkin Elmer Optima 3300DV) according to the method of Ip *et al.* (2007). The general detection limits of ICP-AES for different metal elements are listed in Table 3.4.

Element	Al 308.215	Ca 317.933	Cd 228.802	Co 228.616	Cr 283.563
D.L.	28	10	3	7	7
Element	Cu 327.393	Fe 238.204	Li 670.784	Mg 280.271	Mn 260.568
D.L.	10	5	3	2	2
Element	Ni 231.604	Pb 220.353	V 292.402	Zn 213.857	
D.L.	15	42	9	6	

Table 3.4The general detection limits of ICP-AES for different metalelements (unit: ppb)

A set of quality assurance and quality control measures was strictly implemented throughout the whole process from sampling, sample preparation, acid digestion,

to final concentration determination. Precautions were taken to prevent possible cross contamination between different samples. All the Pyrex test tubes were pre-soaked in 10% HNO₃ acid batch overnight and then rinsed with plenty of distilled deionized water (DIW) before use. For sample digestion, reagent blanks, sample duplicates, and standard reference material (i.e. NIST 1646a for sediments, NIST 2709 for soils, NIST 1515 for mosses, and laboratory internal standards) were inserted among samples in each batch to examine the reliability of the analytical method, and accuracy and precision of obtained data. During the analytical process by ICP-AES, a standard solution (prepared from Comb 21 of Perkin Elmer Ltd.) was measured every ten samples to monitor the equipment performance and control the drifting of data within 10%. It was shown that the relative standard deviations of duplicate samples were basically within 10%. There was no sign of contamination by the results from blank samples. Recovery rates of major and trace elements from standard reference material were generally over 80% except aluminum and lead. The amount of aluminum bound to silicates can not be effectively digested by the current method, while the relatively poor recovery of lead might be due to the low lead concentrations in the standard materials approaching the detection limit. By examining laboratory internal standards, nevertheless, the recovery rates of both elements were found to be around 80%-100%.

3.3.3 Lead-210 dating of sediment cores

Lead-210 occurs naturally as one of the radioisotopes in the ²³⁸U decay series, which has a half-life of 22.26 years. ²¹⁰Pb dating technology is considered as a useful geochronological tool for the determination of time-stratigraphy in undisturbed sediments and was applied to date the sediment cores during the study. Sediment subsamples were isolated for a period of around two half-lives $(9 \sim 10 \text{ months})$ of ²¹⁰Po (Smith and Walton, 1980) so that ²¹⁰Po could accrue sufficiently from its grandparent ²¹⁰Pb, and a secular equilibrium between them was supposed to achieve. ²¹⁰Pb activity was therefore indirectly determined by measuring ²¹⁰Po activity using alpha radioactivity counting. All the selected samples for dating were assayed in a high-resolution alpha spectrometer equipped with silicon surface barrier detectors and a multi-channel analyzer with ²⁰⁹Po used as an internal yield tracer (Turner and Delorme, 1996). ²¹⁰Po and ²⁰⁹Po were extracted, purified, and plated onto silver discs from 0.5 M HCl solution at 70 – 80 °C (Smith and Walton, 1980). Chemical yields, which were estimated from the recovery of ²⁰⁹Po, were also evaluated. The supported ²¹⁰Pb activity was determined by measuring the activity of its parent nuclide ²²⁶Ra. Because of small variations in ²²⁶R activity through each core, the supported ²¹⁰Pb activity was taken to be equal to the average 226 R activity for samples from the top, middle, and bottom portions of the cores.

It is assumed that atmospheric deposition of ²¹⁰Pb is reasonably constant when

averaged over several years. As an exponential decline in ²¹⁰Pb activity with depth was characterized in the present study, the Constant Flux – Constant Sedimentation Rate (CF:CS) model was applied to calculate the sedimentation rate of sediments in the reservoirs (Appleby and Oldfield, 1992).

3.3.4 Lead isotopic composition analysis

Lead isotopic analysis was conducted for selected sediment samples, soil samples, and moss samples to identify the anthropogenic sources of the pollutants and quantify the contribution of anthropogenic inputs. Lead isotopic ratios ²⁰⁴Pb/²⁰⁷Pb, ²⁰⁶Pb/²⁰⁷Pb, and ²⁰⁸Pb/²⁰⁷Pb were determined by Inductively Coupled Plasma – Mass Spectrometry (ICP – MS, Perkin Elmer ELAN 6100 DRC^{plus}). The solutions (after sample digestion) were diluted with 5% high purity HNO_3 to around 25 µg Pb/l depending on the daily instrument performance. Meanwhile, standard solution of a reference material (NIST 981) was also prepared and measured every five samples and also at the very beginning and the end of the whole batch. The analyses were performed with 150 sweeps for each reading and ten measurements were taken for each sample. The relative standard deviation of duplicate readings was generally within 0.5%. The measured mean ratios ²⁰⁴Pb/²⁰⁷Pb, ²⁰⁶Pb/²⁰⁷Pb, and ²⁰⁸Pb/²⁰⁷Pb of QCs were 0.0646±0.0001, 1.0931±0.0016, 2.3702 ±0.0029 respectively and drifts of all QCs were well below 0.5%. The results were in good agreement with the certified values (0.0646, 1.0933, and 2.3704 respectively).

3.3.5 Air back-trajectory analysis

To investigate origins and transport pathways of air masses entering the study areas, air back-trajectories were calculated using nesting meteorological data by the Hybrid Single-Particle Lagrangian Integrated Trajectory, Version 4.7 (HYSPLIT4 model) developed by the National Oceanic and Atmospheric Administration (NOAA) Air Resources Laboratory (Wang et al., 2006). The ending position (*i.e.* latitude and longitude) of each reservoir was determined by taking the mean values of all the sampling sites. The trajectories were set to end at the height of 500 m above the ground level (AGL), which is normally half of the height of the mean daytime planet boundary layer (PBL) and represents the general transport condition in the PBL (Lee et al., 2007). To assess potential seasonal variations of prevailing air mass movements, trajectories were evaluated every six hours (i.e. 0:00, 6:00, 12:00, and 18:00 for local times) per day during the period of Jan. 2006 – Feb. 2007. Cluster means of the trajectories for each season were calculated to identify the dominant air masses. Details regarding the trajectory setup for the four reservoirs were listed in Table 3.5.

Reservoirs	Latitude	Longitude	Height (m)	Starting Time
Tingxi	24.80°N	118.14°E	500	0:00, 6:00, 12:00,
Reservoir				18:00
Xinfengjiang	23.79°N	114.57°E	500	0:00, 6:00, 12:00,
Reservoir				18:00
Luokeng	21.83°N	111.29°E	500	0:00, 6:00, 12:00,
Reservoir				18:00
Songtao	19.36°N	109.54°E	500	0:00, 6:00, 12:00,
Reservoir				18:00

Table 3.5Setup details of air back-trajectories calculated for the fourreservoirs

3.3.6 Statistical analyses

Several statistical analyses were conducted to better interpret the obtained data. Two-sample t-test statistics was adopted to assess the variation of trace metal concentrations or historical fluxes between independent samples at different sampling locations. The association between each two metal elements was assessed by calculating the Pearson correlation coefficient. In this study, SPSS[®] for Windows Release 13.0 was utilized to perform all the statistical analyses.

CHAPTER 4 – ATMOSPHERIC DEPOSITION OF ANTHROPOGENIC HEAVY METALS AT XINFENGJIANG RESERVOIR – IMPLICATION OF REGIONAL AIR POLLUTION

The present study was conducted at Xinfengjiang Reservoir in a remote area of South China with systematic sampling of reservoir sediment cores, mosses, and surface soils in the catchment area. The objectives are: 1) to investigate historical variations of heavy metal concentrations by assessing the vertical profiles of sediment cores; 2) to identify the input sources of trace metals by lead isotopic composition analysis and elemental association analysis; 3) to assess the impact of current air pollution on the ecosystem in a typical remote area of South China.

4.1 Metal concentrations in sediment profiles

The elemental profiles of Al, Fe, Cu, Ni, Pb and Zn in the four sediment cores from Xinfengjiang Reservoir are presented in Fig. 4.1. The distribution of major elements Al and Fe were generally stable above 18 cm in all the sediment cores, followed by a sharp drop of concentrations with increasing depth occurred below 15 cm (Fig. 4.1a and 4.1b). WLH-A showed the slowest decreasing rate while the slopes of WLH-B and WLH-C were generally similar and comparatively steeper than WLH-A. The metal concentrations declined fastest in WLH-D and reached the minimum values at around 20 cm. A peak was observed around 25-30 cm, below which the concentrations were kept constant back to the minimum values.

A continuous decreasing trend of heavy metal concentrations with increasing depth was observed within the depth of 0-20 cm (Fig. 4.1c-4.1f), in contrast to the generally constant concentrations of major elements. However, the distribution patterns of Cu, Ni, Pb and Zn in deep layers were found to be similar to those of Al and Fe, except for WLH-A. In WLH-A, the concentrations of Cu and Ni showed an increasing trend below 20 cm until they tended to decrease at 30 cm. Zn concentrations in the lower profiles of sediments did not vary much. Heavy metal concentrations in the bottom sediments of WLH-A were also comparatively higher than the values in sediments at the same levels of the other three cores. It is also noticed that the distribution patterns of these trace elements in top sediments were not exactly the same although they had a common variation trend (i.e. decreasing trend of heavy metal concentrations with increasing depth). Cu, Ni and Zn concentrations were found to be sharply elevated in the top 5 cm of the sediment cores while the variations of Pb concentrations showed a generally uniform descending trend along the whole length, with the only exception of two abnormally high concentrations at 23 cm and 26 cm observed in WLH-B.



Figure 4.1 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Cu (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Xinfengjiang Reservoir

4.2 Enrichment of heavy metals

As can be seen from the sediment profiles, the decreasing rate of metal concentrations in bottom layers of sediments (below 15 cm) of Core WLH-A was much slower than those of other cores, suggesting the sedimentation rate at the location of WLH-A might be much higher than other sampling locations. In addition, the distinctive distribution patterns of trace elements (e.g. Cu and Ni) in WLH-A from other sediment cores possibly indicate that post-depositional disturbance occurred in bottom sediments. Considering the abnormal condition of Core WLH-A, the other three sediment cores, including WLH-B, WLH-C and WLH-D, were selected for further analysis. In Fig. 4.2, the means and standard deviations of heavy metal concentrations at various depths along the sediment cores at Xinfengjiang Reservoir are presented. Statistical significance of differences in metal concentrations between different layers of sediments was also assessed. As the results showed, heavy metals were gradually enriched in upper layers. The concentrations of Cu, Ni, Pb and Zn in surface sediments (0-5 cm) reached 39.6 mg/kg, 14.9 mg/kg, 43.3 mg/kg and 138 mg/kg respectively. Metal concentrations at 0-5 cm (n = 29) were significantly higher than those below 30 cm (n = 8) (p < 0.01), particularly for Cu and Zn. Cu and Zn concentrations in the surface sediments were over fourfold higher than the values in bottom layers, while Pb concentrations were also elevated almost threefold. Compared with the three trace elements, the enrichment of Ni was relatively small, which was less than twofold. In the upper 20 cm of sediments, the

concentrations of Cu, Pb and Zn showed a significant decreasing trend while variations of Ni concentrations was not significant. There were no significant differences in heavy metal concentrations at the peak of 26-30 cm, compared to the overlying sediment layers (*i.e.* 21-25 cm).



Figure 4.2 Variations of heavy metal concentrations at different depths of the sediment cores at Xinfengjiang Reservoir: Cu (a), Ni (b), Pb (c), and Zn (d). The annotation 'a' indicates a significant difference of metal concentrations with the above sediment layer at p < 0.05 level, while 'b' implies no statistically significant difference between the two adjacent layers.

Although the heavy metals, especially Cu, Pb and Zn, were found to be significantly enriched in sediments, the raw data of heavy metal concentrations are not sufficient enough to infer the presence or quantify the intensity of contamination in sediments (Schropp *et al.*, 1990; Aloupi and Angelidis, 2001). Changes in sediment properties, such as granulometric (particle size) and mineralogical (rock type) variations, can affect the natural background concentrations of metals in sediments (Tam & Yao, 1998; Schiff & Weisberg, 1999). The enrichment of trace metal concentrations in sediments of Xinfengjiang Reservoir might be attributed to natural environmental changes or the contamination due to increasing inputs of anthropogenic pollutants from catchment areas or transported from industrial or urban areas.

The lithogenic major elements Fe and Al are abundant in the crust and are mostly derived from natural materials, thus reflecting the natural environmental conditions. A high correlation of the lithogenic elements and the trace metals is normally expected for uncontaminated sediments and vice versa. Considering the longest core WLH-D, the trace elements were found to be closely correlated with the major elements in sediments below 15 cm ($r_{Cu-Fe} = 0.983$; $r_{Ni-Fe} = 0.977$; $r_{Pb-Fe} = 0.960$; $r_{Zn-Fe} = 0.989$). Therefore, the large increment of trace metal concentrations at 15-20 cm is probably attributed to the natural environmental changes. On the other hand, the trace metal concentrations had poor relationships with the major elements above 15 cm ($r_{Cu-Fe} = 0.406$; $r_{Ni-Fe} = 0.240$; $r_{Pb-Fe} = 0.275$;

 $r_{Zn-Fe} = 0.416$), suggesting that the enrichment of trace metals in the top sediments (0-15 cm) are mainly derived from anthropogenic sources. To identify the significance of the anthropogenic inputs, the enrichment factors of Cu, Ni, Pb and Zn concentrations in top sediments were determined by comparing the trace metal concentrations at 0-2 cm (47.3 mg/kg, 17.4 mg/kg, 42.2 mg/kg, and 153.2 mg/kg respectively) and at 14-15 cm (21.4 mg/kg, 13.9 mg/kg, 30.9 mg/kg, and 97.3 mg/kg respectively). It can be noticed that the enrichment of anthropogenic Cu was most significant at the reservoir. The Cu concentrations in surface sediments were over twofold of the concentrations at 14-15 cm. The Zn concentrations were also elevated by nearly 60%. The Pb and Ni appeared relatively less enriched (*i.e.* 37% for Pb and 25% for Ni) in surface sediments by anthropogenic inputs.

As mentioned in Section 2.3.2, geochemical normalization of metal concentrations to a reference element (normalizer) can be used to distinguish natural contributions from the anthropogenically enriched metal concentrations. In order to further examine the extent of metal contaminations in the sediments, proper normalizers were identified to reduce the natural effects. In Table 4.1, the Pearson correlation coefficients between major (Al and Fe) and trace elements (*i.e.* Cu, Ni, Pb and Zn) in the sediment cores are summarized. The results showed that three trace elements, including Cu, Ni and Zn, had much stronger correlations with Fe. The correlation coefficients between Pb and the two major

elements were similar ($r_{Pb-Al} = 0.665$; $r_{Pb-Fe} = 0.663$). Normally, a possible best normalizer is the highest geochemically sensible tracer, referring the strongest correlator (Yang and Rose, 2005). Fe was thus considered as a better normalizer for Cu, Ni and Zn, while Al and Fe were regarded both appropriate for Pb normalization.

Table 4.1 Pearson correlation coefficients between elemental concentrations in sediment cores at Xinfengjiang Reservoir $(n = 114)^{a}$

Element	Al	Fe	Cu	Ni	Pb	Zn
Al	1					
Fe	0.947	1				
Cu	0.613	0.732	1			
Ni	0.762	0.834	0.861	1		
Pb	0.665	0.663	0.721	0.641	1	
Zn	0.784	0.847	0.933	0.857	0.782	1

a. All the correlations are significant at the 0.01 level (2-tailed).

A set of uncontaminated samples needs to be identified to obtain reliable baseline conditions for defining metal-normalizer relationships. As the concentrations of Cu, Ni, Pb and Zn in sediments below 15 cm of Core WLH-D were found to be highly correlated with Al and Fe ($r_{Cu-Fe} = 0.983$; $r_{Ni-Fe} = 0.977$; $r_{Pb-Fe} = 0.960$; $r_{Zn-Fe} = 0.989$), they were regarded as uncontaminated sediments. The baseline metal-normalizer reference element relationships (simple linear regression lines) for the study area were then established based on metal concentrations in sediments over 15-38 cm of Core WLH-D. The 95% prediction intervals were subsequently derived from the linear regression models for the purpose of comparing individual samples to the baseline relationships.

The scatter plots of heavy metals and the respective normalizers are presented in Fig. 4.3. The linear regression lines and 95% prediction intervals are superimposed on the data. Sediment samples were then assessed depending on the plotted positions relative to the regression lines and prediction intervals. It was suggested that any samples above the upper 95% prediction limit (with high excursions) are probably contaminated by heavy metals from anthropogenic sources, while data points within the prediction limits are usually originated from natural sources (Daskalakis and O'Connor, 1995; Schiff and Weisberg, 1999; Aloupi and Angelidis, 1999). It can be noticed that large amount of samples in upper sediment cores exceeded the prediction limits; while samples in lower layers at low concentrations were generally within the natural ranges. The deviations were larger when the concentrations of heavy metals were higher. These findings probably suggest that trace elements in the sediments of Xinfengjiang Reservoir are derived from natural sources in the early stage. However, the sediments are likely to be anthropogenically contaminated by heavy metals in the upper layers (especially in top 10 cm) where significant elevation of heavy metals occurred. Additional evidence could be obtained from the sediment profiles. The similar distribution patterns and high correlations between heavy metals (*i.e.* Cu, Ni, Pb and Zn) and major elements (*i.e.* Al and Fe) in lower sediment layers also suggest the natural origin of the trace elements. In

upper layers, heavy metals were greatly enriched while Al and Fe concentrations generally remained constant. The poor correlation between major and trace elements further reflects the possible contamination of sediments.



◯ WLH-B ◯ WLH-C ◯ WLH-D

Figure 4.3 Scatter plots of Cu-Fe (a), Ni-Fe (b), Pb-Fe (c) and Zn-Fe (d) relationships in sediments of Xinfengjiang Reservoir. Linear regression baselines (obtained based on the metal concentrations in bottom sediments (*i.e.* 15-38 cm) of Core WLH-D) with 95% confidence intervals are superimposed on the data for comparison.

Till now, few studies have been conducted on historical variations of heavy metal concentrations in lacustrine sediments in remote area of South China. The existing studies mainly focused on the heavy metal contamination in sediments of the Pearl River Estuary and the surrounding coastal area (Li *et al.*, 2000; Liu *et al.*, 2003; Ip *et al.*, 2004). The heavy metal contents in sediment cores were found to have generally increased over the last three decades due to the recent rapid industrial development and urbanization in the estuarine region. Although the enrichment of heavy metals in the coastal and estuarine sediments was dominantly ascribed to wastes discharged from point sources, the increasing atmospheric input of anthropogenic pollutants was also considered as an important contributing source, particularly for Pb.

As Figure 4.3 also indicated, much more data points were found beyond the natural ranges in Cu-Fe, Pb-Fe and Zn-Fe plots. This reflects that heavy metal pollution at the reservoir is particularly serious for Cu, Pb and Zn, although the total enrichment of anthropogenic Pb in upper layer sediments was not as high as that of Cu and Zn. Considering the Ni-Fe plot, only a few samples were observed above the prediction interval, indicating relatively less significant enrichment of anthropogenic Ni in surface sediments. As the contamination of Ni by anthropogenic sources appears not serious in sediments of Xinfengjiang Reservoir, the following discussion will focus on Cu, Pb and Zn.

4.3 Contributions of potential sources to the enrichment

Without obvious direct discharges of pollutants from point sources, heavy metals in the reservoir sediments are supposed to be mainly derived from direct atmospheric deposition over the reservoir area and contributed by inputs from the catchment area via inflow streams, surface runoff, soil leaching or erosions. Due to the high drainage ratio (catchment area vs. open surface water area) of Xinfengjiang Reservoir, which is 15.5:1, the contributions from the catchment might be appreciable, or even more significant than the direct atmospheric inputs (Yang *et al.*, 2002; Shotbolt *et al.*, 2006).

As illustrated in Section 4.2, the concentrations of heavy metals in lower sediment layers (generally below 15 cm) had high correlations with the major elements. The elevated heavy metal concentrations are thus probably attributed to the accelerated erosion rate of natural soils from the watershed, as the increase of natural weathering normally results in accumulation of rock-forming elements (Ip *et al.*, 2004). This was well reflected by the simultaneously elevated Al and Fe concentrations in the lower parts of the sediment cores (see Fig. 4.1).

The results also indicate, however, that the upper layer sediments of the reservoir are anthropogenically enriched in heavy metals. Considering the catchment soils, no significant differences in concentrations of Cu and Pb existed between the surface soils and reservoir bottom sediments. The latter one probably indicates the background levels of the study area. The catchment soils are thus not likely to be contaminated by anthropogenic Cu or Pb. Although Zn concentrations were much higher in surface soils, strong correlations were observed between Zn and the major elements ($r_{Zn-Al} = 0.954$; $r_{Zn-Fe} = 0.831$), suggesting its geological origin. Moreover, no sign of enrichment of major elements was noticed in upper layers of the reservoir sediments. The upper layer sediments are thus considered to be mainly contaminated by anthropogenic heavy metals from atmospheric deposition, which concurrently occurred in both the terrestrial catchment and at the reservoir.

The surface sediments were assumed to be deposited in recent years without post-depositional disturbance of the sediments at the reservoir. As mentioned in Chapter 1, enormous economic development has taken place in South China in the past three decades. The great amount of anthropogenic pollutants released into the atmosphere in the intensively industrialized and urbanized area might be transported over vast distance to the remote area. During May to August, the eastern China and South China coastal region is normally prevailed by the summer monsoon, which advances northward in stepwise way with simultaneous northward movement of precipitation belts (Ding, 1994). Consequently, the air masses transported from the South China coastal region would just sweep across the central and northern Guangdong Province, including Xinfengjiang Reservoir. Previously, Lee *et al.* (2005) assessed the mosses and aerosols in Nanling

Mountains lying along the north border of Guangdong Province, and noticed that the area was susceptible to the strong influences of air pollutants originated from the PRD region. This further suggests that the enrichment of heavy metals in surface sediments of the reservoir is probably ascribed to the increasing atmospheric inputs of air pollutants from regional anthropogenic sources in recent years.

In this study, mosses were collected and analyzed to assess the significance of atmospheric heavy metal deposition recently (around 2-3 years). Most of the samples belong to the species Pogonatum contortum, which is commonly found in China (e.g. Guangdong, Guangxi, Hainan, Sichuan, Henan and the northeast region), Russia (the far east region), Japan and North America (west coast) (Yuan et al., 2006). Although there have been no previous records available for using *Pogonatum contortum* to biomonitor the airborne heavy metal concentrations, the results from this study indicate that this type of mosses is effective in accumulating airborne heavy metals. The mean concentrations and standard deviations of trace elements Cu, Pb and Zn in the mosses were 10.5 \pm 4.36 mg/kg, 47.1 ± 27.9 mg/kg, and 133 ± 44.7 mg/kg respectively. In Europe, several studies have been conducted to monitor the airborne heavy metals by mosses in relatively remote areas (Reimann et al., 2001; Ruhling et al., 2004). They reported that the Cu, Pb and Zn concentrations accumulated in moss samples were generally around 4.5 mg/kg, 4.2 mg/kg, and 32.3 mg/kg. In

comparison, the heavy metal concentrations in mosses of this study were found to be much higher than the values reported in Europe (*i.e.* twofold for Cu, eleven-fold for Pb, and fourfold for Zn), especially for Pb. This probably suggests that the regional atmospheric deposition of air pollutants in South China is much more significant and might contribute substantial amounts of anthropogenic heavy metals in the surface environment of the study area.

As suggested by Schilling et al. (2002), different moss species in distinct meteorological and topographical conditions may have varied cation exchange capacities, which might subsequently influence the metal accumulation in mosses. Therefore, a multi-media approach was adopted in the present study to monitor the atmospheric heavy metals by considering the results of both mosses and reservoir sediment cores. The concentrations of Cu in mosses were noticed to be closely correlated with Zn ($r_{Cu-Zn} = 0.692$), suggesting these two elements are likely to be originated from common pollution sources. However, the relationships between Pb and these two trace elements were not strong ($r_{Pb-Cu} =$ -0.053; $r_{Pb-Zn} = 0.266$), which might be attributed to their different origins or associated particle sizes of the emission sources. Lead is mostly associated with fine atmospheric particles while Cu and Zn are distributed in both fine and coarse particulates (Sweet et al., 1998). Similar elemental association relationships were observed in surface sediments. Cu and Zn were also found to be strongly interrelated ($r_{Cu-Zn} = 0.953$) in the top sediments (0-5 cm) while no close

relationship was observed between Pb and these trace elements ($r_{Pb-Cu} = 0.107$; $r_{Pb-Zn} = 0.142$). The consistency of results from both types of environmental materials further reflects that the significant atmospheric fluxes (including the indirect inputs from the catchment) are probably the dominant input sources of anthropogenic heavy metals in upper layer sediments of the reservoir.

4.4 Identification of regional sources

In order to identify possible origins of the air pollutants, lead isotopic compositions were determined to distinguish between different sources, as each lead containing ore or other natural materials has specific lead isotopic signature and the values are not subject to large changes by chemical or biological fractionation (Renberg *et al.*, 2002; Duzgoren-Aydin *et al.*, 2004). Duzgoren-Aydin *et al.* (2004) stated that anthropogenic lead in the atmosphere was mainly derived from combustion of leaded petrol, industrial discharges, and coal burning. In Table 4.2, data of available backgrounds (*e.g.* natural bedrocks), potential anthropogenic sources (*e.g.* ore deposits, automobile exhausts), as well as the values of aerosols and mosses collected in the PRD region and other surrounding areas retrieved from previous literatures are listed for comparison. The relationships between different types of samples are plotted and depicted in Figure 4.4.

Nature of Sample	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb
Regional Natural Background		
Volcanic rock (HK) ^a	1.221	2.529
Volcanic rock (PRD) ^b	1.199	2.497
Uncontaminated soils (PRD) ^b	1.195	2.482
Regional Pollution Sources		
Automobile exhausts (PRD) ^b	1.160	2.423
Fankou Pb-Zn ore deposit ^b	1.172	2.473
Aerosols and Mosses in South China		
PRD aerosols ^c	1.161-1.177	2.366-2.501
Foshan aerosols (urban) ^b	1.155-1.170	2.457-2.474
Zhongshan aerosols (suburban) ^d	1.169-1.173	2.444-2.461
Nanling mosses (rural) ^d	1.171-1.190	2.452-2.463
Nanling aerosols (rural) ^d	1.171-1.179	2.447-2.469

Table 4.2Lead isotopic compositions of potential natural and anthropogenicsources.

a. Duzgoren-Aydin et al., 2004;

b. Zhu et al., 2001;

c. Wong et al., 2003;

d. Lee et al., 2005.



Figure 4.4 Comparisons of Pb isotopic compositions between sediments, mosses, and surface soils at Xinfengjiang Reservoir, with reference to available data of backgrounds and potential anthropogenic sources in the PRD region

The results showed that the lead isotopic compositions of mosses and soils varied over large ranges (*i.e.* 1.173-1.212 and 1.178-1.225, respectively), possibly indicating the influences of various types of lead sources. However, each of the data set presented a good linear relationship between the natural sources and anthropogenic sources (see Fig. 4.4), indicating a binary mixing effect of geogenic Pb and anthropogenic Pb. The anthropogenic Pb emitted from industrial processes using lead-containing ores and coal combustion (with similar lead isotopic compositions to some of the lead ores) in the PRD region are probably the dominant contributing sources. Although leaded gasoline has been gradually phased out in China since 2000, the lead isotopic compositions of moss

and soil samples also appeared to be influenced by the automobile lead emissions. This might suggest that Pb emitted in the past by the usage of leaded petrol still remains due to its long persistence in the environmental medium (e.g. catchment soils), which becomes another important source of anthropogenic Pb (Bollhofer and Rosman, 2001; Chen *et al.*, 2005). Compared with mosses and soils, the range of values in sediments was much narrower (*i.e.* 1.196-1.207) with the ratios slightly decreased towards the surface (*i.e.* no large variations). This is possibly attributed to the mixing effects of both natural inputs from the catchment (*e.g.* erosion of catchment soils and bedrocks) and atmospheric deposition of anthropogenic pollutants. Moreover, substantial anthropogenic Pb might be retained in the terrestrial catchment caused by its large retention capacity (*i.e.* large catchment area).

As the figure also illustrates, the lead isotopic ratios of mosses were consistent with the reported values of aerosols and mosses in Nanling Mountain, which is situated in a relatively remote area of South China as well. Lee *et al.* (2005) suggested that the lead isotopic compositions of mosses could reflect the isotopic signature of lead of the surrounding ambient air. Therefore, the atmosphere over the study area is regarded to be significantly influenced by regional anthropogenic sources. The increment of Pb concentrations in top layers of sediment cores at Xinfengjiang Reservoir might thus be attributed to the increasing atmospheric deposition of anthropogenic pollutants emitted and transported from the PRD region in recent years. Further investigations are required, however, to provide additional evidence on anthropogenic sources of the heavy metals in sediments of Xinfengjiang Reservoir. The results would be of great importance for establishing proper management policies to alleviate potential impairment of environment and ecosystems at Xinfengjiang Reservoir.

CHAPTER 5 – THE SUITABILITY OF USING RESERVOIR SEDIMENTS AS GEOCHEMICAL ARCHIVES OF ATMOSPHERIC HEAVY METAL INPUTS – FACTORS INFLUENCING TRANSPORT AND DEPOSITION OF AIRBOREN HEAVY METALS

In the present study, the atmospheric heavy metal pollution was assessed at four typical reservoirs (*i.e.* Tingxi Reservoir, Xinfengjiang Reservoir, Luokeng Reservoir and Songtao Reservoir) located in the remote area of the South China coastal region. It is considered that these reservoirs might present different signatures as to the impacts of atmospheric heavy metal pollution due to their distinctive locations, topographical and meteorological conditions, as well as the specific physicochemical conditions of the reservoirs and their catchments. Therefore, the major objectives of this chapter are 1) to assess the long-term variations of heavy metal concentrations in sediments of each reservoir; 2) to evaluate the significance of metal enrichments in recent years; 3) to identify the dominant contributing sources of heavy metal inputs in the sediments; 4) to examine the possible influencing factors affecting atmospheric deposition of anthropogenic heavy metals in sediments of reservoirs in remote areas.

5.1 Concentration Profiles of Metals in Sediment Cores at Different Reservoirs in the Remote Area

The concentration profiles of metals in the sediment cores were assessed at the four reservoirs respectively. Assuming no post-depositional disturbances, the sediment profiles probably mirror the long-term variations of heavy metal inputs in the reservoir.

5.1.1 Tingxi Reservoir

At Tingxi Reservoir, three sediment cores were collected at relatively deep locations across the reservoir. The concentration profiles of some major metals (*i.e.* Al and Fe) and trace elements (*i.e.* Co, Ni, Pb and Zn) in the sediments are presented in Fig. 5.1. As the low Cu concentrations in sediments from Tingxi Reservoir were approaching the detection limit, the data were not used for further discussion.



 \rightarrow TX-A \rightarrow TX-B \rightarrow TX-C

Figure 5.1 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Co (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Tingxi Reservoir

It can be noticed that the distribution patterns of major element Al in the three sediment cores varied much from each other. In Core TX-A, the Al concentration decreased gradually in the top 10 cm and then turned to increase until it reached about 97×10^3 mg/kg at 12 cm, below which the concentration generally kept constant. Similar variation pattern of Al concentration was observed in the longest Core TX-B, however, the sedimentation rate at this location was probably slower. The concentration firstly decreased continuously until it reached about 70×10^3 mg/kg at 11 cm, which was kept steady for the next 6 cm. It then increased sharply to 100×10^3 mg/kg at 17-20 cm. Below 20 cm, there was no large variation of Al concentration with the values kept around 95×10^3 mg/kg. Considering Core TX-C, the concentration was generally constant in the top 20 cm. It then decreased slightly to 60×10^3 mg/kg at 26 cm, below which the concentration was gradually raised back to 75×10^3 mg/kg. Compared to Al, a generally uniform distribution pattern of Fe concentrations was observed in all the three cores. The concentration was kept around 25×10^3 mg/kg, with only a slight elevation in Core TX-B in the top 5 cm.

The concentration profiles of four selected trace elements Co, Ni, Pb and Zn are shown in Fig. 5.1(c)-(f). As the figures show, the concentrations of Co and Ni were generally kept constant along the sediment cores. The distribution patterns were quite similar to the major element Fe. The variation trends of Zn concentrations in the three cores were also found to be similar to those of Al concentrations. However, the concentration profiles of Pb appeared quite different from other heavy metals. Large variations of Pb concentrations were observed along the whole length of the sediment cores in TX-A and TX-B. Moreover, a reserved trend with relatively higher concentrations in the bottom sediments was observed. The only exception was Core TX-C, in which the variation of Pb concentrations was similar to those of both major (*i.e.* Al and Fe) and other trace elements (*i.e.* Co, Ni and Zn).

The results from Pb²¹⁰ dating analysis indicate that Core TX-B is probably subject to some post-depositional disturbances. The ²¹⁰Pb activity was found to be abnormally lower in deeper sediment layers (see Fig. 5.2). The sediment profiles of Core TX-B are thus not suitable as a geochemical archive and cannot provide information on the long-term variations of heavy metal concentrations at the reservoir. The similar distribution patterns of metals in Core TX-A, especially the irregular variation trend of Pb, probably suggests this sediment core is also disturbed to a certain extent. Only Core TX-C is supposed to be well preserved. Both the major and trace elements were uniformly distributed along the sediment core. There were no significant differences in metal concentrations between the top (0-5 cm, n = 5) and bottom (below 25 cm, n = 5) sediments (p > 0.1). The mean concentrations of Co, Ni, Pb and Zn in the surface sediments were 7.62±0.38 mg/kg, 7.20±0.29 mg/kg, 46.6±2.27 mg/kg, and 116±4.48 mg/kg. The Pearson correlation coefficients between the metal elements were determined to assess the origins of heavy metals in the sediments at the reservoir (see Table 5.1). The results showed that all the examined heavy metals were strongly correlated with major elements. A close relationship between major and trace elements normally reflects that the sediments are not significantly contaminated by anthropogenic inputs. The heavy metals in the sediments at Tingxi Reservoir are thus expected to be mainly originated from natural sources. There is no obvious sign of heavy metal enrichment in reservoir sediments caused by atmospheric deposition of anthropogenic air pollutants (including the indirect inputs from the catchment). Therefore, the possible adverse impacts of either local or regional air pollution of heavy metals on the ecosystems of Tingxi Reservoir cannot be clearly observed by examining the sediment profiles.



Figure 5.2 Plot of the unsupported ²¹⁰Pb radioactivity against the depth of the sediment core LK-B from Luokeng Reservoir
Element	Al	Fe	Со	Ni	Pb	Zn
Al	1					
Fe	0.923	1				
Со	0.680	0.734	1			
Ni	0.911	0.933	0.833	1		
Pb	0.727	0.911	0.760	0.857	1	
Zn	0.862	0.944	0.747	0.951	0.902	1

Table 5.1 Pearson correlation coefficients between elemental concentrations in sediments of Tingxi Reservoir $(n = 30)^{a}$

a. All the correlations are significant at the 0.01 level (2-tailed).

5.1.2 Luokeng Reservoir

At Luokeng Reservoir, the concentration profiles of Al, Fe, Cu, Pb and Zn are plotted for two selected sediment cores LK-B and LK-C in Fig. 5.3(a) - 5.3(f) respectively.



Figure 5.3 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Cu (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Luokeng Reservoir

It can be noticed that the variation trends of both major (*i.e.* Al and Fe) and trace elements (*i.e.* Cu, Ni, Pb and Zn) in the two sediment cores closely resembled each other, possibly indicating their similar sedimentation conditions. The Al concentrations gradually decreased with increasing depth in top sediments until they reached the minimum values at around 17 cm. The concentrations were then elevated sharply back to 105×10^3 mg/kg at 20 cm and kept generally constant in the bottom sediments. Similar distribution patterns of Fe concentrations were observed except that Fe concentrations in the top 10 cm were also kept stable, which was around 20×10^3 mg/kg.

Considering the trace elements, the variation trends of Cu, Ni, Pb and Zn were found to be similar to those of major elements Al and Fe. The concentrations of the four heavy metals reached their minimum values at around 15-17 cm and then gradually increased back to the values prior to the decrement. Likewise, the concentrations in sediments below 20 cm were generally kept constant. However, the distribution patterns of trace elements in the top 15 cm of sediments varied from each other. In the sediment profiles of Cu, Ni and Zn, a peak, which was most obvious for Cu, was observed at the depth of 5-10 cm. As for Pb, however, the concentration decreased continuously in the top 15 cm until the minimum value was obtained.

Dating of the sediment core LK-B was conducted by using ²¹⁰Pb dating technology to assess the time-stratigraphy. The results were shown in Table C-2 of Appendix C. The unsupported ²¹⁰Pb was found to vary exponentially with the depth of the sediments (see Fig. 5.4). Therefore, the Constant Flux – Constant Sedimentation Rate (CF:CS) model was applied to date the sediment core (Appleby and Oldfield, 1992). The calculation was conducted based on the following equation: $t = \frac{1}{\lambda} \ln \left(\frac{2^{10} P b^0}{2^{10} P b} \right) = 32.15 \times \ln \left(\frac{2^{10} P b^0}{2^{10} P b} \right)$, where t represents the age, λ is the radioactive decay constant of ²¹⁰Pb ($\lambda = 3.11 \times 10^{-2} \text{ y}^{-1}$), ²¹⁰Pb⁰ is the initial activity of unsupported ²¹⁰Pb, and ²¹⁰Pb is the measured (present-day) activity of unsupported ²¹⁰Pb. According to the results from the dating analysis, the annual sedimentation rate at Luokeng Reservoir = 56.313/32.15 = 1.75cm/year. The ages were then determined based on the sedimentation rate and the corresponding depths. The obtained results were superposed on the sediment profiles in Fig. 5.2. The longest core LK-B was found to have accumulated sediments for a period of 25 years. As the figure presents, the distribution of metals in the sediments at Luokeng Reservoir was generally stable prior to 1995. A common drop of concentrations of both major and trace elements occurred around 1996-1997. During the recent ten years, relatively higher concentrations of Cu, Ni and Zn were noticed in 2002 while Pb concentrations were continuously elevated as time went on.



Figure 5.4 Plot of the unsupported ²¹⁰Pb radioactivity against the depth of the sediment core LK-B from Luokeng Reservoir

The average concentrations of heavy metals Cu, Ni, Pb and Zn in the surface sediments (0-5 cm, n = 10) were 6.45 ± 0.67 mg/kg, 3.60 ± 0.22 mg/kg, 38.7 ± 2.08 mg/kg, and 71.8 ± 4.88 mg/kg respectively, while the concentrations in bottom sediments (below 45 cm, n = 15) were 5.77 ± 0.79 mg/kg, 2.81 ± 0.61 mg/kg, 31.1 ± 4.86 mg/kg, and 62.2 ± 4.72 mg/kg. There have been only slight elevations of heavy metal concentrations (*i.e.* 12% for Cu, 29% for Ni, 24% for Pb, and 17% for Zn) in the past 25 years. The elemental associations were analyzed by calculating the Pearson correlation coefficients. The results are shown in Table 5.2 as follows. The significant correlations of heavy metals in the sediment profiles are attributed to natural changes. The sediments at Luokeng Reservoir are thus not likely to be significantly contaminated by anthropogenic heavy metals.

Element	Al	Fe	Cu	Ni	Pb	Zn	
Al	1						
Fe	0.900	1					
Cu	0.788	0.858	1				
Ni	0.759	0.810	0.863	1			
Pb	0.891	0.770	0.704	0.680	1		
Zn	0.780	0.864	0.828	0.740	0.714	1	

Table 5.2Pearson correlation coefficients between elemental concentrationsin sediment cores at Luokeng Reservoir $(n = 105)^a$

a. All the correlations are significant at the 0.01 level (2-tailed).

5.1.3 Songtao Reservoir

At Songtao Reservoir, three sediment cores, including ST-A, ST-B, and ST-D, were used to assess the long-term variations of metal concentrations in the reservoir sediments. The concentration profiles of Al, Fe, Cu, Ni, Pb and Zn in the sediment cores are presented in Fig. 5.5 (a)-(f) respectively.



Figure 5.5 Concentration profiles of major elements Al (a) and Fe (b), and trace elements Cu (c), Ni (d), Pb (e), and Zn (f) in sediment cores from Songtao Reservoir

As the figures show, the distribution patterns of the metal elements in upper layer sediments of the three cores were generally similar, while they were rather scattered in the bottom sediments below 30 cm. In Core ST-A, a reversed variation trend was observed for the major elements Al and Fe, the concentrations of which were found to be slightly higher in deep layers than in the surface sediments. The variations of lithogenic metals normally indicate changes of the natural environment. It was suggested that the increase of natural weathering of soil materials in the watershed normally result in accumulation of lithogenic elements in aqueous sediments (Ip et al., 2004). Therefore, the reduced concentrations of major metals towards the surface sediments in Core ST-A of Songtao Reservoir might be caused by decelerated soil erosions from the catchment area over time. Considering Core ST-B, the metal concentrations of Al and Fe were generally kept stable around 80×10^3 mg/kg and 35×10^3 mg/kg respectively, except for several peaks occurring at the depth of 30-40 cm. These peaks might be attributed to accelerated weathering of natural soils from the catchment during that period of time. In Core ST-D, the concentrations of Al and Fe were relatively lower in sediments below 30 cm compared to the values in upper layers. The concentrations decreased gradually between 20-30 cm until they reached the minimum values at 30 cm. Below 30 cm, the concentrations fluctuated irregularly with two peaks formed at 35 cm and 40 cm respectively. With regard to the trace elements, the variation trends of Cu, Ni, Pb and Zn appeared similar to those of the major elements.

The ²¹⁰Pb dating analysis was conducted to assess the time-stratigraphy of sediment cores at Songtao Reservoir. Similar to Luokeng Reservoir, the CF:CS model was applied to date the sediment cores at Songtao Reservoir, as the unsupported ²¹⁰Pb also varied exponentially with depth of the sediments (see Fig. 5.6). Core ST-A was found to have been well preserved, providing a historical record of metal concentrations of nearly 50 years. The results were shown in Table C-3 of Appendix C. The annual sedimentation rate was 1.22 cm/year (i.e. 39.345/32.15), which was determined in a similar approach as Luokeng Reservoir. The ages were then calculated and plotted in correspondence with the depths in the sediment profiles (see Fig. 5.4 Y2-axis). The distribution patterns of metal elements at the Songtao Reservoir were found to be generally uniform since the 1980s. However, the sedimentation conditions across the reservoir varied much in early years. Regarding the peaks observed in both Core ST-B and ST-D around 30-40 cm, they are probably caused by accelerated catchment soil erosions due to intensive deforestation activities in the mid to late 1970s.



Figure 5.6 Plot of the unsupported ²¹⁰Pb radioactivity against the depth of the sediment core ST-A from Songtao Reservoir

The mean concentrations of trace metals at different depths along the sediment cores were then determined for ST-A and ST-B, which had accumulated sediments for a longer history. The elemental concentrations of Cu, Ni, Pb and Zn in surface sediments (0-5 cm, n = 10) were 28.4±3.91 mg/kg, 19.3±2.85 mg/kg, 24.4 \pm 3.42 mg/kg, and 98.9 \pm 9.56 mg/kg respectively. On the other hand, the concentrations of the four trace elements in bottom sediments (below 45 cm, n = 17) were 33.0±6.24 mg/kg, 22.5±3.12 mg/kg, 28.9±1.95 mg/kg, and 103±6.93 mg/kg. The results indicate that there has been no large enrichment of heavy metal concentrations (i.e. 16% for Cu, 17% for Ni, 18% for Pb, and 4% for Zn) in sediments of Songtao Reservoir in the past fifty years, with the concentrations gradually and slightly decreased towards the surface. The Pearson correlation coefficients between the metal elements were determined to assess the elemental associations (see Table 5.3). It can be seen that all the metal elements were strongly correlated with each other, reflecting the variations of heavy metals over time are probably attributed to the natural changes. The sediments at Songtao Reservoir are thus not likely to be contaminated by heavy metals from anthropogenic sources.

Element	Al	Fe	Cu	Ni	Pb	Zn
Al	1					
Fe	0.785	1				
Cu	0.780	0.952	1			
Ni	0.735	0.901	0.965	1		
Pb	0.557	0.700	0.769	0.813	1	
Zn	0.783	0.757	0.795	0.799	0.848	1

Table 5.3 Pearson correlation coefficients between elemental concentrations in sediment cores at Songtao Reservoir $(n = 107)^{a}$

a. All the correlations are significant at the 0.01 level (2-tailed).

5.1.4 Xinfengjiang Reservoir

As discussed in Chapter 4, the concentrations of heavy metals Cu, Pb and Zn were found to be elevated significantly in upper layers of sediments at Xinfengjiang Reservoir, reflecting the increasing inputs of regional air pollutants in recent years. The lead isotopic compositions of mosses imply that the atmosphere over the study area has been strongly influenced by the regional anthropogenic sources. Industrial use of lead-containing ores and past released Pb from leaded gasoline in the PRD region are regarded as the major contributors of substantial amount of anthropogenic lead at the reservoir. The increment of Pb concentrations in top sediments at Xinfengjiang Reservoir might thus be attributed to the increasing atmospheric deposition of anthropogenic pollutants from intensive industrial activities and heavy traffic in the PRD region in the past several years.

All the four reservoirs are located in remote area of the South China coastal

region. While the sediment cores from Xinfengjiang Reservoir showed enriched heavy metal concentrations due to increasing atmospheric inputs over time and strong influences of regional anthropogenic sources in recent years, there was no obvious contamination sign of sediments at the other three reservoirs. However, this cannot directly indicate the atmosphere at Tingxi Reservoir, Luokeng Reservoir, and Songtao Reservoir is not impaired by the increasing local or regional emissions of air pollutants in the past several decades. In order to further assess the air quality in the four remote areas, moss samples will be analyzed by determination of metal concentrations and lead isotopic compositions in the following section.

5.2 Biomonitoring of the Air Quality by Mosses

5.2.1 Metal concentrations in mosses

Five moss samples were collected at the catchment area of Tingxi Reservoir. Most of them belong to the species *Hypnum plumaeform*, which is commonly found in dense mats in many Asian countries, such as China, Japan, Korea, Nepal and Philippines. Past investigations have suggested that this type of mosses is effective in accumulating airborne heavy metals (Lee *et al.*, 2005). In the present study, the average concentrations of trace elements Co, Cu, Ni, Pb and Zn in the mosses were 1.32 ± 0.49 mg/kg, 45.1 ± 14.1 mg/kg, 3.96 ± 0.87 mg/kg, 59.0 ± 13.4 mg/kg, and 169 ± 27.6 mg/kg, respectively. The Ni and Pb concentrations were found to be much lower than the values reported in Nanling Mountains, while Cu and Zn concentrations were comparable (Lee *et al.*, 2005). This might be due to the fact that Nanling Moutain is at the key pathway for the long-range transport of air pollutants from northern China to the RPD region, particularly during the dry winter monsoon period. As the sampling work of the two studies was conducted during different time periods, the comparison of metal concentrations in mosses from the two study areas was only for general indication.

At Xinfengjiang Reservoir, a total of nine moss samples were collected and analyzed to assess the significance of atmospheric heavy metal deposition. As illustrated in Section 4.3, most of the samples belong to the species *Pogonatum contortum*, which is considered effective in accumulating airborne heavy metals. The mean concentrations and standard deviations of trace elements Cu, Ni, Pb and Zn in the mosses were 10.5±4.36 mg/kg, 5.00±1.82 mg/kg, 47.1±27.9 mg/kg, and 133±44.7 mg/kg respectively. When compared to the values reported in some remote areas of the Europe, the heavy metal concentrations in mosses at Xinfengjinag Reservoir appeared much higher (Reimann *et al.*, 2001; Ruhling *et al.*, 2004).

Three moss samples of three different species (*i.e. Thuidium glaucinum*, *Claopodium assurgens, and Claopodium aciculum*) were sampled at Luokeng Reservoir. The average concentrations of Cu, Ni, Pb and Zn in the mosses were 9.23±1.81 mg/kg, 1.63±0.25 mg/kg, 36.4±19.5 mg/kg, and 86.4±19.4 mg/kg, respectively. There have been no previous records of using these species as biomonitors of heavy metals. The results showed that considerable Pb and Zn concentrations were accumulated in the moss samples.

At Songtao Reservoir, four moss samples were collected at the height of 650 m of Jianfengling Mountain. All the samples belong to the species *Pogonatum contortum*, which is the same as the samples collected at Xinfengjiang Reservoir. The average concentrations of the trace metals Cu, Ni, Pb and Zn in the mosses were 5.75 ± 0.21 mg/kg, 1.65 ± 0.17 mg/kg, 34.8 ± 3.46 mg/kg, and 59.9 ± 1.42 mg/kg respectively.

In Fig. 5.7, the heavy metal concentrations in mosses from different reservoirs are compared. It can be noticed that the concentrations of most heavy metals (*e.g.* Cu, Pb and Zn) in mosses from Tingxi Reservoir were much higher than the values in other reservoirs. The concentrations were found to be lowest in the mosses collected at Songtao Reservoir. There was no large difference in concentrations of Ni between the four reservoirs and the Ni contents in mosses were rather low compared to other elements. However, the raw data of metal concentrations in mosses may not directly reflect the relative significance of air pollution in the study area. Different moss species might have varied cation

exchange capacities, which would affect metal accumulation in mosses (Schilling et al., 2002). Fortunately, the mosses collected at Xinfengjiang Reservoir and Songtao Reservoir were of the same species. The results showed that the heavy metal concentrations in mosses from Xinfengjiang Reservoir were much higher (i.e. 1.4 times for Pb, twofold for Cu and Zn, and threefold for Ni) than the values at Songtao Reservoir. This probably reflects that the atmospheric deposition of heavy metals is more significant at Xinfengjiang Reservoir. It was reported that the variations in meteorological and topographical conditions (e.g. temperature, humidity, altitude) might have an influence on metal accumulation in the moss samples (Wolterbeek, 2002). In this study, the meteorological conditions at Songtao Reservoir and Xinfengjiang Reservoir are quite similar. The former one has a tropical monsoon climate with average annual temperature around 23.3 °C, while the later one has a subtropical monsoon climate with the average temperature of 19.6 - 21.3 °C. Both of them have abundant annual rainfalls. These two reservoirs are both located at low altitude, which is around 100-200 m. Considering their similar site conditions, the comparison of moss data between the two reservoirs should be rather reliable.



Figure 5.7 Comparison of concentrations of trace elements Cu, Ni, Pb and Zn in moss samples collected at different reservoirs

5.2.2 Lead isotopic compositions

In order to further investigate the potential impacts of regional atmospheric heavy metal pollution on the air quality in remote areas, the lead isotopic compositions of mosses were determined to provide information on possible origins of lead in mosses. Meanwhile, the collected catchment soils were also analyzed to assess the natural geological conditions of the study area. The lead isotopic ratios of selected sediment samples at the reservoirs were also determined for comparison. The obtained results (in terms of 206 Pb/ 207 Pb and 208 Pb/ 207 Pb) are listed in Table 5.4, compared to the values of available backgrounds (*e.g.* natural bedrocks), potential anthropogenic sources (*e.g.* ore deposits, aerosol samples, and mosses) from previous literature.

Nature of Samples	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb
Tingxi Reservoir		
Sediments $(n = 4)$	1.187 ~ 1.189	2.468 ~ 2.473
Surface soils $(n = 4)$	1.175 ~ 1.184	2.465 ~ 2.474
Mosses $(n = 5)$	1.168 ~ 1.175	2.456 ~ 2.467
<u>Xinfengjiang Reservoir</u>		
Sediments $(n = 9)$	1.196 ~ 1.207	2.470 ~ 2.512
Surface soils $(n = 5)$	1.178 ~ 1.225	2.475 ~ 2.504
Mosses $(n = 9)$	1.173 ~ 1.212	2.459 ~ 2.498
<u>Luokeng Reservoir</u>		
Sediments $(n = 4)$	1.227 ~ 1.234	2.500 ~ 2.533
Mosses $(n = 3)$	1.183 ~ 1.228	2.472 ~ 2.529
<u>Songtao Reservoir</u>		
Sediments (n =9)	1.196 ~ 1.203	2.452 ~ 2.486
Surface soils $(n = 11)$	1.208 ~ 1.224	2.507 ~ 2.568
Mosses $(n = 4)$	1.198 ~ 1.203	2.505 ~ 2.511
North China		
Beijing aerosols (urban) ^a	1.148	2.444
Dalian aerosols (urban) ^a	1.136	2.441
Major Pb-Zn ore deposits ^b	1.091 ~ 1.140	2.425 ~ 2.456
Background ^b	1.040 ~ 1.160	2.373 ~ 2.489
C C		
Yangtze River Delta Region		
Shanghai aerosols (urban) ^a	1.157	2.449
Nanjing aerosols (urban) ^a	1.163	2.460
Major Pb-Zn ore deposits ^b	1.130 ~ 1.164	2.370 ~ 2.458
Background ^b	1.152 ~ 1.170	2.449 ~ 2.456
C C		
Southeast China		
PRD aerosols ^c	1.161 ~ 1.177	2.366 ~ 2.501
Zhongshan aerosols (suburban) ^d	1.171	2.456
Nanling mosses (rural) ^d	1.171 ~ 1.190	2.452 ~ 2.463
Automobile exhausts (PRD) ^e	1.160	2.423
Fankou Pb-Zn deposit ^e	1.172	2.473
Fujian basalts ^f	1.189	2.486
Fujian granites ^f	1.188	2.481

Table 5.4Lead isotopic compositions of the moss, catchment soil and selectedsediment samples at the four reservoirs, compared to potential natural andanthropogenic sources

Nature of Samples	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb	
Southeast China (cont.)			
Volcanic rock (HK) ^g	1.221	2.529	
Volcanic rock (PRD) ^e	1.199	2.497	
Uncontaminated soils (PRD) ^e	1.195	2.482	
South China			
Hainan Au deposits ^h	1.213 ± 0.022	2.482 ± 0.014	
Leizhou basalts ^f	1.190	2.482	
Hainan Basalts ⁱ	1.196 ± 0.004	2.491 ± 0.002	
South Asia			
Vietnam – Hanoi ^j	1.167	2.453	
Vietnam – Ho Chi Minh ^j	1.155	2.430	
Tailand – Bangkok ^j	1.127	2.404	
Malaysia – Kuala Lumpur ^j	1.141	2.410	

a. Mukai *et al.*, 2001; b. Zhu *et al.*,2003; c. Wong *et al.*, 2003; d. Lee *et al.*, 2005; e. Zhu *et al.*, 2001; f. Zhu *et al.*, 1995; g. Duzgoren-Aydin *et al.*, 2004; h. Chang *et al.*, 2002; i. Flower *et al.*, 1991; j. Bollhofer, 2001.

As Zhu *et al.* (1995) previously reported, the ²⁰⁶Pb/²⁰⁷Pb ratios of natural sources in Fujian Province (*i.e.* Fujian basalts and Fujian granites) were around 1.189. Similar results were obtained in the present study, as the upper bound value of the catchment surface soils at Tingxi Reservoir was found to be 1.184. It can be noticed that the lead isotopic compositions of reservoir sediments at Tingxi Reservoir were all kept around the background values, possibly reflecting that the sediments are not contaminated by anthropogenic lead. This is consistent with the generally constant distribution pattern of lead in the sediment profile of Core TX-C as illustrated in Section 5.1.1. However, the ²⁰⁶Pb/²⁰⁷Pb ratios of mosses collected at Tingxi Reservoir were much lower, ranging from 1.168 to 1.175. The results might indicate that there are some anthropogenic inputs of lead in mosses as the lead isotopic ratios of anthropogenic sources are normally much lower than the values of geochemical backgrounds (*e.g.* automobile exhausts in the PRD region, aerosols in Beijing and Shanghai). In Fig. 5.8, the lead isotopic ratios of mosses, soils and sediments are plotted in comparison with the values of natural geological sources. As the figure presents, the samples formed a good linear relationship, further reflecting the concurrent contributions of both natural and anthropogenic sources at the reservoir. Since the uptake of mineral substrates from soils is insignificant, the metal contents in mosses are expected to be mainly derived from precipitation or dry deposition (Ruhling *et al.*, 2004). Therefore, the anthropogenic lead in mosses is probably contributed by atmospheric deposition of air pollutants over the whole catchment of Tingxi Reservoir. Additional investigations are needed, however, to identify the potential sources of anthropogenic heavy metals.



Figure 5.8 Comparisons of Pb isotopic compositions between sediments, mosses, and surface soils at Tingxi Reservoir

The lead isotopic compositions of the various samples collected at Xinfengjiang Reservoir have previously been discussed in Section 4.4. The results suggest that the air quality at Xinfengjiang Reservoir is deteriorated by the strong influences of regional air pollution. Industrial use of lead-containing ores in the PRD region and possible existence of Pb from past usage of leaded gasoline might have contributed substantial amount of anthropogenic lead in mosses and soils in the study area.

Luokeng Reservoir is located in the southwest region of Guangdong Province. Past investigations revealed that the lead isotopic ratios ²⁰⁶Pb/²⁰⁷Pb of geological materials in Guangdong, such as volcanic rocks and natural soils, generally ranged from 1.195 to 1.221 (Zhu *et al.*, 2001; Duzgoren-Aydin *et al.*, 2004). The results from the present study showed that the ²⁰⁶Pb/²⁰⁷Pb ratios of sediment samples were generally in the range of 1.227 ~ 1.234, which were slightly higher than the values of natural sources reported previously in Guangdong Province. Therefore, the sediments at Luokeng Reservoir are not likely to be contaminated significantly by anthropogenic lead, which can also be indicated by the high correlations between lead and major elements in the sediment profiles at the reservoir. The ²⁰⁶Pb/²⁰⁷Pb ratios of sediments thus possibly represent the values of geochemical backgrounds in the study area. The lead isotopic ratios of moss samples varied over a wide range from 1.183 to 1.228. The upper bound value

was similar to those of the natural backgrounds while the lower bound value was lower. This might indicate some inputs of anthropogenic lead in mosses at Luokeng Reservoir. The anthropogenic lead is probably derived from the atmospheric inputs of air pollutants from regional emission sources.

In Hainan Province, several studies have been conducted to examine the lead isotopic compositions of geochemical backgrounds such as Au deposits and basalts. The ²⁰⁶Pb/²⁰⁷Pb ratios of the natural materials generally ranged from 1.190 to 1.213 (Tu et al., 1991; Zhu et al., 1995; Chang et al., 2002). In the present study, the ²⁰⁶Pb/²⁰⁷Pb ratios of surface soils collected at Jianfengling Mountain near the Songtao Resevoir were found to be around 1.208 ~ 1.224, similar to the values reported in previous studies. The ²⁰⁶Pb/²⁰⁷Pb ratios of sediments at Songtao Reservoir were generally around 1.20, as high as the values of the natural sources. The sediments are thus not expected to be contaminated by anthropogenic lead. As previous results also indicated, the lead concentrations along the sediment cores were closely correlated with the major elements, reflecting the natural origin of lead in the sediments. Considering the moss samples, the lead isotopic compositions (*i.e.* 206 Pb/ 207 Pb ratios around 1.20) were also found to be quite similar to the values of reservoir sediments. This might reflect that the lead accumulated in mosses is derived from natural sources. There is thus no obvious sign of deterioration of air quality in the study area (i.e. Songtao Reservoir).

Based on the previous results (i.e. heavy metal distribution patterns, Pearson correlation analysis and lead isotopic analysis), it can be noticed that the sediments at Tingxi and Luokeng Reservoir are not likely to be contaminated by anthropogenic lead. However, the low ²⁰⁶Pb/²⁰⁷Pb ratios of mosses at these two reservoirs probably indicate that there are some anthropogenic inputs of lead from atmospheric deposition at the two reservoirs. Considering Songtao Reservoir, the ²⁰⁶Pb/²⁰⁷Pb ratios of both sediments and mosses were close to the values of natural backgrounds, possibly reflecting both the air quality and the aqueous environment are well conserved in this area. In contrast, the sediments at Xinfengjiang Reservior are significantly enriched by heavy metals, such as Cu, Pb and Zn. The atmospheric depositions of air pollutants from regional anthropogenic sources are considered to be the main contributing sources in this area. In summary, the air quality at Tingxi Reservoir, Luokeng Reservoir, and Xinfengjiang Reservoir is likely to be impaired in recent times, although no obvious contamination sign can be observed in sediment profiles of Tingxi Reservoir and Luokeng Reservoir. The only exception among the four remote reservoirs is Songtao Reservoir. It would be interesting to further investigate the air masses passing through these four reservoirs, which might have an influence on the air quality of the study areas.

5.3 Classification of Influencing Air Masses

Transport paths of air masses potentially influencing the study areas were analyzed by conducting air back-trajectory analysis. The three-day back trajectories ending at the four reservoirs were analyzed every six hours (local time 0, 6, 12, 18) for each day during the period of January 2006 to February 2007.

In Fig. 5.9(a) - (e), the mean clusters of back-trajectories of dominant air masses ending at Tingxi Reservoir are plotted for five representative time periods respectively. It can be seen that the reservoir was periodically influenced by four major air masses at an annual cycle, including 1) NIM – air masses originated from the inland Mongolia area, North and Northeast China; 2) NEC – air masses coming from the northeast region of Asia, advancing southwards along the southeast coastline; 3) SCS – air masses from the South China Sea; 4) SEA – air masses coming from the inland area or coastal regions of Southeast Asia. During January-March of 2006, the reservoir was mainly controlled by the air masses NIM and NEC, which were closely associated with the winter monsoons (Ding, 1994). Since the early April, the air masses SCS began to influence this area, concurrently with the impacts of air masses NEC. In July, the region was mostly dominated by air masses SCS and SEA, subject to strong impacts of Asian summer monsoons. The situation was maintained for about two months until the air masses NEC were back to prevail in late August. Since November till

February of the next year, the region was again markedly influenced by the intrusion of Siberian high from the north inland area of Asia. The periodic fluctuation of dominant air mass at Tingxi Reservoir was highly related to the season changes of climate with Asian monsoon activities. As it can also be noticed that the area was dominated by air masses NIM for nearly half of the year, suggesting its strong influences on the air quality of the study area.



Figure 5.9 Cluster means of back trajectories ending at Tingxi Reservoir during different time periods: (a) Jan. 2006 – Mar. 2006; (b) Apr. 2006 – May. 2006; (c) Jun. 2006 – Jul. 2006; (d) Aug. 2006 – Oct. 2006; (e) Nov. 2006 – Feb. 2007

In Fig. 5.10(a) - (e), the mean clusters of dominant air back-trajectories ending at Xinfengjiang Reservoir during five different time periods are plotted respectively. The seasonal variation of prevailing air masses (*i.e.* NIM, NEC, SCS and SEA) at the reservoir was generally similar to that at Tingxi Reservoir, shifting in accordance with the seasonal advances and withdraws of Asian monsoons. Similar classifications of air masses were reported by Lee *et al.* (2005) in Nanling Mountain located in northern Guangdong.



Figure 5.10 Cluster means of back trajectories ending at Xinfengjiang Reservoir during different time periods: (a) Jan. 2006 – Mar. 2006; (b) Apr. 2006 – May. 2006; (c) Jun. 2006 – Jul. 2006; (d) Aug. 2006 – Oct. 2006; (e) Nov. 2006 – Feb. 2007

In Fig. 5.11, the mean back-trajectories clusters of dominant air masses ending at Luokeng Reservoir during four representative time periods are presented. As the figures show, the reservoir was also influenced by the same prevailing air masses annually, including NIM, NEC, SCS and SEA. Although the seasonal variation patterns were generally similar to the previous two reservoirs, the durations of dominances by different air masses differed a bit. As the reservoir is located to the south of the Tropic of Cancer, the air masses NIM was dominant for only three months (from Nov. 2006 to Jan. 2007), which was much shorter compared to the cases at Tingxi and Xinfengjiang Reservoir. Starting from the February of 2006, the area had been dominated by the air masses NEC and SCS till the May. The area was then mostly controlled by the SCS and SEA during June-August, following which the air masses NEC began to prevail again. The dominance by air masses NEC at Luokeng Reservoir was observed for six months of a year.



Figure 5.11 Cluster means of back trajectories ending at Luokeng Reservoir during different time periods: (a) Feb. 2006 – May. 2006; (b) Jun. 2006 – Aug. 2006; (c) Sep. 2006 – Oct. 2006; (d) Nov. 2006 – Jan. 2007

The mean clusters of dominant air back-trajectories influencing Songtao Reservoir are shown in Fig. 5.12(a) - (d) for four representative time periods respectively. The prevailing air masses and seasonal variation patterns were generally similar to those of Luokeng Reservoir, except that the time period dominated by the air masses NIM was further shortened to two months from December to January.



Figure 5.12 Cluster means of back trajectories ending at Songtao Reservoir during different time periods: (a) Feb. 2006 – May. 2006; (b) Jun. 2006 – Aug. 2006; (c) Sep. 2006 – Nov. 2006; (d) Dec. 2006 – Jan. 2007

In summary, all the four reservoirs are located in typical monsoon regions subject to seasonal variations of the climate. Tingxi Reservoir was dominantly influenced by air masses NIM originated from north inland area of Asia (*e.g.* Mongolia area and North China) and air masses NEC advancing southwards along the southeast coastline. The air masses SCS and SEA prevailed only for short periods and mostly came from the south coastal area. The air quality in this area might be influenced by air pollution from both the YRD region and PRD region. The situation was generally similar at Xinfengjiang Reservoir. However, the reservoir is located at a much shorter distance (about 150 km) from the PRD region than Tingxi Reservoir (over 500 km). Therefore, the impacts of air pollution from intensively industrialized PRD region on the air quality at

Xinfengjiang Reservoir might be more significant. At Luokeng Reservoir, the air masses NEC dominated for over half year time, suggesting strong potential impacts on the air quality at this reservoir. The air masses normally transported along the southeast coastal region, including the PRD region located at 260 km northeast of the reservoir. The intrusion period of air masses NIM during winter monsoons was comparatively shorter. Moreover, this area was also influenced by air mass SEA from inland or coastal regions of Southeast Asian countries. It was generally the same case at Songtao Reservoir except that this area was much farther away from the PRD region (around 560 km) while it was quite near to the Southeast Asian region (350-400 km). Therefore, the air pollutants released in Southeast Asian countries are possibly important contributing sources for potential impairment of air quality at Songtao Reservoir. As mentioned in Section 5.1.3, however, no obvious sign of air pollution was observed at this reservoir. This might be due to other influencing factors, which will be discussed in detail in the following section.

5.4 Other Possible Influencing Factors

As demonstrated in the previous section, the air quality of a specific remote area might be strongly affected by the prevailing air masses, especially those originated from or passing through intensively urbanized or industrialized areas. However, the case of Songtao Reservoir was an exception. On the other hand, many previous investigations revealed that undisturbed sediment cores were capable to assess the long-term atmospheric inputs of heavy metals into the freshwater ecosystems in the remote area (Blais and Kalff, 1993; Brannvall et al., 1999; Renberg et al., 2002; Yang and Rose, 2005; Shotbolt et al., 2006). In the present study, nevertheless, the sediment cores at Tingxi and Luokeng Reservoir did not reflect the recent deterioration of air quality. It is then considered that there might be other influencing factors affecting the atmospheric transport, deposition and accumulation of heavy metals in aqueous sediments, such as the topographical conditions, climate, and specific characteristics of the reservoirs and their catchments. In Table 5.5, the key characteristics of the four remote reservoirs are listed and compared.

	Tingxi	Xinfengjiang	Luokeng	Songtao
	Reservoir	Reservoir	Reservoir	Reservoir
Topographical	Elevated in the	Elevated in the	Elevated in the	Elevated in the
conditions	north	north	northeast	northwest
Characteristic water level (m)	38.96 – 70.85 ^a	93 – 116 ^b	92.38 – 114 ^b	165 – 190
Climate	Subtropical	Subtropical	Tropical	Tropical
	monsoon	monsoon	monsoon	monsoon
Temperature (°C)	21 ^c	19.6 – 21.3 ^d	23 ^e	23.3 ^f
Annual	1059-2011.7 ^c	1694 – 1936 ^d	> 1600 ^e	1896.1 ^g
precipitations (mm)				
Area of reservoir water surface	N.A.	370 ^h	N.A.	130 ^g
(km ²)	- :		L	_
Catchment area (km^2)	100.8 ^{a, 1}	5740 ^{b, n}	77 ^b	1496 ^g
Drainage ratio	N.A.	15.5	N.A.	11.5

 Table 5.5
 Comparison of key characteristics of the four remote reservoirs

a. Fujian College of Water Conservancy and Electric Power Website (assessed in August 2008);

b. Guangdong Bureau of Hydrology Official Website (assessed in August 2008);

c. Fujian Government Website in Agriculture and Technology (assessed in August 2008)

d. Heyuan Government Website in Agriculture (assessed in August 2008)

e. Maoming Bureau of Hydrology Official Website (assessed in August 2008)

f. Danzhou Government Official Website (assessed in August 2008)

g. Yang et al., 2007

h. Ye, 2000;

i. Lv, 2007;

The topographical condition might be the most important factor influencing the transport of air masses and subsequently the entry of air pollutants in the study area. As the air back-trajectories shows, the air quality at Songtao Reservoir was likely to be influenced by polluted air masses coming from the Southeast Asia. Past investigations revealed that the leaded gasoline was still the major source of

anthropogenic Pb in the atmosphere in most of the developing countries in the Southeast Asia, such as Thailand, Vietnam, Sri Lanka and Indonesia (Mukai *et al.*, 1993; Bollhofer and Rosman, 2002). However, it was noticed that Songtao Reservoir was elevated in the northwest region, which might prevent the intrusion of air masses transported from the Southeast Asia. In addition, the impacts from the PRD region might be insignificant due to the long distance (around 560 km). These possibly explain why a good air quality can be preserved at Songtao Reservoir.

Previous results suggest that the air quality at the other three reservoirs, namely Tingxi Reservoir, Xinfengjiang Reservoir, and Luokeng Reservoir, is probably deteriorated recently. Based on the data provided in Table 5.5, the altitude of the Xinfengjiang Reservoir is around 100 m, which is around the same level as Luokeng Reservoir. Compared to these two reservoirs, Tingxi Reservoir is located relatively lower. Nevertheless, the three reservoirs all belong to low-altitude reservoirs that air pollutants from both local and regional (or even by long-range transport) sources are readily to enter and influence the atmosphere over the study area. As the results from sediment profiles of metal concentrations indicated, however, that only the heavy metals in surface sediments of Xinfengjiang Reservoir were significantly elevated while the sediments at the other two reservoirs were not likely to be contaminated. It has been suggested that the fate of atmospheric particles and behavior of heavy metals in the atmosphere might be greatly influenced by some meteorological parameters, such as the temperature, humidity, rainfalls and winds (Querol *et al.*, 1997; Yatin *et al.*, 2000; Ragosta *et al.*, 2002). It can be noticed that the meteorological conditions are generally similar at these three reservoirs, including the monsoon climate, annual average temperature, and amount of precipitations. Therefore, the meteorological conditions are possibly not important influencing parameters.

As mentioned in Chapter 2, the reservoir catchment might play a significant role in deposition and transmission of heavy metals at both the terrestrial drainage basin and the freshwater system due to the specific characteristics of the natural surfaces, such as the soil pH, the abundance of Fe/Mn oxides and organic matters, clay content, cation exchange capacity with metal elements, and type of land use (Dillon *et al.*, 1988; Livett, 1988; Blais and Kalff, 1993; Church *et al.*, 2002; Shotbolt *et al.*, 2005). The contributions of atmospheric heavy metal inputs from the catchment might be appreciable in case of a reservoir with a large drainage ratio. As the data show in Table 5.5, the catchment area of Xinfengjiang Reservoir amounts to 5740 km², nearly 60-fold of the catchment area of Tingxi Reservoirs and over 70-fold of Luokeng Reservoir. Therefore, the transmission efficiency through the catchment might be a potential factor affecting the deposition and accumulation of heavy metals in the freshwaters, which may explain the different phenomena in sediment cores from the three reservoirs (*i.e.* Tingxi Reservoir, Xinfengjiang Reservoir, and Luokeng Reservoir).

The subsequent accumulation and distribution of heavy metals in reservoir sediments involve many complicated reaction processes, which depend notably on the amount of scavenging materials, the affinity of metal elements for a particular ligand and their respective concentrations, as well as the redox potential and water pH (Livett, 1988; Gonzalez-Davila, 1995). Once more information on the aquatic environment is available, the distinct behaviors of heavy metals in the freshwater and sediments could be further investigated, which might be useful for explaining the inadequacy of sediment profiles as the geochemical archive to monitor historical atmospheric inputs of heavy metals at Tingxi Reservoir and Luokeng Reservoir.

CHAPTER 6 – CONCLUSIONS AND RECOMMENDATIONS

6.1 Conclusions

Since the onset of industrialization periods worldwide, the drastic emissions of anthropogenic heavy metals have been endangering the air quality in both the urban and remote areas. Based on a multi-media approach, the present study was conducted to investigate the potential impacts of air pollution on the ecosystems of four typical remote areas (*i.e.* Tingxi Reservoir, Xinfengjiang Reservoir, Luokeng Reservoir, and Songtao Reservoir) along the South China coastal region, one of the most rapidly developing areas in China.

By analyzing the reservoir sediment cores, catchment soils, as well as moss samples, the major findings obtained at Xinfengjiang Reservoir were as follows:

- At Xinfengjiang Reservoir, the trace metal concentrations were significantly elevated in surface sediments, especially for Cu, Pb and Zn. The Cu and Zn concentrations in the surface sediments were over fourfold higher than the values in bottom layers, while Pb concentrations were also elevated almost threefold. The enrichment of Ni was relatively insignificant.
- The distribution patterns of major and trace elements were generally similar in bottom sediments. However, there was a decreasing trend of trace metal

concentrations with increasing depths in upper layers of sediments (0-20 cm) while a uniform pattern was observed for major elements.

- 3) When analyzing the relationships between trace metals and their respective normalizers (*e.g.* Fe), much more data points were found beyond the prediction intervals in the Cu-Fe, Pb-Fe and Zn-Fe plots; while only a few samples were observed above the upper limit in the Ni-Fe plot.
- 4) The type of moss species *Pogonatum contortum* collected in the present study were found to be effective in accumulating airborne heavy metals, thus would be a promising bioindicator for monitoring atmospheric heavy metal pollution.
- 5) By comparison of heavy metal concentrations in mosses between different regions, the atmospheric inputs of air pollutants in remote area of South China were found to be more significant than those in European countries.
- 6) The data set of lead isotopic compositions of mosses and soils presented a good linear relationship between the natural sources and anthropogenic sources.

The results from the other three remote reservoirs (*i.e.* Tingxi Reservoir, Luokeng Reservoir and Songtao Reservoir) when compared to the results at Xinfengjiang Reservoir revealed that:

 There was no obvious contamination sign of sediments at Tingxi Reservoir, Luokeng Reservoir and Songtao Reservoir as the concentrations of trace
elements were closely associated with the lithogenic major elements Al and Fe.

- 2) The Cu, Pb and Zn concentrations in moss samples were found highest at Tingxi Reservoir, followed by Xinfengjiang Reservoir. The lowest heavy metal concentrations were observed in mosses from Songtao Reservoir.
- The lead isotopic compositions of mosses at Songtao Reservoir were similar to the values of natural backgrounds. However, low ²⁰⁶Pb/²⁰⁷Pb ratios (around 1.17) were observed in mosses collected from Tingxi Reservoir and Luokeng Reservoir.
- 4) By air back-trajectory analyses, it was observed that the dominant influencing air masses and their seasonal patterns were generally similar at the four reservoirs. However, Xinfengjiang Reservoir might be subject to strong influences from the PRD region, while Songtao Reservoir was mainly influenced by air masses from South China Sea and Southeast Asia (*i.e.* far away from the PRD region).

Therefore, it can be concluded from this study that:

- At Xinfengjiang Reservoir, the sediments are considered to be anthropogenically contaminated in the upper layers where significant elevation of heavy metal concentrations (*e.g.* Cu, Pb and Zn) occurred.
- The enriched anthropogenic heavy metals in top sediments of Xinfengjiang Reservoir are probably derived from increasing atmospheric deposition of air

pollutants in recent years, which concurrently occurred in both the terrestrial catchment and the reservoir.

- 3) The regional transport of anthropogenic Pb emitted from industrial processes using lead-containing ores and coal combustion (with similar lead isotopic compositions to some of the lead ores) in the PRD region is probably the dominant contributing source at Xinfengjiang Reservoir. The long persistence of past-released Pb, possibly from vehicle exhausts of leaded gasoline, may also contribute to the total anthropogenic lead in the atmosphere.
- 4) The sediments at Tingxi Reservoir, Luokeng Reservoir, and Songtao Reservoir are not likely to be contaminated by anthropogenic heavy metals. However, the low ²⁰⁶Pb/²⁰⁷Pb ratios of mosses at Tingxi Reservoir and Luokeng Reservoir indicate that the atmosphere at the two reservoirs is probably impaired and there are some anthropogenic inputs of Pb in the surface environment via atmospheric deposition.
- 5) The inadequacy of the reservoir sediments as geochemical archives of the environment changes at Tingxi Reservoir and Luokeng Reservoir might be attributed to the low transmission efficiency through the reservoir catchments due to their small catchment areas, which may subsequently affect the deposition and accumulation of airborne heavy metals in the sediments of the two reservoirs. Further investigations need to be conducted to provide additional evidences.

6) The air quality is well preserved at Songtao Reservoir, which might be ascribed to the fact that the specific topographical condition of the reservoir (*i.e.* elevated in the northwest region) prevents the intrusion of polluted air masses coming from the Southeast Asia. Moreover, the impacts from the PRD region are possibly less significant due to the long distance.

6.2 Recommendations

With regard to the major findings obtained from the present study, several recommendations were proposed as follows:

- Strict regulations or guidelines on emissions of air particulates from various anthropogenic sources, especially from industrial activities and vehicle exhaust emissions, should be established and enforced to alleviate further deterioration of air quality in South China.
- 2) As an important freshwater supply reservoir for residents living in the PRD region, the sediment quality at Xinfengjiang Reservoir should be carefully controlled to prevent potential impairment of water quality or adverse impacts on the ecosystems of the reservoir area.
- 3) The air quality in remote areas along the South China coastal region should be closely monitored to assess the significance of trans-boundary outflow fluxes of air pollutants (*e.g.* anthropogenic heavy metals).
- 4) As air pollution has become a global environmental issue, the governments in

adjacent regions or even countries should cooperate and coordinate together to work on effective pollution prevention and control measures.

- 5) The awareness of environmental pollution and its potential adverse impacts should be further promoted among the governments and the general public.
- 6) Considering selection of proper reservoirs as geochemical archives for environmental changes, the reservoirs with large catchment areas available (or large drainage ratios) should be preferred.

6.3 Future research

Based on the findings from this study, some suggestions for future research work are proposed.

1) The variation trend of heavy metal concentrations along the sediment cores at Xinfengjiang Reservoir was clearly shown by the sediment profiles. However, limited quantities of sediment samples did not allow the assessment of the chronology of collected sediment cores (*e.g.* the longest core WLH-D) by ²¹⁰Pb dating analysis. The selected sediment cores WLH-B, WLH-C and WLH-D for discussion were assumed to be accumulated without post-depositional disturbances. It is therefore suggested that some additional sediment cores could be collected for dating analysis, which may help to better interpret the historical variation trend of heavy metal inputs in sediments at Xinfengjiang Reservoir

- 2) In the present study, geochemical normalization of trace metal concentrations to a proper reference element (*i.e.* Fe) was adopted to account for the natural mineralogical and granulometric variations, and assess the magnitude of anthropogenic enrichments of heavy metals in the sediments at Xinfengjiang Reservoir. Nevertheless, the particle size distributions of the sediment samples were not determined. If additional sediment cores could be collected during future research study, it would be very useful to evaluate the size distribution of sediment particles, and assess the associations between metal concentrations and different particle sizes, which may provide further evidence on the influences by natural factors.
- 3) The sediment profiles of heavy metal concentrations at Xinfengjiang Reservoir showed that the heavy metals were significantly enriched by anthropogenic sources in upper layers of sediments, especially for Cu and Zn. While being a major sink for heavy metals, the sediments may also act as potential pollutant sources to the water body and aquatic biota. Therefore, further investigations on chemical partitioning of trace metals (*e.g.* Cu, Pb and Zn) in contaminated sediments of the reservoir by sequential extraction analysis are needed to investigate geochemical phases of trace elements in sediments, and provide information on the their mobility, bio-availability to living organisms, as well as the potential adverse impacts to the aquatic ecosystem.
- 4) The air qualities at the four remote reservoirs were effectively biomonitored

by moss samples in this study. However, most of the mosses collected belong to different species. It was thus infeasible to make direct comparisons of heavy metal concentrations in mosses between different sampling locations to assess the relative significance of atmospheric heavy metal deposition in the study area. It is suggested that a suitable widespread moss species should be selected as the biomarker for extensive and comprehensive moss survey at a national or even continental scale to provide a clearer picture of the current situation of atmospheric heavy metal pollution in remote areas of the South China coastal region. Moreover, additional information can also be obtained by simultaneous collection of aerosol samples in the study area.

5) Regarding the results of sediment cores and mosses at Tingxi Reservoir and Luokeng Reservoir, the sediment cores were found to be inadequate as environmental archives to reflect historical variations of atmospheric heavy metal inputs at these two reservoirs. It would thus be of great value to examine the causes by assessing the metal budgets of the whole reservoir systems, including the reservoir water areas and their catchments. The total atmospheric depositions to the receiving water and the terrestrial catchment area need to be determined by air sampling. Stream discharge and leaching fluxes from the catchment should also be evaluated. In addition, the catchment soil characteristics also need to be investigated. As a result, the retention capacity of the catchment and the net deposition fluxes of atmospheric heavy metals in the freshwater system can be estimated for better understanding of the biogeochemical processes of heavy metals in the whole system.

REFERENCES

- Admiraal, W., Tubbing, G.M.J. and Breebaart, L., 1995. Effects of phytoplankton of metal partitioning in the lower River Rhine. *Water Research*, Vol. 29, pp. 941-946.
- Al-Homaidan, A.A., 2007. Heavy metal concentrations in three species of green algae from the Saudi coast of the Arabian Gulf. *Journal of Food*, *Agriculture & Environment*, Vol. 5, pp. 354-358.
- Al-Khashman, O.A., 2005. Study of chemical composition in wet atmospheric precipitation in Eshidiya area, Jordan. *Atmospheric Environment*, Vol. 39, pp. 6175-6183.
- Allen, A.G., Nemitz, E., Shi, J.P., Harrison, R.M. and Greenwood, J.C., 2001. Size distributions of trace metals in atmospheric aerosols in the United Kingdom. *Atmospheric Environment*, Vol. 35, pp. 4581-4591.
- Aloupi, M. and Angelidis, M.O., 2001. Normalization to lithium for the assessment of metal contamination in coastal sediment cores from the Aegean Sea, Greece. *Marine Environmental Research*, Vol. 52, pp. 1-12.
- An, Z., 2000. The history and variability of the East Asian paleomonsoon climate. *Quaternary Science Reviews*, Vol. 19, pp. 171-187.
- Appleby, P.G. and Oldfield, F., 1992. Application of lead-210 to sedimentation studies. In Ivanovich, M., Harmon, R.S., et al. ed., Uranium-series Diseuilibrium: Application to Earth, Marine, and Environmental Sciences, Clarendon Press, Oxford, pp. 731-778.
- Baba, A. and Kaya, A., 2004. Leaching characteristics of solid wastes from thermal power plants of western Turkey and comparison of toxicity methodologies. *Journal of Environmental Management*, Vol. 73, pp. 199-207.
- Backstrom, M., Bohlin, H., Karlsson, S. and Holm, N.G., 2006. Element (Ag, Cd, Cu, Pb, Sb, Tl and Zn), element ratio and lead isotope profiles in a sediment affected by a mining operation episode during the late 19th century. *Water*, *Air, and Soil Pollution*, Vol. 177, pp. 285-311.
- Bech, J., Poschenrieder, C., Llugany, M., Barcelo, J., Tume, P., Tobias, F.J., Barranzuela, J.L. and Vasquez, E.R., 1997. Arsenic and heavy metal contamination of soil and vegetation around a copper mine in Northern Peru. *The Science of the Total Environment*, Vol. 203, pp. 83-91.
- Bellis, D. J., Satake, K., Inagaki, M., Zeng, J. and Oizumi, T., 2005. Seasonal and long-term change in lead deposition in central Japan: evidence for atmospheric transport from continental Asia. *The Science of the Total Environment*, Vol. 341, pp. 149 – 158.
- Belzile, N., Chen, Y.W., Gunn, J.M. and Dixit, S.S., 2004. Sediment trace metal profiles in lakes of Killarney Park, Canada: from regional to continental influence. *Environmental Pollution*, Vol. 130, pp. 239-248.

- Benjamin, M.M. and Honeyman, B.D., 2000. Trace metals. In Jacobson, M.C., Charlson, R.J., Rodhe, H. and Orians, G.H., ed., *Earth System Science: From Biogeochemical Cycles to Global Change*, Academic Press, San Diego, California, USA., pp. 377-418.
- Benoit, J.M., Fitzgerald, W.F. and Damman, A.W.H., 1998. The biogeochemistry of an ombrotrophic bog: evaluation of use as archive of atmospheric mercury deposition. *Environmental Research*, Section A, Vol. 78, pp. 118-133.
- Berg, T. and Steinnes, E., 1997. Use of mosses (*Hylocomium splendens* and *Pleurozium schreberi*) as biomonitors of heavy metal deposition: from relative to absolute deposition values. *Environmental Pollution*, Vol. 98, pp. 61-71.
- Berresheim, H., 1993. Distribution of atmospheric sulfur species over various wetland regions in the southeastern USA. Atmospheric Environment PART A-General Topics, Vol. 27, pp. 211-221.
- Bi, X., Feng, X., Yang, Y., Qiu, G., Li, G., Li, F., Liu, T., Fu, Z. and Jin, Z., 2006. Environmental contamination of heavy metals from zinc smelting areas in Hezhang County, western Guizhou, China. *Environmental International*, Vol. 32, pp. 883-890.
- Blais, J.M. and Kalff, J., 1993. Atmospheric loading of Zn, Cu, Ni, Cr, and Pb to lake-sediments – the role of catchment, lake morphometry, and physicochemical properties of the elements. *Biogeochemistry*, Vol. 23, pp. 1-22.
- Bollhofer, A. and Rosman, K.J.R., 2001. Isotopic source signatures for atmospheric lead: the Northern Hemisphere. *Geochimica et Cosmochimica Acta*, Vol. 65, pp. 1727-1740.
- Bollhofer, A. and Rosman, K.J.R., 2002. The temporal stability in lead isotopic signatures at selected sites in the Southern and Northern Hemispheres. *Geochimica et Cosmochimica Acta*, Vol. 66, pp. 1375-1386.
- Bradl, H.B., 2005. Sources and origins of heavy metals. In Bradl, H.B., ed., Heavy Metals in the Environment: Origin, Interaction and Remediation, Interface Science and Technology – Volume 6, ELSEVIER Academic Press.
- Brannvall, M.L., Bindler, R., Renberg, I., Emteryd, O., Bartnicki, J. and Billstrom, K., 1999. The Medieval metal industry was the cradle of modern large-scale atmospheric lead pollution in Northern Europe. *Environmental Science and Technology*, Vol. 33, pp. 4391 – 4395.
- Cabada, J.C., Rees, S., Takahama, S., Khlystov, A., Pandis, S.N., Davidson, C.I. and Robinson, A.L., 2004. Mass size distributions and size resolved chemical composition of fine particulate matter at the Pittsburgh supersite. *Atmospheric Environment*, Vol. 38, pp. 3127-3141.
- Canavan, R.W., Van Cappellen, P., Zwolsman, J.J.G., van den Berg, G. A. and Slomp, C.P., 2007. Geochemistry of trace metals in a fresh water sediment: Field results and diagenetic modeling. *The Science of the Total Environment*, Vol. 381, pp. 263-279.

- Cao, T., An, L., Wang, M., Lou, Y., Yu, Y., Wu, J., Zhu, Z., Qing, Y. and Glime, J., 2008. Spatial and temporal changes of heavy metal concentrations in mosses and its indication to the environments in the past 40 years in the city of Shanghai, China. *Atmospheric Environment*, Vol. 42, pp. 5390-5402.
- Chan, L.Y. and Kwok, W.S., 2000. Vertical dispersion of suspended particulates in urban area of Hong Kong. *Atmospheric Environment*, Vol. 34, pp. 4403-4412.
- Chan. C. Y., Chan, L. Y., Lam, K. S., Li, Y. S., Harris, J. M., and Oltmans, S. J., 2002. Effects of Asian air pollution transport and photochemistry on carbon monoxide variability and ozone production in subtropical coastal south China. *Journal of Geophysical Research – Atmospheres*, Vol. 107 (D24), Art. No. 4746.
- Chang, X.Y., Zhu, B.Q., Yu, S.J. and Xia, P., 2002. Application of lead isotopes to geochemical exploration in the Baoban gold deposits, Hainan Province, China. *Geochimica*, Vol. 31, pp. 259-265. <in Chinese>
- Che, X.Z., 2004. A study on ecological environment construction of Heyuan City. *Tropical Geography*, Vol. 24(2), pp. 182-186. <in Chinese>
- Chen, Y.D., 1999. Comprehensive usage and management of water resources at Xinfengjiang Reservoir. *Guangdong Water Resources and Hydropower*, Vol. 03, pp. 6-8. <in Chinese>
- Chen, J., Tan, M., Li, Y., Zhang, Y., Lu, W., Tong, Y., Zhang, G. and Li, Y., 2005. A lead isotopic record of Shanghai atmospheric lead emissions in total suspended particles during the period of phasing out of leaded gasoline. *Atmospheric Environment*, Vol. 39, pp. 1245-1253.
- Cheng, S.Q. and Lam, K.C., 1998. An analysis of winds affecting air pollution concentrations in Hong Kong. *Atmospheric Environment*, Vol. 32, pp. 2559-2567.
- Chillrud, S. N., Bopp, R. F., Simpson, H. J., Ross, J. M., Shuster, E. L., Chaky, D. A., Walsh, D. C., Choy, C. C., Tolley, L. and Yarme, A., 1999. Twentieth century atmospheric metal fluxes into Central Park Lake, New York City. *Environmental Science and Technology*, Vol. 33, pp. 657 662.
- Choi, M.P.K., Ho, S.K.M., So, B.K.L, Cai, Z., Lau, A.K.H. and Wong, M.H., 2008. PCDD/F and dioxin-like PCB in Hong Kong air in relation to their regional transport in the Pearl River Delta region. *Chemosphere*, Vol. 71, pp. 211-218.
- Church, T.M., Scudlark, J.R. and Conko, K.M., 2002. Atmospheric and fluvial sources of trace elements to the Delaware Inland Bays. In Lipnick, R.L., Mason, R.P., Phillips, M.L. and Pittman, C.U., ed., *Chemicals in the Environment: Fate, Impacts, and Remediation*, Oxford University Press, USA, pp. 243-257.
- Colombo, J.C., Landoni, P. and Bilos, C., 1999. Sources, distribution and variability of airborne particles and hydrocarbons in La Plata area, Argentina. *Environmental Pollution*, Vol. 104, pp. 305-314.

- Councell, T.B., Duckenfield, K.U., Landa, E.R. and Callender, E., 2004. Tire-wear particles as a source of zinc to the environment. *Environmental Science and Technology*, Vol. 38, pp. 4206-4214.
- Cubukcu, A. and Tuysuz, N., 2007. Trace element concentrations of soils, plants and waters caused by a copper smelting plant and other industries, Northeast Turkey. *Environmental Geology*, Vol. 52, pp. 93-108.
- Cuong, D.T., Karuppiah, S., Obbard, J.P., 2008. Distribution of heavy metals in the dissolved and suspended phase of the sea-surface microlayer, seawater column and in sediments of Singapore's coastal environment. *Environmental Monitoring and Assessment*, Vol. 138, pp. 255-272.
- Danzhou Government Official Website. See <u>http://www.danzhou.gov.cn</u> (assessed in August 2008). < in Chinese>
- Daskalakis, K.D. and O'Connor, T.P., 1995. Normalization and elemental sediment contamination in the coastal United States. *Environmental Science* and Technology, Vol. 29, pp. 470-477.
- Datar, S.V., Mukhopadhyay, B. and Srivastava, H.N., 1996. Trends in background air pollution parameters over India. *Atmospheric Environment*, Vol. 30, pp. 3677-3682.
- Dauvalter, V., 1994. Heavy-metals in lake-sediments of the Kola-Peninsula, Russia. *The Science of the Total Environment*, Vol. 158, pp. 51-61.
- Deboudt, K., Flament, P., Weis, D., Mennessier, J. and Maquinghen, P., 1999. Assessment of pollution aerosols sources above the Straits of Dover using lead isotope geochemistry. *The Science of the Total Environment*, Vol. 236, pp. 57-74.
- DelValls, T.A., Forja, J.M., Gonzalez-Mazo, E., Gomez-Parra, A. and Blasco J., 1998. Determining contamination sources in marine sediments using multivariate analysis. *Trac-Trends in Analytical Chemistry*, Vol. 17, pp. 181-192.
- Deng, R.C., Xie, S.H. and Chen, S., 2007. Evaluation on current water quality in Songtao Reservoir and analysis of changing trend of its pollution. *Journal of Anhui Agricultural Science*, Vol. 35, pp. 7379-7380. <in Chinese>
- Desboeufs, K.V., Sofikitis, A., Losno, R., Colin, J.L. and Ausset, P., 2005. Dissolution and solubility of trace metals from natural and anthropogenic aerosol particulate matter. *Chemosphere*, Vol. 58, pp. 195-203.
- DesJardins, M.J., Telmer, K. and Beauchamp, S., 2004. Apportioning atmospheric pollution to Canadian and American sources in Kejimkujik National Park, Nova Scotia, using Pb isotopes in precipitation. *Atmospheric Environment*, Vol. 38, pp. 6875-6881.
- Dillon, P.J., Evans, H.E. and Scholer, P.J., 1988. The effects of acidification on metal budgets of lakes and catchments. *Biogeochemistry*, Vol. 5, pp. 201-220.

- Ding, Y., 1994. The short-range fluctuations of monsoons and their association with the major weather events in China. In Ding, Y., ed., Atmospheric Sciences Library – Volume 16: Monsoons over China, Kluwer Academic Publishers, Dordrecht, The Netherlands, pp. 174-243.
- Douglas, G.B. and Adeney, J.A., 2000. Diagenetic cycling of trace elements in the bottom sediments of the Swan River Estuary, Western Australia. *Applied Geochemistry*, Vol. 15, pp. 551-566.
- Duce, R.A., Ray, B.J., Quinn, J.G., Wade, T.L., Piotrowi, S.R. and Olney, C.E., 1972. Enrichment of heavy metals and organic compounds in the surface microlayer of Narragansett Bay, Rhode Island. *Science*, Vol. 176, pp. 161-163.
- Duzgoren-Aydin, N.S., Li, X.D. and Wong, S.C., 2004. Lead contamination and isotope signatures in the urban environment of Hong Kong. *Environment International*, Vol. 30, pp. 209-217.
- Duzgoren-Aydin, N.S., 2007. Sources and characteristics of lead pollution in the urban environment of Guangzhou. *The Science of the Total Environment*, Vol. 385, pp. 182-195.
- Eby, G. N., 2004. The continental environment. In Eby, G.N., ed., Principles of Environmental Geochemistry, Brooks/Cole, Pacific Grove, CA, USA, pp. 313-385.
- Eleftheriadis, K. and Colbeck, I., 2001. Coarse atmospheric aerosol: size distributions of trace elements. *Atmospheric Environment*, Vol. 35, pp. 5321-5330.
- Espinosa, A.J.F., Rodriguez, M.T., de la Rosa, F.J.B. and Sanchez, J.C.J., 2001. Size distribution of metals in urban aerosols in Seville (Spain). *Atmospheric Environment*, Vol. 35, pp. 2595-2601.
- Fang, M., Zheng, M., Wang, F., Chim, K.S. and Kot, S.C., 1999. The long-range transport of aerosols from northern China to Hong Kong – a multi-technique study. *Atmospheric Environment*, Vol. 33, pp. 1803 – 1817.
- Fang, G.C., Chang, C.N., Wu, Y.S., Wang, V., *et al.*, 2000. The study of fine and coarse particles, and metallic elements for the daytime and night-time in a suburban area of central Taiwan, Taichung. *Chemosphere*, Vol. 41, pp. 639-644.
- Fang, G.C., Wu, Y.S., Huang, S.H. and Rau, J.Y., 2005. Review of atmospheric metallic elements in Asia during 2000-2004. *Atmospheric Environment*, Vol. 39, pp. 3003-3013.
- Fang, G.C., Wu, Y.S., Lee, W.J., Chou, T.Y. and Lin, I.C., 2006. Seasonal variation in concentration and metallic constituents of atmospheric particulates near the western coast of central Taiwan. *Toxicology and Industrial Health*, Vol. 22, 193-201.
- Fang, Y.T., Gundersen, P., Mo, J.M. and Zhu, W.X., 2008. Input and output of dissolved organic and inorganic nitrogen in subtropical forests of South China under high air pollution. *Biogeosciences*, Vol. 5, pp. 339-352.

- Farmer, J.G., Eades, L.J., Mackenzie, A.B., Kirika, A. and Bailey-Watts, T.E., 1996. Stable lead isotope record of lead pollution in Loch Lomond sediments since 1630 AD. *Environmental Science and Technology*, Vol. 30, pp. 3080-3083.
- Farmer, J.G., Eades, L.J., Atkins, H. and Chamberlain, D.F., 2002. Historical trends in the lead isotopic composition of archival *Sphagnum* mosses from Scotland (1838-2000). *Environmental Science and Technology*, Vol. 36, pp. 152-157.
- Feng, H., Han, X.F., Zhang, W.G. and Yu, L.Z., 2004. A preliminary study of heavy metal contamination in Yangtze River intertidal zone due to urbanization. *Marine Pollution Bulletin*, Vol. 49, pp. 910-915.
- Fitzgerald, W.F., Engstrom, D.R., Mason, R.P. and Nater, E.A., 1998. The case for atmospheric mercury contamination in remote areas. *Environmental Science & Technology*, Vol. 32, pp. 1-7.
- Fjeld, E., Rognerud, S. and Steinnes, E., 1994. Influence of environmental-factors on heavy-metal concentration in lake-sediments in southern Norway Indicated by path-analysis. *Canadian Journal of Fisheries* and Aquatic Sciences, Vol. 51, pp. 1708-1720.
- Flower, R.J., Politov, S.V., Rippey, B., Rose, N.L., Appleby, P.G. and Stevenson, A.C., 1997. Sedimentary records of the extent and impact of atmospheric contamination from a remote Siberian highland lake. *Holocene*, Vol. 7, pp. 161-173.
- Foster, I.D.L., Charlesworth, S.M. and Keen, D.H., 1991. A comparative study of heavy-metal contamination and pollution in reservoirs in the English midlands, UK. *Hydrobiologia*, Vol. 214, pp. 155-162.
- Fujian Bureau of Geology Official Website. See <u>http://www.fjdkj.gov.cn</u> (accessed August 2008). <in Chinese>
- Fujian Bureau of Statistics Official Website. See <u>http://www.stats-fj.gov.cn/</u> (accessed August 2008). <in Chinese>
- Fujian College of Water Conservancy and Electric Power Website. See http://www.fjsdxy.com/slx/slkweb/dxgk.htm (assessed in August 2008). <in Chinese>
- Fujian Environment Protection Bureau Official Website. See <u>http://www.fjepb.gov.cn/</u> (accessed August 2008). <in Chinese>
- Fujian Government Official Website. See <u>http://www.fujian.gov.cn/</u> (accessed August 2008). <in Chinese>
- Fujian Government Website in Agriculture and Technology. See <u>http://www.fjsp.gov.cn/</u> (accessed in August 2008). <in Chinese>
- Guangdong Bureau of Hydrology Official Website. See <u>http://www.gdsw.gov.cn</u> (accessed August 2008). <in Chinese>
- Guangdong Bureau of Statistics Official Website. See <u>http://www.gdstats.gov.cn/</u> (accessed April 2008). <in Chinese>
- Guangdong Environment Protection Bureau Official Website. See <u>http://www.gdepb.gov.cn/</u> (accessed August 2008). <in Chinese>

- Guangdong Government Official Website. See <u>http://www.gd.gov.cn/</u> (accessed August 2008). <in Chinese>
- Garban, B., Ollivon, D., Jairy, A., Carru, A.M. and Chesterikoff, A., 1999. The role of phytoplankton in pollutant transfer processes in rivers. Example of River Marne (France). *Biogeochemistry*, Vol. 44, pp. 1-27.
- Garcia-Hernandez, J. Garcia-Rico, L., Jara-Marini, M.E., Barraza-Guardado, R. and Weaver, A.H., 2005. Concentrations of heavy metals in sediment and organisms during a harmful algal bloom (HAB) at Kun Kaak Bay, Sonora, Mexico. *Marine Pollution Bulletin*, Vol. 50, pp. 733-739.
- Goldstein, A.H., Millet, D.B., McKay, M., Jaegle, L., Horowitz, L., Cooper, O., Hudman, R., Jacob, D.J., Oltmans, S. and Clarke, A., 2004. Impact of Asian emissions on observations at Trinidad Head, California, during ITCT 2K2. *Journal of Geophysical Research-Atmospheres*, Vol. 109, Art. No. D23S17.
- Golomb, D., Ryan, D., Eby, N., Underhill, J. and Zemba, S., 1997. Atmospheric deposition of toxics onto Massachusetts Bay. I. Metals. *Atmospheric Environment*, Vol. 31, pp. 1349 – 1359.
- Gonzalez-Davila, M., 1995. The role of phytoplankton cells on the control of heavy metal concentration in seawater. *Marine Chemistry*, Vol. 48, pp. 215-236.
- Gouin, T., Harner, T., Blanchard, P. and Mackay, D., 2005. Passive and active air samplers as complementary methods for investigating persistent organic pollutants in the Great Lakes basin. *Environmental Science and Technology*, Vol. 39, pp. 9115 – 9122.
- Guieu, C., Chester, R., Nimmo, M., Martin, J. M., Guerzoni, S., Nicolas, E., Mateu, J. and Keyse, S., 1997. Atmospheric input of dissolved and particulate metals to the northwestern Mediterranean. *Deep-Sea Research II-Topical Studies in Oceanography*, Vol. 44, pp. 655 – 674.
- Guo, H., Lee, S. C. and Ho, K. F., 2003. Particle-associated polycyclic aromatic hydrocarbons in urban air of Hong Kong. *Atmospheric Environment*, Vol. 37, pp. 5307 – 5317.
- Guo, H., Wang, T., Blake, D. R., Simpson, I. J., Kwok, Y. H. and Li, Y. S., 2006. Regional and local contributions to ambient non-methane volatile organic compounds at a polluted rural/coastal site in Pearl River Delta, China. *Atmospheric Environment*, Vol. 40, pp. 2345 – 2359.
- Hagler, G.S.W., Bergin, M.H., Salmon, L.G., Yu, J.Z., *et al.*, 2006. Source areas and chemical composition of fine particulate matter in the Pearl River Delta region of China. *Atmospheric Environment*, Vol. 40, pp. 3802-3815.
- Hainan Government Official Website. See <u>http://www.hainan.gov.cn</u> (accessed August 2008). <in Chinese>
- Han, Y.J., Holsen, T.M., Hopke, P.K., Cheong, J.P., Kim H. and Yi, S.M., 2004. Identification of source locations for atmospheric dry deposition of heavy metals during yellow-sand events in Seoul, Korea in 1998 using hybrid receptor models. *Atmospheric Environment*, Vol. 38, pp. 5353-5361.

- Han, Y.M., Du, P.X., Cao, J.J. and Posmentier, E.S., 2006. Multivariate analysis of heavy metal contamination in urban dusts of Xi'an, Central China. *The Science of the Total Environment*, Vol. 355, pp. 176-186.
- Han, Y.M., Jin, Z.D., Cao, J.J, Posmentier, E.S. and An, Z.S., 2007. Atmospheric Cu and Pb deposition and transport in lake sediments in a remote mountain area, northern China. *Water, Air, and Soil Pollution*, Vol. 179, pp. 167-181.
- Hans, P., 1991. Metal compounds in the atmosphere. In Merian, E., ed., *Metal and Their Compounds in the Environment*, VCH Publishers, New York, NY, pp. 257-286.
- Harmens, H., Buse, A., Buker, P., Norris, D., et al., 2004. Heavy metal concentrations in European mosses: 2000/2001 survey. Journal of Atmospheric Chemistry, Vol. 49, pp. 425-436.
- Harrison, R.M., Deacon, A.R., Jones, M.R. and Appleby, R.S., 1997. Sources and processes affecting concentrations of PM₁₀ and PM_{2.5} particulate matter in Birmingham (U.K.). *Atmospheric Environment*, Vol. 31, pp. 4103-4117.
- Hart, K.M., Isabelle, L.M. and Pankow, J.F., 1992. High-volume air sampler for particle and gas sampling. 1. Design and gas sampling performance. *Environmental Science and Technology*, Vol. 26, pp. 1048-1052.
- Heyuan Government Website in Agriculture. See <u>http://agri.heyuan.gov.cn/</u> (assessed in August 2008). <in Chinese>
- Hjortenkrans, D., Bergback, B. and Haggerud, A., 2006. New metal emission patterns in road traffic environments. *Environmental Monitoring and Assessment*, Vol. 117, pp. 85-98.
- HK Environmental Protection Department Official Website. See <u>http://www.epd.gov.hk/epd/</u> (accessed August 2008)
- HKSAR Government Official Website. See <u>http://www.gov.hk/</u> (accessed August 2008).
- Hsu, S.C., Liu, S.C., Jeng, W.L., Lin, F.J., Huang, Y.T., Lung, S.C.C., Liu, T.H. and Tu, J.Y., 2005. Variations of Cd/Pb and Zn/Pb ratios in Taipei aerosols reflecting long-range transport or local pollution emissions. *The Science of the Total Environment*, Vol. 347, pp. 111-121.
- Huh, C.A. and Chen, H.Y., 1999. History of lead pollution recorded in East China Sea sediments. *Marine Pollution Bulletin*, Vol. 38, pp. 545-549.
- In, H.J. and Park, S.U., 2003. The soil particle size dependent emission parameterization for an Asian dust (Yellow Sand) observed in Korea in April 2002. *Atmospheric Environment*, Vol. 37, pp. 4625 – 4636.
- Injuk, J., Van Grieken, R. and De Leeuw, G., 1998. Deposition of atmospheric trace elements into the North Sea: coastal, ship, platform measurements and model predictions. *Atmospheric Environment*, Vol. 32, pp. 3011 – 3025.
- Ip, C.C.M., Li, X.D., Zhang, G., Farmer, J.G., Wai, O.W.H. and Li, Y.S., 2004. Over one hundred years of trace metal fluxes in the sediments of the Pearl River Estuary, South China. *Environmental Pollution*, Vol. 132, pp. 157-172.

- Ip, C.C.M., Li, X.D., Zhang, G., Wong, C.S.C. and Zhang, W.L., 2005. Heavy metal and Pb isotopic compositions of aquatic organisms in the Pearl River Estuary, South China. *Environmental Pollution*, Vol. 138, pp. 494-504.
- Ip, C.C.M., Li, X., Zhang, G., Wai, O.W.H. and Li, Y., 2007. Trace metal distribution in sediments of the Pearl River Estuary and the surrounding coastal area, South China. *Environmental Pollution*, Vol. 147, pp. 311-323.
- Jackson, B.P., Winger, P.V. and Lasier, P.J., 2004. Atmospheric lead deposition to Okefenokee Swamp, Georgia, USA. *Environmental Pollution*, Vol. 130, pp. 445-451.
- Jacob, D.J., Logan, J.A. and Murti, P.P., 1999. Effect of rising Asian emissions on surface ozone in the United States. *Geophysical Research Letters*, Vol. 26, pp. 2175-2178.
- Jaffe, D., Mahura, A., Kelley, J., Atkins, J., Novelli, P.C. and Merrill, J., 1997. Impact of Asian emissions on the remote North Pacific atmosphere: Interpretation of CO data from Shemya, Guam, Midway and Mauna Loa. *Journal of Geophysical Research-Atmospheres*, Vol. 102, pp. 28627-28636.
- Jaffe, D., Anderson, T., Covert, D., Kotchenruther, R., et al., 1999. Transport of Asian air pollution to North America. Geophysical Research Letters, Vol. 26, pp. 711-714.
- Jaffe, D. McKendry, I., Anderson T. and Price H., 2003. Six 'new' episodes of trans-Pacific transport of air pollutants. *Atmospheric Environment*, Vol. 37, pp. 391-404.
- Jaffe, D., Prestbo, E., Swartzendruber, P., Weiss-Penzias, P., Kato, S., Takami, A., Hatakeyama, S. and Kajii, Y., 2005. Export of atmospheric mercury from Asia. *Atmospheric Environment*, Vol. 39, pp. 3029-3038.
- Kjoller, C., Postma, D. and Larsen, F., 2004. Groundwater acidification and the mobilization of trace metals in a sandy aquifer. *Environmental Science and Technology*, Vol. 38, pp. 2829-2835.
- Klanova, J., Eupr, P., Kohoutek, J. and Harner, T., 2008. Assessing the influence of meteorological parameters on the performance of polyurethane foam-based passive air samplers. *Environmental Science and Technology*, Vol. 42, pp. 550-555.
- Kober, B., Wessels, M., Bollhofer, A. and Mangini, A., 1999. Pb isotopes in sediments of Lake Constance, Central Europe constrain the heavy metal pathways and the pollution history of the catchment, the lake and the regional atmosphere. *Geochimica et Cosmochimica Acta*, Vol. 63, pp. 1293-1303.
- Krishna, A.K. and Govil, P.K., 2005. Heavy metal distribution and contamination in soils of Thane-Belapur industrial development area, Mumbai, Western India. *Environmental Geology*, Vol. 47, pp. 1054-1061.
- Kritz, M., LeRoulley, J.C. and Danielsen, E., 1988. The China clipper fast advective transport of radon-rich air from the Asian boundary-layer to the upper troposphere near California. *Chemical Geology*, Vol. 70, pp. 96 – 96.

- Kubin, E. and Lippo, H., 1996. Atmospheric heavy metal deposition in Finland from 1985 to 1990. *Applied Geochemistry*, Vol. 11, pp. 155-161.
- Lam, G.C.K., Leung, D.Y.C., Niewiadomski, M., Pang, S.W., Lee, A.W.F. and Louie, P.K.K., 1999. Street-level concentrations of nitrogen dioxide and suspended particulate matter in Hong Kong. *Atmospheric Environment*, Vol. 33, pp. 1-11.
- Lantzy, R.J. and Machenzie, F.T., 1979. Atmospheric trace-metals global cycles and assessment of mans impact. *Geochimica et Cosmochimica Acta*, Vol. 43, pp. 511-525.
- Lavilla, I., Filgueiras, A.V., Valverde, F., Millos, J., Palanca, A. and Bendicho, C., 2006. Depth profile of trace elements in a sediment core of a high-altitude lake deposit at the Pyrenees, Spain. *Water, Air, and Soil Pollution*, Vol. 172, pp. 273-293.
- Le Roux, G., Aubert, D., Stille, P., Krachler, M., Kober, B., Cheburkin, A., Bonani, G. and Shotyk, W., 2005. Recent atmospheric Pb deposition at a rural site in southern Germany assessed using a peat core and snowpack, and comparison with other archives. *Atmospheric Environment*, Vol. 39, pp. 6790 – 6801.
- Lee, C.S.L., Li, X., Zhang, G., Peng, X. and Zhang, Li., 2005. Biomonitoring of trace metals in the atmosphere using moss (*Hypnum plumaeforme*) in the Nanling Mountains and the Pearl River Delta, Southern China. *Atmospheric Environment*, Vol. 39, pp. 397-407.
- Lee, C.S., Li, X., Shi, W., Cheung, S.C., Thornton, I., 2006. Metal contamination in urban, suburban, and country park soils of Hong Kong: a study based on GIS and multivariate statistics. *The Science of the Total Environment*, Vol. 356, pp. 45-61.
- Lee, C.S.L., Li, X., Zhang, G., Li, J., Ding, A. and Wang, T., 2007. Heavy metals and Pb isotopic composition of aerosols in urban and suburban areas of Hong Kong and Guangzhou, South China – Evidence of the long-range transport of air contaminants. *Atmospheric Environment*, Vol. 41, pp. 432-447.
- Li, X.D., Wai, O.W.H., Li, Y.S., Coles, B.J., Ramsey, M.H. and Thornton, I., 2000. Heavy metal distribution in sediment profiles of the Pearl River Estuary, South China. *Applied Geochemistry*, Vol. 15, pp. 567-581.
- Lim, M.C.H., Ayoko, G.A., Morawska, L., Ristovski, Z.D., Jayaratne, E.R. and Kokot, S., 2006. A comparative study of the elemental composition of the exhaust emissions of cars powered by liquefied petroleum gas and unleaded petrol. *Atmospheric Environment*, Vol. 40, pp. 3111-3122.
- Lion, L.W., Altmann, R.S. and Leckle, J.O., 1982. Trace-metal adsorption characteristics of estuarine particulate matter: evaluation of contributions of Fe/Mn oxide and organic surface coatings. *Environmental Science and Technology*, Vol. 16, pp. 660-666.

- Liu, W.H., Zhao, J.Z., Ouyang, Z.Y., Soderlund, L. and Liu, G.H., 2005. Impacts of sewage irrigation on heavy metal distribution and contamination in Beijing, China. *Environmental International*, Vol. 31, pp. 805-812.
- Liu, W.X., Li, X.D., Shen, Z.G., Wang, D.C., Wai, O.W.H. and Li, Y.S., 2003. Multivariate statistical study of heavy metal enrichment in sediments of the Pearl River Estuary. *Environmental Pollution*, Vol. 121, pp. 377-388.
- Livett, E.A., 1988. Geochemical monitoring of atmospheric heavy metal pollution: theory and applications. *Advances in Ecological Research*, Vol. 18, pp. 65-177.
- Loska, K. and Wiechula, D., 2003. Application of principal component analysis for the estimation of source of heavy metal contamination in surface sediments from the Rybnik Reservoir. *Chemosphere*, Vol. 51, pp. 723-733.
- Loska, K., Wiechula, D. and Korus, I., 2004. Metal contamination of farming soils affected by industry. *Environment International*, Vol. 30, pp. 159-165.
- Louie, P.K.K., Watson, J.G., Chow, J.C., Chen, A., Sin, D.W.M. and Lau, A.K.H., 2005. Seasonal characteristics and regional transport of PM_{2.5} in Hong Kong. *Atmospheric Environment*, Vol. 39, pp. 1695-1710.
- Lv, X.Q., 2007. The study on water resource protection at Tingxi Reservoir, Tong'an. *Technology in Western China – Ecological Environment*, pp. 23-24. <in Chinese>
- Maoming Bureau of Hydrology Official Website. See <u>http://www.gdmmsw.com/</u> (assessed in August 2008). <in Chinese>
- Mason R.P., Lawson, N.M. and Sheu, G., 2002. The urban atmosphere: an important source of trace metals to nearby waters?. In Lipnick, R.L., Mason, R.P., Phillips, M.L. and Pittman, C.U., ed., *Chemicals in the Environment: Fate, Impacts, and Remediation*, Oxford University Press, USA, pp. 243-257.
- Matsuki, A., Iwasaka, Y., Osada, K., Matsunaga, K., Kido, M., Inomata, Y., Trochkine, D. Nishita, C., Nezuka, T., Sakai, T., Zhang, D. and Kwon, S., 2003. Seasonal dependence of the long-range transport and vertical distribution of free tropospheric aerosols over east Asia: on the basis of aircraft and lidar measurements and isentropic trajectory analysis. *Journal* of Geophysical Research-Atmospheres, Vol. 108, Art. No. 8663.
- McKendry, I.G., Hacker, J.P. and Stull, R., 2001. Long-range transport of Asian dust to the Lower Fraser Valley, British Columbia, Canada. *Journal of Geophysical Research-Atmospheres*, Vol. 106, pp. 18361-18370.
- Mehra, A., Farago, M.E. and Banerjee, D.K., 1998. Impact of fly ash from coal-fired power stations in Delhi, with particular reference to metal contamination. *Environmental Monitoring and Assessment*, Vol. 50, pp. 15-35.
- Monastra, V., Derry, L.A. and Chadwick, O.A., 2004. Multiple sources of lead in soils from a Hawaiian chronosequence. *Chemical Geology*, Vol. 209, pp. 215-231.

- Monna, F., Lancelot, J., Croudace, I.W., Cundy, A.B. and Lewis, J. T., 1997. Pb isotopic composition of airborne particulate material from France and the southern United Kingdom: implications for Pb pollution sources in urban areas. *Environmental Science and Technology*, Vol. 31, pp. 2277-2286.
- Morgan, J.J. and Stumm, W., 1991. Chemical processes in the environment, relevance of chemical speciation. In Merian, E., ed., *Metal and Their Compounds in the Environment*, VCH Publishers, New York, NY, pp. 67-103.
- Mori, I., Nishikawa, M., Tanimura, T. and Quan, H., 2003. Change in size distribution and chemical composition of kosa (Asian dust) aerosol during long-range transport. *Atmospheric Environment*, Vol. 37, pp. 4253 – 4263.
- Morselli, L., Olivieri, P., Brusori, B. and Passarini, F., 2003. Soluble and insoluble fractions of heavy metals in wet and dry atmospheric depositions in Bologna, Italy. *Environmental Pollution*, Vol. 124, pp. 457-469.
- Motelay-Massei, A., Ollivon, D., Tiphagne, K. and Garban, B., 2005. Atmospheric bulk deposition of trace metals to the Seine River Basin, France: concentrations, sources and evolution from 1988 to 2001 in Paris. *Water, Air, and Soil Pollution*, Vol. 164, pp. 119-135.
- Mukai, H., Furuta, N., Fujii, T., Ambe, Y., Sakamoto, K. and Hashimoto, Y., 1993. Characterization of sources of lead in the urban air of Asia using ratios of stable lead isotopes. *Environment of Science and Technology*, Vol. 27, pp. 1347-1356.
- Mukai, H., Tanaka, A. and Fujii, T., 1994. Lead isotope ratios of airborne particulate matter as tracers of long-range transport of air pollutants around Japan. *Journal of Geophysical Research-Atmospheres*, Vol. 99, pp. 3717-3726.
- Mukai, H., Machida, T., Tanaka, A., Vera, Y. P. and Uematsu, M., 2001. Lead isotope ratios in the urban air of eastern and central Russia. *Atmospheric Environment*, Vol. 35, pp. 2783 2793.
- Nakamura, T., Matsumoto, K. and Uematsu, M., 2005. Chemical characteristics of aerosols transported from Asia to the East China Sea: an evaluation of anthropogenic combined nitrogen deposition in autumn. *Atmospheric Environment*, Vol. 39, pp. 1749 1758.
- Nriagu, J.O., 1989. A global assessment of natural sources of atmospheric trace metals. *Nature*, Vol. 338, pp. 47-49.
- Nyangababo, J.T., Henry, L. and Omutange, E., 2005. Heavy metal contamination in plants, sediments, and air precipitation of Katonga, Simiyu, and Nyando wetlands of Lake Victoria Basin, East Africa. *Bulletin of Environmental Contamination and Toxicology*, Vol. 75, pp. 189-196.
- Nziguheba, G. and Smolders, E., 2008. Inputs of trace elements in agricultural soils via phosphate fertilizers in European countries. *The Science of the Total Environment*, Vol. 390, pp. 53-57.

- Outridge, P.M., Evans, R.D., Wagemann, R. and Stewart, R.E.A., 1997. Historical trends of heavy metals and stable lead isotopes in beluga (*Delphinapterus leucas*) and walrus (*Odobenus rosmarus rosmarus*) in the Canadian Arctic. *The Science of the Total Environment*, Vol. 203, pp. 209-219.
- Pacyna, J.M., 1986. Atmospheric trace elements from natural and anthropogenic sources. In Nriagu, J.O. and Davidson, C.I., ed., *Toxic Metals in the Atmosphere*, Wiley, New York.
- Park, J.G. and Curtis, L.R., 1997. Mercury distribution in sediments and bioaccumulation by fish in two Oregon Reservoirs: point-source and nonpoint-source impacted system. *Archives of Environmental Contamination and Toxicology*, Vol. 33, pp. 423-429.
- Parrish, D.D., Hahn, C.J., Williams, E.J., Norton, R.B., Fehsenfeld, F.C., Singh, H.B., Shetter, J.D., Gandrud, B.W. and Ridley, B.A., 1992. Indications of photochemical histories of Pacific air masses from measurements of atmospheric trace species at Point Arena, California. *Journal of Geophysical Research-Atmospheres*, Vol. 97, pp. 15883-15901.
- Petaloti, C., Triantafyllou, A., Kouimtzis, T. and Samara, C., 2006. Trace elements in atmospheric particulate matter over a coal burning power production area of western Macedonia, Greece. *Chemosphere*, Vol. 65, 2233-2243.
- Plant, J.A. and Raiswell, R., 1983. Principles of environmental geochemistry. In Thornton, I., ed., *Applied Environmental Geochemistry*, Academic Press, London, UK, pp. 1-39.
- Praharaj, T. Powell, M.A., Hart, B.R. and Tripathy, S., 2002. Leachability of elements from sub-bituminous coal fly ash from India. *Environmental International*, Vol. 27, pp. 609-615.
- Querol, X., Alastuey, A., Lopez-Soler, A., Mantilla, E. and Plana, F., 1996. Mineral composition of atmospheric particulates around a large coal-fired power station. *Atmospheric Environment*, Vol. 30, pp. 3557-3572.
- Querol, X., Alastuey, A., Puicercus, J.A., Mantilla, E., Ruiz, C.R., Lopez-Soler, A., Plana, F. and Juan, R., 1998. Seasonal evolution of suspended particles around a large coal-fired power station: chemical characterization. *Atmospheric Environment*, Vol. 32, pp. 719-731.
- Querol, X., Alastuey, A., Lopez-Soler, A., Plana, F., Mantilla, E., Juan, R., Ruiz, C.R. and Orden, A.L., 1999. Characterization of atmospheric particulates around a coal-fired power station. *International Journal of Coal Geology*, Vol. 40, pp. 175-188.
- Ragosta, M., Caggiano, R., D'Emilio, M. and Macchiato, M., 2002. Source origin and parameters influencing levels of heavy metals in TSP, in an industrial background area of Southern Italy. *Atmospheric Environment*, Vol. 36, pp. 3071-3087.

- Reimann, C., De Caritat, P., Halleraker, J.H., Finne, T.E., *et al.*, 1997. Regional atmospheric deposition patterns of Ag, As, Bi, Cd, Hg, Mo, Sb and Ti in a 188,000 km² area in the European Arctic as displayed by terrestrial moss samples long-range atmospheric transport vs local impact. *Atmospheric Environment*, Vol. 31, pp. 3887-3901.
- Renberg, I., Brannvall, M.L., Bindler, R. and Emteryd, O., 2002. Stable lead isotopes and lake sediments – a useful combination for the study of atmospheric lead pollution history. *The Science of the Total Environment*, Vol. 292, pp. 45-54.
- Riget, F., Asmund, G. and Aastrup, P., 2000. The use of lichen (*Cetraria nivalis*) and moss (*Rhacomitrium lanuginosum*) as monitors for atmospheric deposition in Greenland. *The Science of the Total Environment*, Vol. 245, pp. 137-148.
- Romic, M. and Romic, D., 2003. Heavy metals distribution in agricultural topsoils in urban area. *Environmental Geology*, Vol. 43, pp. 795-805.
- Rose, N. L. and Rippey, B., 2002. The historical record of PAH, PCB, trace metal and fly-ash particle deposition at a remote lake in north-west Scotland. *Environmental Pollution*, Vol. 117, pp. 121-132.
- Rosen, D. and Dumayne-Peaty, L., 2001. Human impact on the vegetation of South Wales during late historical times: palynological and palaeoenvironmental results from Crymlyn Bog NNR, West Glamorgan, Wales, UK. *The Holocene*, Vol. 11, pp. 11-23.
- Rosman, K.J.R., Ly, C. and Steinnes, E., 1998. Spatial and temporal variation in isotopic composition of atmospheric lead in Norwegian moss. *Environmental Science and Technology*, Vol. 32, pp. 2542-2546.
- Rossini, P., Guerzoni, S., Molinaroli, E., Rampazzo, G., Lazzari, A. D. and Zancanaro, A., 2005. Atmospheric bulk deposition to the lagoon of Venice Part I. Fluxes of metals, nutrients and organic contaminants. *Environment International*, Vol. 31, pp. 959 – 974.
- Ruhling, A. and Tyler, G., 1968. An ecological approach to lead problem. *Botaniska Notiser*, Vol. 121, pp. 321-342.
- Ruhling, A. and Tyler, G., 1969. Ecology of heavy metals a regional and historical study. *Botaniska Notiser*, Vol. 122, pp. 248-259.
- Ruhling, A. and Tyler, G., 2004. Changes in the atmospheric deposition of minor and rare elements between 1975 and 2000 in south Sweden, as measured by moss analysis. *Environmental Pollution*, Vol. 131, pp. 417-423.
- Salma, I., Ocskay, R., Raes, N. and Maenhaut, W., 2005. Fine structure of mass size distributions in an urban environment. *Atmospheric Environment*, Vol. 39, pp. 5363-5374.
- Salomons, W., 1986. Impact of atmospheric inputs on the hydrospheric trace metal cycle. In Nriagu, J.O. and Davidson, C.I., ed., *Toxic Metals in the Atmosphere*, Wiley, New York.

- Saltali, K., Mendil, D.A. and Sari, H., 2005. Assessment of trace metal contents of fertilizers and accumulation risk in soils, Turkey. *Agrochimica*, Vol. 49, pp. 104-111.
- Samara, C. and Voutsa, D., 2005. Size distribution of airborne particulate matter and associated heavy metals in the roadside environment. *Chemosphere*, Vol. 59, pp. 1197-1206.
- Sanchez, J., Marino, N., Vaquero, M.C., Ansorena, J. and Legorburu, I., 1998. Metal pollution by old lead-zinc mines in Urumea River Valley (Basque Country, Spain). Soil, biota and sediment. *Water, Air, and Soil Pollution*, Vol. 107, pp. 303-319.
- Sandroni, V. and Migon, C., 2002. Atmospheric deposition of metallic pollutants over the Ligurian Sea: labile and residual inputs. *Chemosphere*, Vol. 47, pp. 753 764.
- Satake, K., Tanaka, A. and Kimura, K., 1996. Accumulation of lead in tree trunk bark pockets as pollution time capsules. *The Science of the Total Environment*, Vol. 181, pp. 25 30.
- Scheinberg, H., 1991. Copper. In Merian, E., ed., *Metal and Their Compounds in the Environment*, VCH Publishers, New York, NY, pp. 893-908.
- Schiff, K.C. and Weisberg, S.B., 1999. Iron as a reference element for determining trace metal enrichment in Southern California coastal shelf sediments. *Marine Environment Research*, Vol. 48, pp. 161-176.
- Schilling, J.S. and Lehman, M.E., 2002. Bioindication of atmospheric heavy metal deposition in the Southeastern US using the moss *Thuidium delicatulum*. *Atmospheric Environment*, Vol. 36, pp. 1611-1618.
- Schmitt, H.W. and Sticher, H., 1991. Heavy metal compounds in the soil. In Merian, E., ed., *Metal and Their Compounds in the Environment*, VCH Publishers, New York, NY, pp. 311-331.
- Schropp, S.J., Lewis, F.G., Windom, H.L., Ryan, J.D., et al., 1990. Interpretation of metal concentrations in estuarine sediments of Florida using Aluminum as a reference element. *Estuaries*, Vol. 13, pp. 227-235.
- Schwedt, G., 2001. Material cycles. In Schwedt, G., ed., *The Essential Guide to Environmental Chemistry*, John Wiley & Sons, Ltd., Chichester, West Sussex, UK, pp. 2-29.
- Shah, M.H. and Shaheen, N., 2008. Annual and seasonal variations of trace metals in atmospheric suspended particulate matter in Islamabad, Pakistan. *Water, Air, and Soil Pollution*, Vol. 190, pp. 13-25.
- Shaheen, N., Shah, M.H. and Jaffar, M., 2005. A study of airborne selected metals and particle size distribution in relation to climatic variables and their source identification. *Water, Air, and Soil Pollution*, Vol. 164, pp. 275-294.
- Shevchenko, V., Lisitzin, A., Vinogradova, A. and Stein, R., 2003. Heavy metals in aerosols over the seas of the Russian Arctic. *The Science of the Total Environment*, Vol. 306, pp. 11-25.

- Shotbolt, L.A., Thomas, A.D. and Hutchinson, S.M., 2005. The use of reservoir sediments as environmental archives of catchment inputs and atmospheric pollution. *Progress in Physical Geography*, Vol. 29, pp. 337 – 361.
- Shotbolt, L., Hutchinson, S.M. and Thomas, A.D., 2006. Sediment stratigraphy and heavy metal fluxes to reservoirs in the southern Pennine uplands, UK. *Journal of Paleolimnology*, Vol. 35, pp. 305-322.
- Shotbolt, L., Buker, P. and Ashmore, M.R., 2007. Reconstructing temporal trends in heavy metal deposition: assessing the value of herbarium moss samples. *Environmental Pollution*, Vol. 147, pp. 120-130.
- Shotyk, W., Blaser, P., Grunig, A. and Cheburkin, A.K., 2000. A new approach for quantifying cumulative, anthropogenic, atmospheric lead deposition using peat cores from bogs: Pb in eight Swiss peat bog profiles. *The Science* of the Total Environment, Vol. 249, pp. 281-295.
- Siefert, R.L., Johansen, A.M. and Hoffmann, M.R., 1999. Chemical characterization of ambient aerosol collected during the southwest monsoon and intermonsoon seasons over the Arabian Sea: Labile-Fe(II) and other trace metals. *Journal of Geophysical Research-Atmospheres*, Vol. 104, pp. 3511-3526.
- Siegel, F. R., 2002. Sources and origins of the metals. In Siegel, F.R., ed., *Environmental Geochemistry of Potentially Toxic Metals*, Springer-Verlag, Berlin, Germany, pp. 15-44.
- Singh, K.P., Malik, A., Sinha, S., Singh, V.K. and Murthy, R.C., 2005. Estimation of sources of heavy metal contamination in sediments of Gomti River (India) using principal component analysis. *Water, Air, and Soil Pollution*, Vol. 166, pp. 321-341.
- Smith, J.N. and Walton, A., 1980. Sediment accumulation rates and geochronologies measured in the Saguenay Fjord using the Pb-210 dating method. *Geochimica et Cosmochimica Acta*, Vol. 44, pp. 225-240.
- Srivastava, A. and Jain, V.K., 2007. Size distribution and source identification of total suspended particulate matter and associated heavy metals in the urban atmosphere of Delhi. *Chemosphere*, Vol. 68, pp. 579-589.
- Steinnes, E., Solberg, W., Petersen, H.M. and Wren, C.D., 1989. Heavy metal pollution by long range atmospheric transport in natural soils of southern Norway. *Water, Air, and Soil Pollution*, Vol. 45, pp. 207-218.
- Steinnes, E. and Henriksen, A., 1993. Metals in small Norwegian Lakes: relation to atmospheric deposition of pollutants. Water, Air, and Soil Pollution, Vol. 71, pp. 167-174.
- Steinnes, E., Allen, R.O., Peterson, H.M., Rambaek, J.P. and Varskog, P., 1997. Evidence of large scale heavy-metal contamination of natural surface soils in Norway from long-range atmospheric transport. *The Science of the Total Environment*, Vol. 205, pp. 255-266.
- Sweet, C.W., Weiss, A. and Vermette, S.J., 1998. Atmospheric deposition of trace metals at three sites near the Great Lakes. *Water, Air, and Soil Pollution*, Vol. 103, pp. 423-439.

- Szczepaniak, K. and Biziuk, M., 2003. Aspects of the biomonitoring studies using mosses and lichens as indicators of metal pollution. *Environmental Research*, Vol. 93, pp. 221-230.
- Takeda, K., Marumoto, K., Minamikawa, T., Sakugawa, H. and Fujiwara, K., 2000. Three-year determination of trace metals and the lead isotope ratio in rain and snow depositions collected in Higashi-Hiroshima, Japan. *Atmospheric Environment*, Vol. 34, pp. 4525 – 4535.
- Tam, N.F.Y. and Yao, M.W.Y., 1998. Normalization and heavy metal contamination in mangrove sediments. *The Science of the Total Environment*, Vol. 216, pp. 33-39.
- Tan, H., He, J.L. Liang, L., Lazoff, S., Sommer, J., Xiao, Z.F. and Lindqvist, O., 2000. Atmospheric mercury deposition in Guizhou, China. *The Science of the Total Environment*, Vol. 259, pp. 223-230.
- Tasdemir, Y., Kural, C., Cindoruk, S.S. and Vardar, N., 2006. Assessment of trace element concentrations and their estimated dry deposition fluxes in an urban atmosphere. *Atmospheric Research*, Vol. 81, pp. 17-35.
- Tessier, A., Campbell, P. G. C. and Bisson, M., 1979. Sequential extraction procedure for the speciation of particulate trace metals. *Analytical Chemistry*, Vol. 51, pp. 844 – 851.
- Tu, K., Flower, M.F.J., Carlson, R.W., Zhang, M. and Xie, G., 1991. Sr, Nd, and Pb isotopic compositions of Hainan basalts (South China): Implications for a subcontinental lithosphere Dupal source. *Geology*, Vol. 19, pp. 567-569.
- Turner, L.J. and Delorme, L.D., 1996. Assessment of ²¹⁰Pb data from Canadian lakes using the CIC and CRS models. *Environmental Geology*, Vol. 28, pp. 78-87.
- Udom, B.E., Mbagwu, J.S.C., Adesodun, J.K. and Agbim, N.N., 2004. Distributions of zinc, copper, cadmium and lead in a tropical ultisol after long-term disposal of sewage sludge. *Environmental International*, Vol. 30, pp. 467-470.
- Ukonmaanaho, L., Starr, M., Mannio, J. and Ruoho-Airola, T., 2001. Heavy metal budgets for two headwater forested catchments in background areas of Finland. *Environmental Pollution*, Vol. 114, pp. 63-75.
- Ukonmaanaho, L., Nieminen, T. M., Rausch, N., and Shotyk, W., 2004. Heavy metal and arsenic profiles in ombrogenous peat cores from four differently loaded areas in Finland. *Water, Air, and Soil Pollution*, Vol. 158, pp. 277 294.
- Van Meter, P.C. and Callender, E., 1997. Water-quality trends in White Rock Creek Basin from 1912-1994 identified using sediment cores from White Rock Lake Reservoir, Dallas, Texas. *Journal of Paleolimnology*, Vol. 17, pp. 239-249.
- Wang, X., Carmichael, G., Chen, D., Tang, Y. and Wang, T., 2005. Impacts of different emissions sources on air quality during March 2001 in the Pearl River Delta (PRD) region. *Atmospheric Environment*, Vol. 39, pp. 5227-5241.

- Wang, X., Sato, T. and Xing, B., 2006. Size distribution and anthropogenic sources apportionment of airborne trace metals in Kanazawa, Japan. *Chemosphere*, Vol. 65, pp. 2440-2448.
- Wedepohl, K.H., 1991. The composition of the upper earth's crust and the natural cycles of selected metals. Metals in the natural raw materials. Natural Sources. In Merian, E., ed., *Metal and Their Compounds in the Environment*, VCH Publishers, New York, NY, pp. 1-18.
- Weisberg, S.B., Wilson, H.T., Heimbuch, D.G., Windom, H.L. and Summers, J.K., 2000. Comparison of sediment metal:aluminum relationships between the eastern and gulf coasts of the United States. *Environmental Monitoring* and Assessment, Vol. 61, pp. 373-385.
- Weiss, D., Shotyk, W. and Kempf, O., 1999. Archives of atmospheric lead pollution. *Naturwissenschaften*, Vol. 86, pp. 262-275.
- Williams, M.R., Millward, G.E., Nimmo, M. and Fones, G., 1998. Fluxes of Cu, Pb and Mn to the North-Eastern Irish Sea: the importance of sedimental and atmospheric inputs. *Marine Pollution Bulletin*, Vol. 36, pp. 366-375.
- Witt, M., Baker, A.R. and Jickells, T.D., 2006. Atmospheric trace metals over the Atlantic and South Indian Oceans: Investigation of metal concentrations and lead isotope ratios in coastal and remote marine aerosols. *Atmospheric Environment*, Vol. 40, pp. 5435-5451.
- Wolterbeek, B., 2002. Biomonitoring of trace element air pollution: principles, possibilities and perspectives. *Environmental Pollution*, Vol. 120, pp. 11-21.
- Wong, C.S.C., Li, X.D., Zhang G., Qi, S.H. and Peng, X.Z., 2003. Atmospheric deposition of heavy metals in the Pearl River Delta. *Atmospheric Environment*, Vol. 37, pp. 767 – 776.
- Wu, C.X., Qi, S.H., Su, Q.K., Fang, M. and Wang, W., 2006a. Atmospheric deposition of heavy metals to Xinghua Bay, Fujian Province. *Environmental Chemistry*, Vol. 25(6), pp. 781-784. <in Chinese>
- Wu, C. X., Qi, S. H., Fang, M. and S, Q. K., 2006b. Precipitation Characteristics of Heavy Metal in Dustfall to Quanzhou Bay of Fujian Province. *Research* of Environmental Sciences, Vol. 19(6), pp. 27 – 30. <in Chinese>
- Wu, Y., Hou, X., Cheng, X., Yao, S., Xia, W. and Wang, S., 2007. Combining geochemical and statistical methods to distinguish anthropogenic source of metals in lacustrine sediment: a case study in Dongjiu Lake, Taihu Lake catchment, China. *Environmental Geology*, Vol. 52, pp. 1467-1474.
- Wurl, O. and Obbard, J.P., 2004. A review of pollutants in the sea-surface microlayer (SML): a unique habitat for marine organisms. *Marine Pollution Bulletin*, Vol. 48, pp. 1016-1030.
- Xiao, Z., Sommar, J. and Lindqvist, O., 1998. Atmospheric mercury deposition on Fanjing Mountain Nature Reserve, Guizhou, China. *Chemosphere*, Vol. 36, pp. 2191-2200.
- Yang, H., Rose, N.L. and Battarbee, R.W., 2002. Distribution of some trace metals in Lochnagar, a Scottish mountain lake ecosystem and its catchment. *Science of the Total Environment*, Vol. 285, pp. 197-208.

- Yang, H. and Rose, N., 2005. Trace element pollution records in some UK lake sediments, their history, influence factors and regional differences. *Environmental International*, Vol. 31, pp. 63-75.
- Yang, Q., Li. H. and Long, F., 2007. Heavy metals of vegetables and soils of vegetable bases in Chongqing, Southwest China. *Environmental Monitoring* and Assessment, Vol. 130, pp. 271-279.
- Yang, S.W., Wang, L.A., Yue, J.H., Jiao, Y.J., Zhou, S.C. and Mao, Y.Y., 2007. Research of character and actuality of MSW in Songtao Reservoir region, Hainan Province. *Journal of Chongqing University (Natural Science Edition)*, Vol. 30, pp. 123-126. <in Chinese>
- Yatin, M., Tuncel, S., Aras, N.K., Olmez, I., Aygun, S. and Tuncel, G., 2000. Atmospheric trace elements in Ankara, Turkey: 1. factors affecting chemical composition of fine particles. *Atmospheric Environment*, Vol. 34, pp. 1305-1318.
- Ye, S.E., 2000. Protect the water quality of Xinfengjiang Reservoir. *Pearl River Water Transport*, Vol. 09, pp. 48-49. <in Chinese>
- Yeloff, D.E., Labadz, J.C. and Hunt, C.O., 2005. The suitability of a south Pennine (UK) reservoir as an archive of recent environmental change. *Journal of Paleolimnology*, Vol. 34, pp. 339 – 348.
- Yu, B.Z., Yang, X.A., Huang, X.Y. and Yuan, L.Z., 2006. Safety assessment on water resource in Xiamen. *Xiamen Technology*, pp. 25-29. <in Chinese>
- Yuan, Z.L., 2006. New distribution records of Polytrichaceae in Henan. *Journal* of Henan Agricultural University, Vol. 40(5), pp. 527-535. <in Chinese>
- Yue, J.H, Zhang, Y.L., Zhou, H.Y. and Zhao, Y., 2005. The framework of constructing Songtao Reservoir eco-friendly watershed. *China Population*, *Resources and Environment*, Vol. 15, pp. 90-94. <in Chinese>
- Zarcinas, B.A., Ishak, C.F., McLaughlin, M.J. and Cozens, G., 2004. Heavy metals in soils and crops in southeast Asia. 1. Peninsular Malaysia. *Environmental Geochemistry and Health*, Vol. 26, pp. 343-357.
- Zechmeister, H.G., 1995. Correlation between altitude and heavy metal deposition in the Alps. *Environmental Pollution*, Vol. 89, pp. 73-80.
- Zeng, C.S., 2000. Assessment of geotechnical conditions for dam construction at Songtao Reservoir. *Pearl River*, Vol. 6, pp. 28-30. <in Chinese>
- Zhang, X., Zhuang, G., Guo, J., Yin, K. and Zhang, P., 2007. Characterization of aerosol over the Northern South China Sea during two cruises in 2003. *Atmospheric Environment*, Vol. 41, pp. 7821-7836.
- Zhao, Z.Z., Bi, H., Yang, Y.G., Zhu, W.H. and Liu, Q., 2005. Spatial distribution pattern of trace elements contents of latosol in the west of Hainan Island. *Earth and Environment*, Vol. 33, pp. 69-73. <in Chinese>
- Zhu, B., 1995. The mapping of geochemical provinces in China based on Pb isotopes. *Journal of Geochemical Exploration*, Vol. 55, pp. 171-181.
- Zhu, B., Chen, Y. and Peng, J., 2001. Lead isotope geochemistry of the urban environment in the Pearl River Delta. *Applied Geochemistry*, Vol. 16, pp. 409-417.

Zhu, B., Chen, Y. and Chang, X., 2003. Application of Pb isotopic mapping to environment evaluation in China. *Chemical Speciation and Bioavailability*, Vol. 14, pp.49-56.

APPENDIX A – ANALYTICAL RESULTS OF SEDIMENT CORES

Table A-1Metal concentrations (unit: mg/kg, dry weight) in sediment coreTX-A from Tingxi Reservoir.

Depth	Al	Fe	Co	Ni	Ph	Zn
(cm)	1 31	ĽV	0	111	10	2411
1	88300	26400	7.05	6.50	34.8	117
2	96100	27700	7.89	7.05	43.1	121
3	98200	27400	7.50	7.16	39.4	122
4	91800	25000	6.49	6.51	34.0	112
5	89800	24600	6.73	6.59	34.7	108
6	84700	25200	7.17	6.55	41.8	106
7	72600	23900	6.65	5.74	38.4	96.5
8	77300	24600	7.20	6.08	38.4	102
9	71300	23800	7.04	5.86	36.7	100
10	58700	22000	6.12	4.86	35.5	86.5
11	75090	24300	6.92	5.83	46.6	96.6
12	97200	25600	7.22	6.73	52.2	104
13	91000	24100	6.96	6.28	50.2	100
14	87000	24400	7.15	6.36	57.1	109
15	84200	25200	7.45	6.32	62.4	113
16	91500	26600	8.24	6.82	61.4	118
17	88000	24800	7.74	6.87	49.6	107
18	86300	24700	7.65	6.75	43.4	103
19	79300	24000	7.19	6.41	52.1	101
20	86300	24400	8.16	6.57	47.4	104
21	82200	23700	6.91	6.01	53.2	102
22	85100	25100	7.29	6.07	56.1	107
23	86700	25300	7.91	6.33	58.4	109
24	86300	25800	8.27	6.47	60.6	114
25	89500	26000	8.26	6.40	64.1	117
26	87400	25800	8.32	6.41	63.5	118
27	87800	26100	8.33	6.36	64.4	120
28	86600	26000	8.26	6.49	64.5	119
29	87800	26100	8.13	6.36	64.2	119
30	85100	26000	7.86	6.50	65.0	115
31	82000	25900	7.87	6.27	64.1	113
32	79000	25000	7.68	6.20	63	111

Depth	A 1	Fo	Ca	NI;	Dh	7
(cm) A1	AI	ге	CO	INI	ru	ZII
1	103000	29500	8.04	6.73	41.5	135
2	117000	32300	9.76	7.54	59.4	144
3	104000	27600	8.34	6.71	48.4	127
4	128000	31300	7.64	7.92	47.9	151
5	87400	23200	7.11	5.61	36.9	103
6	104000	27700	9.02	6.65	57.5	122
7	86900	26200	8.80	5.68	49.3	110
8	84900	26400	8.21	5.49	45.2	111
9	87900	27100	7.17	6.83	42.4	117
10	81600	24800	8.55	5.61	47.9	107
11	69900	23200	8.95	5.06	47.0	99.8
12	70200	22900	8.06	5.33	48.8	103
13	69200	22500	6.86	5.49	49.1	102
14	74400	23800	8.20	5.72	49.5	107
15	69900	21900	7.45	5.30	50.0	99.3
16	69700	22800	7.43	5.40	49.1	100
17	66200	22100	7.36	5.06	44.9	98
18	71600	23100	7.63	5.48	52.8	104
19	87400	27400	7.58	6.38	53.1	121
20	103000	26700	7.85	6.77	62.0	120
21	91000	23300	6.76	5.91	55.7	99
22	86800	23600	7.26	5.79	62.1	108
23	85600	23900	6.99	5.94	60.3	118
24	85600	24200	7.45	6.00	62.5	120
25	86000	24700	7.87	6.12	66.7	131
26	85200	24900	8.01	6.08	65.0	129
27	106000	28400	7.90	7.18	57.0	123
28	100000	27900	7.28	7.28	41.3	121
29	90900	23400	7.03	5.74	55.3	108
30	94900	25600	8.61	6.29	60.6	114
31	98100	27100	8.48	7.13	63.7	124
32	93700	25500	8.15	7.19	64.2	126
33	92400	26500	8.01	6.64	62.1	118
34	84000	26000	7.94	6.41	58.8	127
35	82100	25500	7.78	6.57	61.3	120
36	89800	28400	8.31	6.63	68.7	125
37	94100	29000	8.02	6.52	62.3	124
38	106000	28600	8.18	6.90	59.8	130
39	103000	27100	7.35	6.98	57.4	132

Table A-2Metal concentrations (unit: mg/kg, dry weight) in sediment coreTX-B from Tingxi Reservoir.

(cont.)

Depth	Al	Fe	Со	Ni	Pb	Zn	
(cm)							
40	98700	26100	7.54	7.09	59.3	129	
41	107000	28600	8.53	7.78	67.2	130	
42	107000	27400	7.92	7.52	54.5	126	
43	96900	26100	8.26	7.08	56.6	117	
44	101000	27900	8.16	7.41	61.0	126	
45	90100	24900	8.02	7.02	55.0	117	
46	94700	28900	7.63	6.91	54.2	116	

Depth (cm)	Al	Fe	Со	Ni	Pb	Zn
1	90500	24800	7.59	7.18	45.4	114
2	83700	23300	7.30	6.78	43.7	110
3	88100	24600	7.86	7.20	47.3	113
4	90800	25000	7.22	7.24	46.7	119
5	94500	26100	8.11	7.58	49.8	121
6	88800	25600	8.23	7.32	49.8	117
7	88300	25400	8.21	7.38	50.0	118
8	87100	25100	8.10	7.25	49.5	120
9	89500	25700	8.52	7.29	49.3	113
10	90900	25700	8.46	7.49	50.5	121
11	92900	26200	7.37	7.73	50.6	126
12	89400	25200	7.80	7.45	49.5	115
13	85500	24500	7.52	7.11	47.1	116
14	88200	24200	6.82	6.36	43.9	109
15	89700	24700	7.05	6.34	45.4	110
16	86600	24100	6.75	6.40	45.3	106
17	89700	24600	6.14	6.53	44.9	109
18	89500	24700	6.56	6.41	44.6	109
19	86000	24000	6.05	6.27	43.4	106
20	86800	24300	6.61	6.13	44.4	105
21	80500	23100	6.91	6.36	43.1	101
22	89000	23200	6.78	5.72	42.8	98.2
23	79500	22900	6.78	5.82	44.2	101
24	77700	23000	6.46	5.68	44.2	106
25	68800	21600	6.77	4.98	41.6	93.5
26	60200	20100	5.43	4.07	36.0	79.5
27	65300	22300	5.75	5.14	42.8	100
28	68500	22500	6.93	5.19	43.6	97.2
29	64400	22000	5.46	4.96	42.0	97.9
30	75500	24000	5.63	5.65	48.5	104

Table A-3Metal concentrations (unit: mg/kg, dry weight) in sediment coreTX-C from Tingxi Reservoir.

Depth (cm)	Al	Fe	Cu	Ni	Pb	Zn	
0.5							
1.0	124000	35600	44.0	17.8	47.3	159	
1.5							
2.0	114000	34800	42.1	15.6	45.3	152	
2.5							
3.0	119000	38500	33.1	12.5	48.1	129	
3.5							
4.0	120000	31500	23.2	10.1	50.9	97	
4.5							
5.0	124000	33600	25.0	12.9	51.9	108	
5.5							
6.0	117000	34600	25.2	15.5	45.3	104	
6.5							
7.0	115000	33800	25.8	13.8	45.2	112	
7.5							
8.0	112000	33800	26.4	14.1	39.2	110	
8.5							
9.0	115000	33900	26.6	14.0	33.6	103	
9.5							
10	125000	33000	19.8	11.3	33.0	88.4	
11	121000	33400	19.7	11.6	35.9	85.9	
12	126000	34400	22.7	13.1	37.0	102	
13	130000	36400	24.5	13.9	36.5	103	
14	125000	35600	22.2	13.7	30.0	95.9	
15	130306	36900	21.3	13.3	31.0	91.4	
16	127172	35400	19.9	13.2	33.5	89.8	
17	132000	38000	24.1	15.8	32.5	96.3	
18	122000	36600	21.3	14.6	28.3	82.1	
19	128000	38000	20.7	13.9	29.4	85.0	
20	109000	36200	20.5	13.3	32.2	77.3	
21	116000	37300	21.0	13.8	30.1	82.1	
22	116000	37500	21.9	14.9	32.3	85.6	
23	112000	36500	22.4	15.4	30.2	85.1	
24	103000	35100	23.0	16.0	28.5	83.7	
25	105000	36300	24.6	17.1	29.4	88.2	
26	83600	31800	28.3	19.3	27.1	85.3	
27	74300	28900	26.7	18.8	30.7	80.7	
28	71400	29200	29.8	19.6	32.6	86.3	
29	72900	27600	29.2	18.2	31.5	83.4	

Table A-4Metal concentrations (unit: mg/kg, dry weight) in sediment coreWLH-A from Xinfengjiang Reservoir.

(cont.)

Depth (cm)	Al	Fe	Со	Ni	Pb	Zn
30	74900	29500	29.5	20.4	26.4	85.6
31	79100	29600	28.0	19.9	23.3	87.7
32	60600	23900	22.7	15.6	21.2	70.5

Depth		Fo	Cu	NJ	Dh	Zn
(cm)	AI	ге	Cu	INI	ΓU	ZII
0.5	118000	32300	41.5	17.0	46.8	146
1.0	115000	34300	41.2	17.8	33.3	145
1.5	115000	34000	60.4	20.3	44.3	186
2.0	120000	32000	31.5	10.4	49.6	124
2.5	114000	38300	28.0	11.2	43.2	119
3.0	121000	36900	33.6	11.7	48.8	134
3.5	115000	36500	29.8	11.3	46.7	129
4.0	125000	32500	36.3	14.4	46.8	130
4.5	127000	32200	26.3	11.7	42.6	116
5.0	127000	33600	18.7	9.8	46.1	101
5.5	124000	33300	22.6	12.7	48.6	109
6.0	128000	33400	23.2	12.8	48.5	114
6.5	124000	35800	22.7	15.5	38.8	106
7.0	125000	34100	27.0	14.5	48.7	117
7.5	114000	35100	27.1	15.7	48.6	117
8.0	118000	34100	26.4	14.9	38.1	121
8.5	129000	33700	19.1	11.7	34.2	101
9.0	116000	34500	29.0	16.0	39.0	115
9.5	120000	33500	23.4	13.8	31.9	107
10	120000	35100	28.0	15.5	31.2	106
11	120000	33600	22.5	13.5	30.2	98.5
12	119000	33300	19.1	13.0	32.1	92.3
13	123000	33900	21.8	13.5	32.5	97.3
14	124000	35200	21.6	14.2	34.4	103
15	108000	34400	19.7	12.9	35.0	87.7
16	101000	32300	20.3	13.5	36.4	84.4
17	114000	32200	19.7	14.0	30.5	85.5
18	113000	33300	19.6	14.5	26.8	81.9
19	112000	31800	17.9	12.5	30.8	79.7
20	100000	28300	16.1	12.0	31.3	74.2
21	93200	27000	17.5	12.8	32.0	72.9
22	81100	23600	15.4	11.4	32.9	65.2
23	71100	21100	16.6	11.5	58.0	66.2
24	45400	11800	7.96	5.65	24.8	38.2
25	49100	10400	6.68	5.19	27.5	38.9
26	50600	10800	7.47	5.65	43.9	43.6

Table A-5Metal concentrations (unit: mg/kg, dry weight) in sediment coreWLH-B from Xinfengjiang Reservoir.

Depth	A 1	Fo	Cu	N;	Dh	Zn
(cm)	AI	ге	Cu	INI	10	Z 11
0.5	114000	36900	55.2	19.7	42.4	159
1.0	106000	39100	57.5	20.7	36.3	172
1.5	113000	38400	39.4	13.3	34.3	128
2.0	120000	38500	36.4	11.8	37.5	129
2.5	117000	33800	40.4	14.2	39.5	143
3.0	106000	34600	30.7	12.0	40.1	100
3.5	123000	34700	21.8	9.81	37.2	91.5
4.0	114000	33500	23.5	10.5	38.9	97.5
4.5	118000	34800	25.6	11.9	38.6	111
5.0	110000	36400	23.3	13.0	30.8	94.5
5.5	110000	37000	32.8	16.3	39.7	113
6.0	119000	37200	31.8	15.2	40.9	126
6.5	118000	37500	30.3	16.0	36.1	127
7.0	124000	38500	24.2	12.3	33.5	112
7.5	126000	36800	23.4	12.7	28.0	114
8.0	111000	36500	34.8	18.2	39.1	129
8.5	124000	36300	20.4	11.8	24.6	98.6
9.0	117000	36900	31.7	16.2	27.4	108
9.5	98400	34500	20.3	12.3	27.4	84.8
10	114000	34600	17.1	11.5	25.7	86.3
11	101000	35800	22.7	11.8	31.2	87.6
12	118000	37100	26.3	15.0	27.3	107
13	122000	36500	22.4	13.6	29.6	107
14	112000	37100	20.8	13.5	28.1	102
15	125000	37400	20.0	13.6	20.9	97.2
16	118000	36000	23.7	15.3	25.3	101
17	116000	36000	21.2	14.3	23.2	90.9
18	109000	36300	19.9	13.3	22.8	85.7
19	95000	33300	15.0	10.3	22.8	76.8
20	77400	30100	11.5	6.35	24.0	71.3
21	59800	25200	10.3	3.69	24.6	64.1

Table A-6Metal concentrations (unit: mg/kg, dry weight) in sediment coreWLH-C from Xinfengjiang Reservoir.

Depth (cm)	Al	Fe	Cu	Ni	Pb	Zn
0.5						
1.0	109000	37900	50.7	21.1	51.1	163
1.5	105000	36000	46.6	17.2	42.5	142
2.0	93700	38800	59.9	22.1	45.7	192
2.5	109000	39200	56.6	20.3	42.6	187
3.0	113000	43900	50.9	17.5	47.1	171
3.5	114000	42000	46.6	15.2	47.3	152
4.0	118000	40000	50.2	16.4	48.3	160
4.5	99800	37800	43.4	16.6	54.5	143
5.0	112000	37500	41.1	14.8	52.9	126
5.5	110000	36900	38.9	15.1	53.8	121
6.0	121000	39000	36.5	16.1	46.1	116
6.5	114000	39600	36.5	17.2	47.0	121
7.0	121000	40600	37.3	17.7	49.9	124
7.5	124000	39800	40.3	17.8	47.7	130
8.0	118000	40400	34.1	16.9	42.0	131
8.5	120000	38200	35.4	14.8	35.9	110
9.0	115000	39000	51.8	18.8	47.1	130
9.5	110000	39700	36.0	16.5	39.5	114
10	122000	39200	39.7	16.1	30.6	105
11	121000	37900	34.5	15.7	32.2	100
12	119000	39300	31.9	15.4	39.0	97.4
13	120000	36300	24.9	15.0	33.2	97.1
14	121000	36300	23.9	14.6	34.8	100
15	119000	36100	22.2	14.3	32.1	94.2
16	120000	36300	22.2	15.0	28.4	86.0
17	112000	35000	22.9	15.7	27.4	81.1
18	97200	31000	20.2	13.4	25.0	68.0
19	95700	30600	21.4	14.7	29.3	65.8
20	64400	24100	20.9	14.3	27.0	60.6
21	22800	10200	8.80	6.50	17.0	33.7
22	20100	9320	8.40	6.00	15.2	27.3
23	17100	7930	6.70	5.00	14.0	26.3
24	16400	9050	6.50	4.70	12.5	23.9
25	18000	8000	7.20	4.90	17.6	25.5
26	44000	15900	11.0	7.90	18.5	41.3
27	97600	30600	19.7	14.2	25.1	70.5
28	77900	25400	17.0	11.3	24.9	55.5
29	78200	26100	19.8	13.8	26.1	64.7

Table A-7Metal concentrations (unit: mg/kg, dry weight) in sediment coreWLH-D from Xinfengjiang Reservoir.
Depth	Al	Fe	Со	Ni	Pb	Zn
(cm)						
30	29100	12000	10.3	7.40	18.4	33.1
31	18400	9260	7.70	5.30	14.5	28.2
32	17000	8020	6.90	5.10	13.3	23.4
33	15600	7680	6.50	4.80	11.9	22.4
34	20900	9380	8.00	6.10	15.2	27.3
35	18900	8790	7.30	5.30	12.8	24.6
36	16300	8050	7.30	5.00	14.8	26.8
37	17000	8230	7.70	5.40	13.7	24.8
38	18600	8670	7.70	5.50	13.8	28.1

Depth	A1	Ea	Cu	NI:	ու	7
(cm)	AI	ге	Cu	INI	PD	LII
1	124000	25700	5.71	3.99	36.2	70.2
2	105000	26400	6.92	4.36	33.8	68.9
3	124000	26500	5.67	4.58	24.1	76.3
4	116000	27500	6.51	4.50	23.9	81.1
5	124000	28200	5.56	4.75	21.5	80.5
6	113000	27800	6.59	4.72	24.8	75.9
7	117000	27200	5.51	4.40	22.9	77.0
8	107000	26500	6.02	4.01	24.0	76.3
9	99300	25800	4.98	3.13	21.9	78.8
10	93400	25400	3.95	2.61	19.9	79.1
11	91600	26000	6.54	2.33	17.7	84.6
12	84900	24800	4.49	2.18	18.4	81.7
13	83300	24700	4.70	2.26	18.1	80.1
14	95200	25700	4.65	2.98	17.7	75.2
15	112000	27600	6.19	4.06	19.7	74.6
16	104000	25700	4.27	3.50	18.6	73.1
17	99700	25700	3.58	2.92	18.0	75.1
18	80800	24800	3.03	2.19	18.5	77.6
19	89700	24800	3.31	2.56	16.4	74.6
20	110000	26500	4.67	4.03	18.5	72.8
21	115000	27300	5.48	4.78	20.6	72.2
22	107000	24400	3.74	3.49	17.9	68.3
23	112000	24900	4.31	3.88	20.9	67.3
24	109000	25200	5.80	3.74	19.8	65.9
25	115000	26200	5.17	4.27	20.9	67.5
26	119000	25700	4.68	3.88	23.0	72.3
27	114000	24800	3.74	3.72	26.5	68.0
28	121000	24900	3.83	3.90	27.7	71.5
29	113000	24600	2.60	3.61	23.4	69.0
30	103000	23500	3.48	3.49	20.3	65.9
31	60500	16600	1.95	1.60	14.4	55.3
32	57600	16000	2.12	1.35	11.2	52.4
33	50700	16200	2.80	1.13	15.3	54.1
34	42800	12300	2.39	0.91	12.5	41.1
35	38500	11000	2.34	0.86	11.7	38.6
36	38900	11200	1.78	1.00	12.2	42.1
37	39700	10900	3.27	1.01	10.8	40.5

Table A-8Metal concentrations (unit: mg/kg, dry weight) in sediment coreLK-A from Luokeng Reservoir.

Depth	Al	Fe	Cu	Ni	Pb	Zn
(cm)				- 1-	- ~	
1	112000	19700	5.67	3.60	37.4	66.8
2	115000	21900	6.66	3.91	40.3	71.9
3	119000	20900	6.17	3.84	38.4	67.8
4	122000	21700	6.29	3.82	35.8	69.3
5	119000	20000	7.95	3.37	39.2	70.2
6	107000	21600	8.03	4.32	32.8	81.7
7	108000	21600	8.37	4.28	32.5	74.5
8	106000	22200	7.67	4.41	32.9	74.4
9	106000	22100	7.30	4.40	31.1	74.3
10	104000	21200	6.91	3.94	31.5	72.3
11	106000	21100	6.78	5.84	33.0	74.4
12	81600	17900	6.34	3.11	29.1	60.7
13	69100	16300	4.97	2.59	26.4	62.3
14	58300	15000	4.73	2.26	23.8	56.2
15	32000	10900	2.96	1.40	13.2	42.8
16	67800	16600	5.29	2.52	22.9	56.1
17	91600	19200	6.26	3.38	29.1	65.6
18	98200	20300	6.25	3.41	31.0	71.0
19	91200	19500	5.96	3.25	30.6	67.7
20	84700	18900	5.85	2.93	30.5	64.9
21	79000	18200	5.52	2.63	28.5	67.1
22	98000	20400	6.42	3.36	33.1	72.2
23	107000	21400	6.72	3.89	32.1	70.4
24	112000	22500	7.12	4.11	33.9	70.0
25	114000	23400	7.08	4.39	34.7	76.2
26	105000	20600	6.26	3.75	33.0	67.5
27	104000	20900	7.17	3.58	34.0	71.2
28	104000	21100	6.99	3.95	34.6	69.8
29	97600	19700	6.37	3.47	32.7	64.7
30	101000	20200	6.43	3.56	35.6	67.5
31	108000	20700	6.58	3.71	36.2	66.5
32	110000	19900	6.58	3.99	34.0	61.9
33	121000	21100	6.60	4.13	35.6	65.8
34	105000	19500	6.33	3.59	36.6	62.2
35	105000	20100	6.17	3.23	36.1	63.8
36	94100	19600	5.80	2.97	34.2	66.2
37	93100	20300	5.92	3.59	32.7	61.8
38	107000	23400	6.35	3.52	35.3	63.4
39	99200	20700	6.13	3.38	33.7	61.4

Table A-9Metal concentrations (unit: mg/kg, dry weight) in sediment coreLK-B from Luokeng Reservoir.

Depth	Al	Fe	Co	Ni	Pb	Zn	
(cm)							
40	109000	21800	6.59	3.79	33.6	63.6	
41	109000	21300	7.82	3.84	33.9	61.9	
42	118000	22400	7.82	4.30	35.5	67.7	
43	124000	21800	7.05	4.01	38.2	69.1	
44	113000	20300	6.91	3.60	39.1	65.2	
45	115000	19300	6.54	3.37	43.4	68.6	
46	124000	19600	6.90	3.71	43.7	68.8	
47	97000	19300	6.30	3.09	31.2	60.0	
48	95500	20200	6.15	3.25	29.0	64.2	
49	74600	18200	5.69	2.66	25.0	62.7	
50	99900	20500	6.34	3.15	34.1	68.8	
51	98600	20600	6.24	3.14	34.3	67.8	
52	106000	21900	6.51	3.32	33.1	67.6	
53	124000	22600	6.22	3.91	33.9	62.2	
54	113000	20900	5.89	2.55	31.3	60.6	
55	92200	21300	5.88	2.33	25.2	57.2	

Depth	A 1	E	Cu	NI:	ու	7
(cm)	AI	re	Cu	INI	PD	Zn
1	131000	22200	6.65	3.57	41.9	76.4
2	134000	24900	6.96	3.78	41.6	81.0
3	128000	21400	5.64	3.35	38.7	69.8
4	132000	23100	6.30	3.50	36.4	73.2
5	130000	22300	6.25	3.31	37.4	71.4
6	133000	20900	8.23	2.95	42.3	74.7
7	121000	22800	8.54	3.81	34.1	84.2
8	129000	24500	8.41	4.23	34.0	90.6
9	114000	22800	7.48	3.86	33.8	82.7
10	114000	23000	7.17	3.74	32.7	80.3
11	89200	19200	5.67	2.84	27.1	63.5
12	111000	21900	6.71	3.58	32.4	73.7
13	108000	21300	6.97	3.33	32.8	70.9
14	103000	21600	6.84	3.07	33.9	71.1
15	75000	17500	5.01	2.26	26.3	63.5
16	71600	17200	4.72	1.98	27.7	64.7
17	28300	9880	2.12	0.63	12.5	34.5
18	24100	10900	2.81	0.73	11.7	34.0
19	31600	11900	2.71	0.82	13.8	41.7
20	60500	15200	4.05	1.58	20.3	49.3
21	98800	20600	5.86	2.87	29.8	70.0
22	107000	21500	6.24	3.15	31.7	72.5
23	108000	21800	5.98	2.89	32.1	75.3
24	97700	20600	5.60	2.49	31.2	74.8
25	96800	21000	6.64	2.46	31.7	74.5
26	84800	18200	5.15	2.17	25.8	60.8
27	109000	21000	5.96	3.11	29.7	67.1
28	109000	20500	6.13	3.12	29.2	66.8
29	107000	20300	6.24	3.14	29.5	62.8
30	116000	21900	6.76	3.50	30.8	67.2
31	107000	20500	6.22	3.20	30.5	63.0
32	101000	19700	6.08	2.88	30.4	61.1
33	91300	18300	5.07	2.38	27.9	61.4
34	85500	17100	4.86	2.19	27.5	57.8
35	94300	17800	4.75	2.39	31.0	59.1
36	90600	17000	4.64	2.29	30.6	57.4
37	93800	17700	4.95	2.37	32.8	60.8
38	104000	18700	5.70	2.84	31.0	60.0
39	121000	20400	5.66	3.17	31.1	67.7

Table A-10Metal concentrations (unit: mg/kg, dry weight) in sediment coreLK-C from Luokeng Reservoir.

Depth	Al	Fe	Со	Ni	Pb	Zn
(cm)						
40	108000	18700	5.43	2.93	34.0	62.8
41	107000	19400	5.27	2.70	34.2	67.0
42	98500	19200	5.09	2.31	34.0	63.1
43	96700	19600	5.17	2.11	36.0	65.3
44	92800	19300	5.19	2.01	34.1	65.4
45	84100	18500	5.07	1.88	32.7	63.4
46	77100	17700	4.18	1.81	29.0	59.8
47	79600	17200	4.59	2.06	25.5	52.4
48	97700	18300	4.84	2.36	31.0	59.9
49	110000	18600	4.82	2.39	33.5	62.7
50	91300	18400	5.96	2.37	26.1	57.6

Depth	A 1	Fo	Cu	NI;	Dh	Zn
(cm)	AI	ге	Cu	111	F D	Z111
1	104000	35200	29.0	18.5	22.9	103
2	81000	29100	21.4	14.7	19.0	84.2
3	89800	32800	27.1	17.5	22.4	92.1
4	81100	30100	24.4	16.5	20.3	87.7
5	72500	29600	24.9	16.8	23.8	90.3
6	75700	30000	25.3	16.9	23.5	94.4
7	75900	30100	24.7	16.7	22.8	93.5
8	73300	29400	24.8	16.5	22.3	90.1
9	77600	31100	26.6	17.3	23.4	93.7
10	76200	30500	25.9	17.1	21.2	89.8
11	89800	37200	32.2	19.5	24.4	102
12	90200	40100	33.7	19.8	27.7	104
13	93200	42400	35.8	20.9	27.5	109
14	96500	41800	35.5	21.7	29.4	115
15	90400	34300	27.6	18.7	24.2	100
16	77200	30700	23.6	16.9	22.3	93.0
17	84400	35600	30.7	19.7	25.7	104
18	98900	40900	36.1	22.4	28.8	121
19	86300	32500	27.6	18.8	25.4	99.6
20	70900	28600	22.7	16.4	24.2	88.5
21	56000	24700	19.7	14.4	22.4	76.2
22	57000	25000	19.6	14.3	20.5	75.8
23	81900	34900	31.1	20.4	27.7	98.4
24	76800	31000	25.3	17.6	24.0	91.9
25	71500	29400	24.2	16.4	25.5	92.8
26	76100	31300	25.4	17.9	25.2	93.2
27	75600	31700	26.7	17.4	24.5	94.4
28	77700	37800	33.6	20.5	26.1	95.1
29	88000	39300	31.5	20.1	27.2	101
30	101000	40000	37.7	24.5	34.2	114
31	101000	41300	33.9	22.8	32.0	114
32	104000	41200	33.6	22.2	28.6	111
33	105000	42300	35.2	23.9	29.1	116
34	101000	40300	36.0	24.4	29.7	111
35	95000	35500	30.9	22.0	27.1	104
36	74100	32300	27.1	19.5	26.5	96.7
37	75800	32900	28.1	19.5	27.5	99.0
38	84600	35000	30.2	21.3	28.1	105
39	75600	33900	28.8	20.5	26.4	96.3

Table A-11Metal concentrations (unit: mg/kg, dry weight) in sediment coreST-A from Songtao Reservoir.

Depth	Al	Fe	Со	Ni	Pb	Zn	
(cm)			2.5				
40	97100	44000	38.9	24.9	29.2	107	
41	106000	45400	35.7	23.3	29.9	106	
42	93300	42400	34.6	23.5	29.5	108	
43	108000	44100	36.3	24.7	29.5	117	
44	96800	39300	32.7	23.5	29.6	111	
45	88200	35100	28.4	21.1	27.9	103	
46	79600	32400	26.5	19.1	25.1	94.0	
47	80700	33800	27.3	20.0	28.7	97.5	
48	78300	33400	27.7	19.4	28.2	96.4	
49	77600	33400	26.0	18.9	26.8	92.8	
50	80300	34100	29.4	20.5	27.4	95.2	
51	96700	42500	38.0	25.2	28.6	102	
52	107000	46600	43.5	27.9	29.5	108	
53	107000	42800	39.8	26.4	33.6	116	
54	110000	44500	42.4	26.9	31.8	114	
55	102000	39800	37.9	24.2	30.2	111	
56	110000	44700	42.2	26.7	30.7	111	
57	99000	38600	34.2	24.2	27.7	102	

Depth	A 1	Fo	Cu	N;	Dh	7n
(cm)	AI	ге	Cu	INI	10	Z 411
1	110000	41500	33.3	22.6	25.2	111
2	107000	39000	32.9	22.9	24.7	107
3	83500	34300	31.5	21.2	28.7	104
4	78600	33700	29.5	20.4	27.1	101
5	89200	36100	30.2	21.6	29.5	109
6	80300	33400	27.5	20.2	27.7	103
7	81700	33500	28.9	20.3	27.0	106
8	82300	34800	29.0	20.4	27.9	103
9	85400	34300	29.4	20.6	28.2	109
10	78800	33100	29.3	20.6	28.1	103
11	84800	35000	30.0	21.2	29.5	107
12	77200	33100	28.0	20.1	26.9	101
13	88800	35400	30.4	21.5	29.4	113
14	81600	33500	29.7	21.0	28.9	107
15	83100	34400	30.8	21.1	29.0	109
16	82500	34400	30.8	21.4	29.6	112
17	80700	33500	28.9	20.6	28.9	106
18	84100	34200	31.0	21.3	30.0	110
19	81100	34700	29.5	20.9	27.7	106
20	89200	35900	31.0	21.6	29.3	113
21	86100	34700	30.4	20.9	29.1	109
22	86100	34800	30.2	20.8	29.7	105
23	83900	34300	29.4	21.1	29.3	106
24	75900	32900	28.8	19.7	28.0	100
25	82800	34200	29.4	20.2	28.1	108
26	79700	33100	28.4	19.7	27.0	101
27	77000	32900	27.6	19.6	27.1	98.9
28	72700	32400	26.6	19.0	25.8	94.8
29	57600	26000	20.1	15.5	21.1	75.5
30	77200	40200	32.7	23.8	27.8	97.1
31	93400	52500	46.4	30.8	32.5	120
32	92500	60600	45.9	28.0	32.7	115
33	107000	41900	34.7	23.8	27.9	113
34	103000	45400	41.5	28.5	31.1	120
35	97800	49600	46.5	31.5	30.9	112
36	99500	56900	51.3	32.7	35.1	119
37	106000	45200	43.6	29.7	30.9	112
38	97600	47600	47.9	32.8	30.2	111
39	94900	38200	37.0	25.4	31.2	115

Table A-12Metal concentrations (unit: mg/kg, dry weight) in sediment coreST-B from Songtao Reservoir.

Depth	Al	Fe	Со	Ni	Pb	Zn	
(cm)							
40	87000	35900	32.1	22.2	30.5	109	
41	83600	34300	31.0	21.9	29.3	108	
42	79000	32900	30.2	20.8	29.4	104	
43	81900	33100	30.0	21.4	29.2	103	
44	77200	32200	28.6	20.4	27.2	102	
45	76800	32700	29.9	21.0	28.4	103	
46	79200	32200	29.1	20.8	28.5	104	
47	73300	32200	29.1	20.5	28.4	102	
48	80800	33100	29.3	20.8	28.0	103	
49	75300	33000	29.4	20.7	28.4	102	
50	78400	32800	28.6	20.7	28.9	103	

Depth	A 1	Fo	Cu	N;	Dh	7n
(cm)	AI	ге	Cu	INI	ΓU	Z111
1	97400	34300	24.3	17.2	13.0	101
2	91700	34800	26.8	18.1	15.9	103
3	93200	40100	31.8	20.9	21.6	115
4	97100	43200	35.7	21.7	19.2	123
5	86100	41200	33.3	21.0	22.9	116
6	84600	36000	26.3	18.5	16.5	103
7	89600	36100	27.4	19.2	15.5	107
8	72300	34400	30.0	19.6	24.4	103
9	73400	34400	29.7	19.5	23.9	102
10	98000	37200	31.5	21.1	21.0	110
11	98400	37100	30.7	21.4	21.6	113
12	98700	38100	32.0	21.4	23.5	113
13	88900	37000	32.6	21.5	22.1	113
14	96300	39400	33.9	22.0	20.1	114
15	99100	41400	34.9	21.8	16.7	111
16	97200	41500	31.0	21.1	17.1	107
17	88000	40700	32.7	21.2	21.5	109
18	93900	38500	30.4	21.3	20.8	110
19	81500	37200	34.6	22.1	22.4	108
20	98900	38700	32.4	22.0	16.7	108
21	105000	41200	36.4	23.5	18.3	112
22	98200	39600	34.8	23.0	19.9	112
23	97200	38500	33.5	22.6	21.7	112
24	97800	38300	35.2	22.8	19.4	112
25	101000	38900	35.4	22.4	16.2	108
26	81600	39300	36.3	22.9	21.9	106
27	99500	38100	32.4	20.1	14.3	101
28	95800	36500	29.9	19.8	14.3	101
29	89900	35600	29.6	19.8	14.8	95.7
30	95000	35800	29.9	19.6	13.5	98.0
31	90100	35000	30.7	20.0	13.6	95.1
32	95000	35300	29.7	19.4	14.9	93.5
33	90100	32700	25.7	18.2	14.0	83.0
34	85900	30900	24.7	17.4	13.2	80.9
35	79600	29500	23.4	16.6	14.2	80.6
36	72200	28200	20.4	15.3	14.2	75.2
37	72900	29900	23.4	17.0	14.9	83.0
38	70700	29800	21.0	16.3	14.4	83.8
39	83000	34600	26.5	18.5	12.2	98.2

Table A-13Metal concentrations (unit: mg/kg, dry weight) in sediment coreST-C from Songtao Reservoir.

Depth	Al	Fe	Со	Ni	Pb	Zn	
(cm)							
40	85000	35300	27.0	18.9	12.6	98.9	
41	88800	36400	27.3	19.0	12.5	102	
42	92400	37700	28.8	19.3	10.7	103	
43	94000	38700	29.8	19.8	11.3	107	
44	92700	38500	29.2	19.6	12.1	105	
45	90700	37700	27.4	19.0	12.1	103	
46	82200	35500	25.9	17.9	11.9	94.5	
47	67300	31600	22.4	16.3	14.4	83.0	
48	69900	30400	23.2	16.5	12.1	85.2	
49	67500	29200	21.2	15.4	14.0	81.7	
50	67900	28700	20.6	15.2	15.0	81.8	
51	56200	24500	16.5	12.8	13.4	74.4	

Depth	A 1	Ea	C	NI:	ու	7
(cm)	AI	re	Cu	INI	PD	Zn
1	94600	36800	32.1	17.4	19.0	119
2	95200	36100	29.6	17.1	19.1	114
3	96500	35900	29.0	16.2	21.4	118
4	93900	36200	27.4	16.6	25.2	117
5	90500	35500	29.9	17.6	24.7	120
6	93200	35900	28.8	17.2	24.7	117
7	98000	35800	31.2	17.9	26.2	121
8	95700	36500	29.8	17.1	24.1	117
9	87400	39400	33.6	18.4	24.9	116
10	90900	40500	34.2	19.4	25.1	119
11	102000	38000	30.0	17.2	23.0	119
12	95600	35300	29.4	16.1	24.7	118
13	90200	36400	31.3	17.5	23.6	117
14	97200	38100	35.1	19.1	20.0	120
15	88000	37500	34.8	19.2	22.0	114
16	86800	36000	32.4	19.1	21.3	111
17	89400	35900	32.0	19.5	18.3	110
18	91900	35800	31.9	19.5	15.6	108
19	90400	37000	33.0	20.2	18.5	109
20	102000	37600	34.2	20.6	15.3	114
21	87000	37000	32.5	19.8	17.8	108
22	95200	38000	32.6	19.8	17.0	113
23	80000	33400	26.4	15.4	17.7	102
24	77200	34400	25.9	14.9	19.6	105
25	76400	33100	28.0	15.7	21.6	110
26	68900	28000	22.9	11.2	17.8	87.0
27	93800	30800	27.0	13.6	19.3	110
28	76500	27600	22.3	12.6	19.0	92.4
29	83400	28800	21.5	12.7	16.1	95.0
30	36600	17200	14.9	7.26	17.5	58.2
31	41000	17400	14.4	7.02	16.1	58.8
32	41400	16500	13.5	6.78	14.8	55.1
33	58800	20700	17.0	9.47	15.3	70.4
34	52000	19300	16.2	7.77	18.1	68.1
35	70300	26400	28.7	12.7	23.5	103
36	68000	23800	23.2	11.9	24.0	97.7
37	78700	25800	27.3	13.2	24.7	110
38	48600	18500	18.2	8.39	18.7	72.8
39	31800	14100	13.1	5.93	13.2	56.0

Table A-14Metal concentrations (unit: mg/kg, dry weight) in sediment coreST-D from Songtao Reservoir.

Depth	Al	Fe	Со	Ni	Pb	Zn	
(cm)							
40	48800	20100	19.1	8.04	16.8	88.0	
41	51400	20700	19.4	8.55	16.8	84.5	
42	48500	20400	21.4	8.53	18.4	81.9	
43	45700	18900	18.5	8.04	18.2	74.4	
44	41200	17500	16.9	7.51	17.2	65.0	
45	30800	16400	17.8	7.04	18.5	60.2	

	Pb Iso	topic Mean	Ratio		RSD (%)			
Sample Name	²⁰⁴ Pb/ ²⁰⁷ Pb	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb	²⁰⁴ Pb/ ²⁰⁷ Pb	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb		
TX-A-Sd4	0.06414	1.18925	2.46972	0.4999	0.2486	0.2600		
TX-A-Sd14	0.06399	1.18800	2.46790	0.5187	0.2697	0.2548		
TX-A-Sd20	0.06406	1.18816	2.47300	0.4177	0.2997	0.3003		
TX-A-Sd30	0.06392	1.18746	2.46880	0.5273	0.3866	0.3130		
WLH-B-Sd3	0.06365	1.19764	2.48494	0.5976	0.3713	0.3575		
WLH-B-Sd11	0.06375	1.20317	2.49587	0.4191	0.2497	0.3470		
WLH-B-Sd21	0.06397	1.20110	2.48358	0.5481	0.3559	0.3637		
WLH-B-Sd33	0.06365	1.20735	2.51221	0.6053	0.3854	0.4189		
WLH-B-Sd36	0.06352	1.20665	2.51114	0.3737	0.2992	0.2379		
WLH-C-Sd2	0.06385	1.19673	2.46986	0.4131	0.3359	0.2923		
WLH-C-Sd13	0.06399	1.19561	2.47133	0.5061	0.3160	0.2686		
WLH-C-Sd20	0.06408	1.20224	2.47330	0.3923	0.2905	0.0954		
WLH-C-Sd30	0.06392	1.20194	2.48816	0.3631	0.2703	0.2324		
LK-A-Sd2	0.06393	1.22679	2.49962	0.2510	0.3002	0.2133		
LK-A-Sd11	0.06380	1.22976	2.51551	0.4608	0.1877	0.2916		
LK-A-Sd27	0.06373	1.22961	2.51283	0.3750	0.1948	0.1663		
LK-A-Sd36	0.06344	1.23422	2.53317	0.3707	0.2690	0.2276		
ST-C-Sd2	0.06420	1.19579	2.46015	0.4382	0.2205	0.1934		
ST-C-Sd21	0.06389	1.20016	2.47307	0.6559	0.2369	0.1620		
ST-C-Sd32	0.06389	1.19727	2.47414	0.3900	0.2356	0.1462		
ST-C-Sd49	0.06417	1.20037	2.47038	0.4170	0.2623	0.2018		
ST-D-Sd2	0.06436	1.19950	2.45242	0.2779	0.1504	0.2189		
ST-D-Sd8	0.06423	1.20224	2.45838	0.4805	0.2536	0.2062		
ST-D-Sd20	0.06438	1.20298	2.45860	0.3896	0.2820	0.2009		
ST-D-Sd37	0.06420	1.20133	2.46813	0.4502	0.2630	0.1973		
ST-D-Sd44	0.06411	1.19748	2.48635	0.3040	0.2587	0.1706		

Table A-15Lead isotopic compositions of selected sediment samples from thereservoirs.

APPENDIX B – ANALYTICAL RESULTS OF SURFACE SOILS & MOSSES

Table B-1Concentrations (unit: mg/kg, dry weight) of surface soils from thecatchment of the reservoirs.

Sample Name	Al	Fe	Cu	Ni	Pb	Zn
TX 100	23200	8910	32.5	5.40	47.6	102
TX 150	18800	6540	44.2	5.02	53.2	110
TX 250	15100	5000	37.7	3.85	38.2	86.8
TX 290	24600	5570	20.4	3.00	23.0	52.1
WLH 200	42100	16500	6.70	3.37	10.4	32.4
WLH 290	47100	9750	3.26	0.08	27.9	63.5
WLH 400	31800	11100	3.64	6.85	9.95	22.0
WLH 490	58800	21500	13.9	7.81	8.65	55.9
WLH 550	66200	26800	7.02	6.98	19.0	78.4
WLH 615	88000	45100	8.69	12.9	0.00	105
ST - I	58400	14600	1.51	2.37	1.36	26.8
ST - II	57100	13900	1.51	2.28	2.64	27.9
$\mathrm{ST}-\mathrm{III}$	56800	14100	1.28	2.35	3.08	25.9
ST - IV	49500	12200	1.29	2.14	3.02	24.1
ST – JFL no. 1	44900	11500	3.00	0.08	28.7	24.6
ST – JFL no. 2	64100	17200	8.69	0.00	19.1	49.3
ST – JFL no. 3	36400	8440	1.66	0.00	6.27	20.5
ST – JFL no. 4	45200	10800	2.93	0.82	7.26	26.8
ST – JFL no. 5	39600	9670	2.56	0.66	6.22	24.7
ST – JFL no. 6	63300	12100	1.80	1.91	26.5	25.2

Sample Name	Name of Moss Species	Cu	Ni	Pb	Zn
XMTX - A	Hypnum plumaeforme Wils.	47.2	3.46	65.0	199
XMTX - B	Hypnum plumaeforme Wils.	32.1	3.10	53.9	163
XMTX - C	Hypnum plumaeforme Wils.	34.2	3.58	61.8	174
XMTX - D	Hypnum plumaeforme Wils.	68.1	5.48	74.3	180
XMTX - E	Entodon macropodus(Hedw.)C.Muell.	42.2	3.80	38.3	125
WLH-GS 200	Pogonatum contortum(Brid.)Lesq.	7.92	2.77	34.3	164
WLH-GS 290	Pogonatum contortum(Brid.)Lesq.	9.27	4.00	116	152
WLH-GS 400	Pogonatum contortum(Brid.)Lesq.	7.15	5.73	41.0	103
WLH-GS 490	Pogonatum contortum(Brid.)Lesq.	7.93	3.30	45.9	82.0
WLH-GS 550	Pogonatum contortum(Brid.)Lesq.	8.18	3.81	56.2	132
WLH-GS 600	Plagiochila dendroides(Nees)Lindenb.	12.7	5.43	36.7	164
WLH-GS 615	Pogonatum contortum(Brid.)Lesq.	9.55	4.55	23.0	111
WLH-GS 700	Thudium cymbifolium(Doz.et Molk.)Doz.et Molk.	11.0	7.00	23.7	78.6
WLH-GS 760	Thudium cymbifolium(Doz.et Molk.)Doz.et Molk.	21.2	8.43	47.8	216
LK270	<i>Thuidium glaucinum(Mitt.)Bosch. et Scande Lac.</i>	11.3	1.86	58.6	108
LK340	Claopodium assurgens(sull.et Lesq.)Card.	8.50	1.62	21.8	80.5
LK360	Claopodium aciculum(Broth.)Broth.	7.89	1.42	28.9	70.7
ST - I	Pogonatum contortum(Brid.)Lesq.	5.50	1.78	29.8	59.0
ST - II	Pogonatum contortum(Brid.)Lesq.	5.72	1.69	36.9	60.9
$\mathrm{ST}-\mathrm{III}$	Pogonatum contortum(Brid.)Lesq.	5.99	1.45	35.2	61.2
ST - IV	Pogonatum contortum(Brid.)Lesq.	5.79	1.68	37.3	58.3

Table B-2Concentrations (unit: mg/kg, dry weight) of moss samples collectedfrom the catchment of the reservoirs.

	Ph Iso	tonic Mean	Ratio		RSD (%)		
Sample Name	²⁰⁴ Pb/ ²⁰⁷ Pb	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb	²⁰⁴ Pb/ ²⁰⁷ Pb	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb	
TX 100	0.06383	1.18011	2.47282	0.4551	0.3239	0.3897	
TX 150	0.06391	1.17955	2.47382	0.6915	0.3628	0.2886	
TX 250	0.06388	1.17533	2.46457	0.7002	0.3574	0.3768	
TX 290	0.06394	1.18409	2.46544	0.3701	0.3899	0.3555	
WLH 200	0.06380	1.20138	2.47694	0.3542	0.2836	0.2199	
WLH 290	0.06362	1.22499	2.50396	0.4526	0.2415	0.4097	
WLH 400	0.06412	1.17778	2.47461	0.4492	0.3046	0.3127	
WLH 490	0.06396	1.20174	2.47499	0.4758	0.2635	0.2097	
WLH 550	0.06389	1.20112	2.48908	0.4881	0.2321	0.2720	
WLH 615							
ST - I	0.06438	1.20822	2.50673	0.4496	0.2087	0.2489	
ST - II	0.06436	1.20869	2.51155	0.2730	0.2925	0.3063	
ST - III	0.06422	1.20921	2.52424	0.1954	0.2996	0.1737	
ST - IV	0.06367	1.21300	2.53387	0.4350	0.2268	0.3037	
ST – JFL no. 1	0.06391	1.21005	2.52877	0.3139	0.2510	0.2301	
ST – JFL no. 2	0.06398	1.21600	2.55082	0.4257	0.3886	0.2356	
ST – JFL no. 3	0.06391	1.22155	2.56819	0.3584	0.2875	0.3090	
ST – JFL no. 4	0.06402	1.21881	2.55009	0.6471	0.1806	0.3574	
ST – JFL no. 5	0.06402	1.22431	2.53670	0.3740	0.1719	0.4177	
ST – JFL no. 6	0.06371	1.22362	2.52590	0.3217	0.3095	0.2857	

Table B-3Lead isotopic compositions of surface soil samples collected fromthe catchment area of the reservoirs.

	Pb Iso	topic Mean	Ratio		RSD (%)			
Sample Name	²⁰⁴ Pb/ ²⁰⁷ Pb	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb	²⁰⁴ Pb/ ²⁰⁷ Pb	²⁰⁶ Pb/ ²⁰⁷ Pb	²⁰⁸ Pb/ ²⁰⁷ Pb		
XMTX – A	0.06429	1.17129	2.46483	0.3718	0.3529	0.3661		
XMTX - B	0.06405	1.17444	2.46399	0.6063	0.2715	0.3951		
XMTX – C	0.06403	1.17525	2.46689	0.5242	0.2471	0.3213		
XMTX – D	0.06413	1.17066	2.46317	0.4607	0.2165	0.2457		
XMTX - E	0.06385	1.16758	2.45566	0.4092	0.3648	0.3215		
WLH-GS 200	0.06384	1.18553	2.47409	0.4989	0.2840	0.3238		
WLH-GS 290	0.06381	1.21162	2.49761	0.4479	0.3245	0.3538		
WLH-GS 400	0.06408	1.17705	2.47743	0.4591	0.3506	0.3671		
WLH-GS 490	0.06367	1.19516	2.48889	0.3578	0.2582	0.2456		
WLH-GS 550	0.06377	1.18659	2.47935	0.2076	0.1419	0.1853		
WLH-GS 600	0.06387	1.17437	2.46498	0.4407	0.3175	0.2579		
WLH-GS 615	0.06385	1.17982	2.46877	0.3834	0.1942	0.2073		
WLH-GS 700	0.06385	1.17333	2.45865	0.5048	0.2676	0.2890		
WLH-GS 760	0.06383	1.17370	2.46201	0.7091	0.4150	0.4599		
LK270	0.06401	1.18334	2.47226	0.4176	0.3895	0.2993		
LK340	0.06384	1.19862	2.48899	0.4476	0.2807	0.3325		
LK360	0.06370	1.22842	2.52902	0.5605	0.2637	0.3067		
ST - I	0.06376	1.19777	2.50486	0.5132	0.2318	0.2739		
ST - II	0.06389	1.20219	2.51099	0.6586	0.6008	0.4093		
ST - III	0.06415	1.20167	2.50972	0.2501	0.3027	0.1960		
ST - IV	0.06389	1.20324	2.51011	0.4721	0.2746	0.3657		

Table B-4 Lead isotopic compositions of moss samples collected from the catchment area of the reservoirs.

APPENDIX C – ANALYTICAL RESULTS OF ²¹⁰Pb

DATING ANALYSIS

Table C-1The concentrations of unsupported ²¹⁰Pb against the depths of thesediment core TX-B at Luokeng Reservoir by ²¹⁰Pb dating analysis

Depth (cm)	Pb (dpm)	Ra (dpm)	Unsupported Pb(dpm)	Average Depth (cm)
1	5.789	2.093	3.697	0.5
3	6.031	2.084	3.946	2.5
5	4.333	2.460	1.874	4.5
8	4.466	2.236	2.230	7.5
11	5.046	2.043	3.003	10.5
15	5.415	1.902	3.513	14.5
19	5.675	2.268	3.407	18.5
24	8.871	2.514	6.357	23.5
29	7.375	1.865	5.509	28.5
34	6.664	1.023	5.642	33.5
39	8.986	1.720	7.266	38.5
45	9.438	2.329	7.109	44.5

Note: 1 dpm (decay per minute) = 1/60 Bq (becquerel) = 0.45045 picocurie (pCi)

Depth (cm)	Pb (dpm)	Ra (dpm)	Unsupported Pb(dpm)	Average Depth (cm)	Year
1	20.704	3.103	17.601	0.5	2005
3	24.623	3.258	21.365	2.5	2004
5	22.682	3.065	19.616	4.5	2003
8	21.033	2.765	18.268	7.5	2001
11	23.795	2.639	21.156	10.5	2000
15	16.841	2.549	14.292	13.5	1998
19	15.456	2.487	12.970	18.5	1995
24	19.145	2.468	16.677	23.5	1992
29	18.096	2.685	15.411	28.5	1989
34	15.333	2.499	12.834	33.5	1987
39	16.642	2.689	13.952	38.5	1984
45	15.745	2.759	12.986	43.5	1981
50	12.452	3.285	9.168	48.5	1978
56	11.573	2.927	8.645	53.5	1975

Table C-2The concentrations of unsupported ²¹⁰Pb against the depths of thesediment core LK-B at Luokeng Reservoir by ²¹⁰Pb dating analysis

Note: 1 dpm (decay per minute) = 1/60 Bq (becquerel) = 0.45045 picocurie (pCi)

Depth (cm)	Pb (dpm)	Ra (dpm)	Unsupported Pb(dpm)	Average Depth (cm)	Year
1	9.210	0.821	8.389	0.5	2005
3	9.290	1.155	8.136	2.5	2004
5	10.437	1.627	8.810	4.5	2002
8	9.853	1.449	8.404	7.5	1999
11	13.139	1.685	11.454	10.5	1997
14	7.130	0.795	6.335	14.5	1994
19	7.558	0.777	6.782	18.5	1990
24	6.921	0.744	6.177	23.5	1986
29	8.945	1.298	7.647	28.5	1982
34	9.248	1.635	7.613	33.5	1978
39	5.327	1.545	3.782	38.5	1974
44	5.314	1.795	3.519	44.5	1969
49	5.207	1.610	3.597	49.5	1965
54	4.986	1.747	3.240	55.5	1960

Table C-3The concentrations of unsupported ²¹⁰Pb against the depths of thesediment core ST-A at Songtao Reservoir by ²¹⁰Pb dating analysis

Note: 1 dpm (decay per minute) = 1/60 Bq (becquerel) = 0.45045 picocurie (pCi)